DETERMINANTS OF WEIGHT GAIN PREVENTION IN ADULT WOMEN

ΒY

CATHERINE JEAN METZGAR

DISSERTATION

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Food Science and Human Nutrition with a concentration in Human Nutrition in the Graduate College of the University of Illinois at Urbana-Champaign, 2016

Urbana, Illinois

Doctoral Committee:

Associate Professor Manabu T. Nakamura, Chair Professor and Department Head Sharon M. Nickols-Richardson, Director of Research Assistant Professor Margarita Teran-Garcia Associate Professor Yuan-Xiang Pan Assistant Professor Brenna Ellison

ABSTRACT

Although the prevalence of overweight and obesity has remained stable in recent years, weight management remains a challenge for a large sector of the population. Particularly during young adulthood, women are at increased risk for excess weight gain. Rather than focusing on the treatment of excess body weight through energy restriction, which has proven to be ineffective in facilitating long-term weight loss, an alternative and more robust approach may be to emphasize the improvement of health and lifestyle behaviors to aid in prevention of weight gain over time. A 1-year randomized controlled trial of weight gain prevention was conducted in a sample of premenopausal women to determine the effects of a weight gain prevention that included nutrition education on body weight (BW) change and other health outcomes over one year. This trial also aimed to compare the delivery of nutrition education by a registered dietitian to a counselor.

Women (n=87) were randomized to a control group (CON; n=29) or weight gain prevention intervention delivered by a registered dietitian (RDG; n=29) or counselor (CSG; n=29). Eighty-one women (mean±SD, age: 31.4±8.1 y; BW: 76.1±19.0 kg; body mass index: 27.9±6.8 kg/m²) completed baseline testing and were included in intention-to-treat analyses (CON=26; RDG=26; CSG=29). During the intervention period, women in the RDG and CSG groups attended 16 weekly and 8 monthly 1-hour nutrition education sessions. Anthropometric, blood pressure, dietary intake, physical activity, biochemical markers of health, eating behaviors, health perceptions and mediators of behavior change data were collected and evaluated at baseline and every three months thereafter. All data were analyzed using the Statistical Package for the Social Sciences (version 22.0, 2013).

The weight gain prevention intervention was successful in preventing weight gain over one year; however, BW change between the RDG, CSG and CON was not significantly different, and 62% of the original sample successfully prevented weight gain. Few differences were observed by group or over time using intention-to-treat analyses. Body fat percentage was significantly lower in the RDG compared to the CSG and CON at all time points (P<0.001). In the CON, systolic blood pressure significantly changed over time (P<0.001), and a group x time interaction for systolic blood pressure was observed (P<0.01). Intake of fruit servings per day

ii

differed significantly at month 6 and 12 between the RDG and CON (P<0.01). No significant group differences were observed for additional anthropometric measurements (P>0.01), resting heart rate (P>0.01), systolic and diastolic blood pressure (P>0.01), macronutrient intake (P>0.01), food group servings (P>0.01), total energy expenditure (P>0.01) or biochemical markers of health (P>0.01). There were no significant effects of time for any anthropometric measurements (P>0.01), resting heart rate (P>0.01), diastolic blood pressure (P>0.01), dietary intake (P>0.01), total energy expenditure (P>0.01) or biochemical markers of health (all P>0.01). A cross-sectional examination of eating behaviors and grit, a non-cognitive personality trait, revealed that disinhibition was a significant predictor of BW and body mass index (BMI). Significant associations between grit and cognitive eating restraint (CER; r=0.23, P<0.05), disinhibition (r=-0.47, P<0.01), hunger (r=-0.19, P=0.05), BW (r=-0.24, P<0.05) and BMI (r=-0.23, P<0.05) were found. Over time, baseline grit did not predict BW change, but it was negatively associated with BW and BMI at month 12 (r=-0.25, P<0.05; r=-0.23, P<0.05). Disinhibition was the only predictor of month 12 BW, and women who successfully prevented weight gain had significantly lower levels of disinhibition at baseline (P<0.05) and significantly increased CER over the intervention (P<0.05).

Though the current weight gain prevention found no significant effects of nutrition education on weight gain prevention over time, a large proportion of individuals were able to maintain BW during the study interval. Further, disinhibition and changes in CER were related to successful weight gain prevention. Future interventions that address additional indicators of health and explore strategies to increase CER and manage disinhibition in order to facilitate prevention of weight gain over time are needed.

iii

ACKNOWLEDGMENTS

To all of you who have played a role in my personal and professional development, thank you.

Specifically, I would like to thank my doctoral advisor, Dr. Sharon Nickols-Richardson, for the opportunity to pursue my doctoral degree at both The Pennsylvania State University and the University of Illinois at Urbana-Champaign. When I think back to the first time our paths crossed while I was interning at The Hershey Company, I would have never imagined that I would be your graduate student a few short years later. Thank you for believing in me and encouraging me to pursue my dreams. Your feedback and support at each step of my graduate career have been invaluable in shaping me as a nutrition scientist and curious individual. Thank you for all of the experiences you have given me and for always helping me to see the big picture. You never cease to inspire me and I am thankful for the opportunity to learn from and with you.

Additionally, I would like to thank my committee members, Dr. Nakamura, Dr. Pan, Dr. Teran-Garcia and Dr. Ellison, for their support of my research and for helping me to problem solve along the way. Additionally, a special thank you to Dr. Garrow and Dr. Teran-Garcia for allowing me use their lab space and equipment to complete my biochemical assays. Thank you to the Department of Food Science and Human Nutrition for the opportunity to complete my doctoral studies, and to the other graduate students for welcoming me to the FSHN Department and providing me with support as I completed my research.

Thank you to the N-R Lab members, Carli Liguori, Henna Muzaffar, Cassandra Nikolaus, Amanda Oakley and April Winslow, for your support and feedback on my research project. Special thanks to Cassandra, Amanda and April for giving up your early mornings and Saturdays to help with data collection and biochemical analyses. Without you, my project would not have been possible. Additionally, a great team of undergraduate students including Virginia Anderson, Samantha Blumenthal, Kelly Chuang, Angie Hwang, Michelle Nussbaum, Joanna Manero, Lauren Schwerzler, Haley Snell, Lauren Williams and Rachel Woo, were essential in recruiting

iv

participants, entering data and helping with testing sessions. I am grateful for the opportunity to work with all of you and thank you for letting me mentor and teach you.

Most importantly, my research would not have been possible without the dedication of my participants with whom I enjoyed building relationships with during the year over which the study took place. These women reminded me that I am making a difference by sharing my passion for nutrition with others to make an impact, no matter how big or small that may be. I am grateful for funding and support provided by The Hershey Company and the University of Illinois at Urbana-Champaign.

To my parents, thank you for raising me to appreciate each and every opportunity and not to take anything for granted. You both have instilled the value of hard work in me and have always encouraged me to pursue my goals. I hope you both are as proud of me as I am proud of you. To my sister, Carina, thank you making me feel good about myself when you told me you found me in PubMed. I can't wait to see what the future has in store for you. Finally, thank you to my husband, Phil. Thank you for never letting me give up and for being my number one supporter through everything. I know our journey has been full of twist and turns, but all that matters is that we get to live this adventure of life together. I love you all.

TABLE OF CONTENTS

LIST OF ABBREVIATIONS	vii
CHAPTER 1: Introduction	1
CHAPTER 2: Literature review	7
CHAPTER 3: Preliminary research	30
CHAPTER 4: Materials and methods	85
CHAPTER 5: Effect of nutrition education on weight gain prevention: a randomized controlled trial	112
CHAPTER 6: Grit: association with eating behaviors, body weight and body mass index	. 141
CHAPTER 7: The role of grit in body weight regulation over time	150
CHAPTER 8: Conclusions and future directions	163
APPENDIX A: IRB approval letter	168
APPENDIX B: Permission to reprint	174

LIST OF ABBREVIATIONS

ACC	American College of Cardiology
AHA	American Heart Association
ANCOVA	analysis of covariance
ANOVA	analysis of variance
BF%	body fat percentage
BMI	body mass index
BW	body weight
CER	cognitive eating restraint
CON	control group
CSG	counselor group
CV	coefficient of variation
DC	dark chocolate
EI	Eating Inventory
ELISA	enzyme-linked immunosorbent assay
ERD	energy-restricted diet
FITT	frequency, intensity, time and type
FM	fat mass
GOAL	Groningen Overweight and Lifestyle
Grit-S	Short Grit Scale
HDL-C	high-density lipoprotein cholesterol
HOMA-IR	homeostasis model of assessment-insulin resistance
IRB	Institutional Review Board
kcal	kilocalories
LDL-C	low-density lipoprotein cholesterol
MET	metabolic equivalent
NC	non-chocolate
NDSR	Nutrition Data System for Research
РОР	Pound of Prevention
RDG	registered dietitian group
SCT	social cognitive theory
SD	standard deviation
SF-36	Short-Form 36 Health Survey
SNAP	Study of Novel Approaches to Weight Gain Prevention
svgs	servings
TG	triglycerides or triacylglycerides
TOS	The Obesity Society
UIUC	University of Illinois at Urbana-Champaign

While prevalence has stabilized in recent years, overweight and obesity remain major public health concerns for the United States population (1). Overweight and obesity are associated with metabolic abnormalities (2,3), decreased quality of life (4,5) and increased healthcare costs (6). While treatment efforts for overweight and obesity have focused primarily on weight loss through energy restriction, no methods have been effective in promoting longterm weight loss for a majority of individuals in the United States or decreasing the prevalence of overweight and obesity (7,8). Further, while weight loss is known to promote metabolic benefits (9-11), it may also result in some negative health effects such as an increased risk for weight cycling (12), disruption of hormonal milieu (13) and decreased bone mineral density (14).

It is obvious that a new approach to weight management is needed. Weight gain prevention is one such approach that removes the focus from energy restriction and weight loss and instead focuses on improving lifestyle behaviors to manage weight over time. Overweight and obesity generally do not develop rapidly; rather, they are the result of small, cumulative weight gains over time (15,16). Particularly, women are at highest risk for weight gain during young adulthood and midlife (15,16) as a result of various life transitions (17). Little is known regarding weight gain prevention, as few studies have addressed primary weight gain prevention (18-23) and none have included all individuals for which weight gain prevention is applicable, that is normal weight, overweight and obese individuals. Further, as determinants of weight gain prevention most likely differ between men and women, studies that examine one gender will help to inform gender-specific recommendations and intervention methods.

This dissertation examines the findings from a randomized controlled trial that sought to prevent weight gain over one year in normal weight, overweight and obese premenopausal women using nutrition education. Additionally, this research sought to compare the delivery of a weight gain prevention intervention by a registered dietitian (i.e., an individual with formal nutrition training) to a counselor (i.e., an individual with no formal nutrition training).

The primary aim of this research was to prevent the burden of obesity among adult women by identifying determinants of weight gain prevention as guided by the Social Cognitive Theory (24). Specific objectives for this research included:

- Filling the scientific gap in the understanding of the determinants of weight gain prevention in women;
- 2) Aiding women in maintaining body weight, within 3%, over the study duration;
- Providing general nutrition education and fitness knowledge to induce behavior change; and
- Identifying optimal strategies and key recommendations for prospective weight gain prevention.

The first hypothesis was that women participating in a weight gain prevention intervention who were randomized to receive nutrition education would maintain body weight over a 1-year interval when compared to a control condition. Second, it was hypothesized that women receiving nutrition education from a registered dietitian would have less weight gain over the 1-year interval when compared to women receiving the same nutrition education from a counselor.

Within this dissertation, Chapter 2 reviews the literature and provides an overview of the current state of obesity, including current treatment approaches, and describes why a new frame of mind is necessary. The current body of research in relation to weight gain prevention is further examined and summarized in Chapter 2. Chapter 3 presents two preliminary studies that examined weight loss and weight loss maintenance in women. These studies took a quantitative and qualitative approach, and findings informed the design and intervention aspects of the current dissertation research. The methodology of the current randomized controlled trial for weight gain prevention is described in Chapter 4, and major outcomes and findings are presented in Chapter 5. Chapter 6 includes the results from a cross-sectional examination of the relationship between grit, a personality trait related to perseverance, and the eating behaviors of cognitive eating restraint, disinhibition and hunger. Grit was further

examined as a predictor of weight regulation over time in Chapter 7. Finally, Chapter 8 provides a brief summary of the major findings of this dissertation and offers opportunities and directions for future research. Supporting documentation, including Institutional Review Board approval and permission to reprint articles, are included in the Appendices.

Results from this comprehensive investigation further add to the meager body of evidence that has examined weight gain prevention and provide avenues for improving future interventions. In addition, this research provides support that weight gain prevention is relevant for all individuals, across healthy, overweight and obese body mass index categories, and aligns with the mission of public health to *prevent* diseases, such as overweight and obesity, and *promote* health rather than solely focus on *treatment* approaches for excess body weight.

References

- 1. Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. JAMA. 2012;307:491-7.
- Després JP. Body fat distribution and risk of cardiovascular disease: an update. Circulation. 2012;126:1301-13.
- Poirier P, Giles TD, Bray GA, et al. Obesity and cardiovascular disease: a pathophysiology, evaluation, and effect of weight loss. Arterioscler Thromb Vasc Biol. 2006;26:968-76.
- Fontaine KR, Barofsky I. Obesity and health-related quality of life. Obes Rev. 2001;2;173-82.
- 5. Kushner RF, Foster GD. Obesity and quality of life. Nutrition. 2000;16:947-52.
- Wang Y, Boydoun MA, Liang L, Caballero B, Kumanyika SK. Will all Americans become overweight or obese? Estimating the progression and cost of the US obesity epidemic. Obesity (Silver Spring). 2008;16:2323-30.
- 7. Mann T, Tomiyama J, Westling E, Lew AM, Samuels B, Chatman J. Medicare's search for effective obesity treatments: diets are not the answer. Am Psychol. 2007;62:220-33.
- 8. Lindvall K, Jenkins P, Emmelin M, et al. Primary weight maintenance: an observational study exploring candidate variables for intervention. Nutr J. 2013;12:97.
- Shlisky JD, Durward CM, Zack MK, Gugger CK, Campbell JK, Nickols-Richardson SM. An energy-restricted dietary pattern, including moderate protein and increased nonfat dairy intake combined with walking promotes beneficial body composition and metabolic changes in women with excess adiposity: a randomized comparative trial. Food Sci Nutr. 2015;3:376-93.
- 10. Ades PA, Savage PD. Potential benefits of weight loss in coronary heart disease. Prog Cardiovasc Dis. 2014;56:448-56.
- Goldstein DJ. Beneficial health effects of modest weight loss. Int J Obes Relat Metab Disord. 1992;16:397-415.

- Neumark-Sztainer D, Wall M, Story M, Standish AR. Dieting and unhealthy weight control behaviors during adolescence: association with 10-year changes in body mass index. J Adolesc Health. 2012;51:80-6.
- Schwarz NA, Rigby BR, La Bounty P, Shelmadine B, Bowden RG. A review of weight control strategies and their effects on the regulation of hormonal balance. J Nutr Metab. 2011:237932. doi: 10.1155/2011/237932.
- Hunter GR, Plaisance EP, Fischer G. Weight loss and bone mineral density. Curr Opin Endocrinol Diabetes Obes. 2014;21:358-62.
- 15. Williamson DF, Kahn HS, Remington PL, Anda RF. The 10-year incidence of overweight and major weight gain in US adults. Arch Intern Med. 1990;150:665-72.
- 16. Truesdale KP, Stevens J, Lewis CE, Schreiner PJ, Loria CM, Cai J. Changes in risk factors for cardiovascular disease by baseline weight status in young adults who maintain or gain weight over 15 years: the CARDIA study. Int J Obes (Lond). 2006;30:1397-407.
- Metzgar CJ, Preston AG, Miller DL, Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. J Hum Nutr Diet. 2015;28:593-603.
- 18. Forster JL, Jeffery RW, Schmid TL, Kramer FM. Preventing weight gain in adults: a pound of prevention. Health Psychol. 1988;7:515-25.
- Jeffery RW, French SA. Preventing weight gain in adults: design, methods and one year results from the Pound of Prevention study. Int J Obes Relat Metab Disord. 1997;21:457-64.
- Levine MD, Klem ML, Kalarchian MA, et al. Weight gain prevention among women.
 Obesity (Silver Spring). 2007;15:1267-77.
- ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing weight gain: one-year results of a randomized lifestyle intervention. Am J Prev Med. 2009;37:270-7.
- 22. Bennett GG, Foley P, Levine E, et al. Behavioral treatment for weight gain prevention among black women in primary care practice: a randomized clinical trial. JAMA Intern Med. 2013;173:1770-7.

- 23. Wing RR, Tate D, Espeland M, et al. Weight gain prevention in young adults: design of the study of novel approaches to weight gain prevention (SNAP) randomized controlled trial. BMC Public Health. 2013:13;300. doi: 10.1186/1471-2458-13-300.
- 24. Bandura A. Social foundation of thought and action: a social cognitive theory. Englewood Cliffs, NJ: Prentice-Hall; 1986.

CHAPTER 2: Literature review

Excess body weight, in the form of overweight and/or obesity, is a growing global public health concern. While the prevalence of overweight and obesity has stabilized in recent years, more than two-thirds of adults in the United States, including sixty-some percent of women, remain overweight or obese (1), and it is estimated that the prevalence of overweight and obesity among women will reach 80% by 2022 (2).

Overweight and obesity generally result from gradual weight gain over decades, averaging one to two pounds per year (3-5). Rates of weight change have been estimated at 0.39 kilograms (0.86 pounds) per year for white American women and 0.54 kilograms (1.2 pounds) for black American women (6). Young adulthood has been identified as a stage of the lifespan during which individuals are at high risk of excess weight gain (6-9). For women, young adulthood encompasses many life transitions and potential stressors, such as pregnancy, postpartum, increased family responsibilities and changes in health status that may accompany aging (10). Consequently, weight management programs and interventions should target this period of the lifespan.

The economic burden of obesity

Obesity has great economic effects on the nation (2, 11-15). Obesity has been estimated to increase healthcare costs by \$3,508 per person each year, and in 2010, it was estimated that obesity among adults increased annual healthcare costs by nearly \$316 billion (13). By 2020, researchers estimate that \$394 to \$438 billion will be spent on direct healthcare costs related to obesity (2), and by 2030, this cost will escalate to roughly \$860 to \$957 billion or 16-18% of total healthcare costs (2). At the individual level, those with excess body weight have 30% higher healthcare costs, compared to individuals of normal body weight (12). Statistical models indicate that a 5-10% reduction in excess body weight (13) or a one percent decline in the predicted prevalence of obesity (14) may result in substantial healthcare cost savings, however, most individuals are unable to maintain weight losses (16-21) and return to previous baseline body weight within three to five years (17-19). Sustained levels of obesity prevalence (14),

which may be achieved by prevention of weight gain via behavioral lifestyle modifications, could result in similar cost savings by preventing or managing weight-related co-morbidities in the absence of weight change.

Obesity and quality of life

Indirect costs of overweight and obesity, including decreased quality of life (22,23), may have negative effects at the societal and individual levels (15). Decreased quality of life may be manifested in ways such as decreased self-esteem, lowered interest in activities and reduced mobility and physical functioning. Reductions in excess body weight have been shown to improve quality of life in the short term (24,25), specifically related to the Short-Form 36 Health Survey (SF-36) constructs of physical functioning, vitality, role-physical, mental health and general health (24); however, these improvements may be diminished over the long term, regardless of whether an individual is able to maintain weight loss (26). More research is needed to demonstrate the efficacy of weight loss through obesity treatment in improving quality of life (23) and how weight regain may impact quality of life. The effect of other nonweight related avenues to improve quality of life, including weight gain prevention, should also be explored.

Treatment of excess body weight

While weight loss via dieting or caloric restriction is the generally recommended treatment for excess body weight, it is only effective in reducing the prevalence of overweight and obesity if weight loss can be maintained (20). Undoubtedly, dieting and caloric restriction lead to short-term weight loss (20, 27-33); however, these losses are frequently not maintained over the long term (16-21). Regardless of the method or program used to induce weight loss, six months is the average length of adherence (17). Further, the likelihood of maintaining weight loss decreases over time (34). Within three to five years after initial weight loss, most individuals regain all of the weight lost, with half of this weight regain occurring in the first year following weight loss (17-19). Even more dismal, a review of 14 studies found that in eight of the studies reviewed, weight was regained to levels above participants' starting body weight

(20). Interventions that incorporate both dietary and physical activity components may be most effective in preventing weight regain (35,36) but long-term follow up (>24 months) is limited. Other factors found to delay or slow weight regain include face-to-face interventions (35) and extended care/continued contact (10,37). As it appears that weight loss maintenance is a missing piece of many weight loss trials and that most individuals are unable to maintain weight loss, it is difficult to classify diets, and even short-term weight loss, as effective treatments for excess body weight (20, 38). New approaches to understanding the determinants of weight gain prevention are necessary (38).

Although modest weight losses of 5-10% of body weight are associated with improvements in obesity-related risk factors, few studies support that these health benefits are maintained with weight regain (20, 39). Further, weight regain could lead to weight cycling, or repeated losses and gains of body weight. Although true relationships remain inconclusive, weight cycling has been associated with an increased risk for morbidity and mortality, possibly due to the negative effects on metabolic, psychological and mental outcomes (20, 40-42). Additionally, weight cycling has been found to be predictive of weight regain (21). Taken together, these findings demonstrate that long-term benefits of weight loss are minimal, and that weight cycling and/or weight regain may pose further health risks. Therefore, it may be time to reconsider weight loss as the most successful avenue for addressing the burden of excess body weight.

Weight gain prevention: an alternative to weight loss

Weight gain prevention may be a more robust alternative to weight loss. While small weight gains over the course of one year may seem clinically insignificant, these small weight gains accumulate over time and contribute to the development of overweight and obesity. Once excess body weight is present, weight loss and weight loss maintenance prove to be challenging for many individuals and success varies widely depending on an individual's behavior, physiology and environment. While behavior and environment can be altered, physiology and genetic predisposition are much less pliable. The treatment or intervention required to prevent weight gain is less intensive than that needed to promote weight loss (5)

and removes the burden or stigma placed on individuals to lose weight (38). Primary weight maintenance, also referred to as weight gain prevention, offers a more realistic avenue to curtail the obesity epidemic by slowing weight gain and mitigating risk factors associated with weight gain over time (43).

Additionally, weight gain prevention is applicable for all segments of the population, regardless of whether an individual is normal weight, overweight or obese. In individuals with overweight and obesity, weight gain prevention may play a role in decreasing the risk of developing concomitant diseases or other risk factors, or may aid in controlling or improving current diseases and related symptoms (44). Although obesity is associated with other co-morbidities and risk factors, it is important to note that not all individuals with obesity have, or will develop, metabolic abnormalities (45,46) and that weight gain prevention may play a role in the management and prevention of developing these abnormalities. Individuals with overweight are at greatest risk for becoming obese; therefore, weight gain prevention is especially important for this sector of the population (47). Promoting weight gain in individuals of normal weight also will help prevent the progression to overweight and/or obesity.

Weight loss may not be a realistic or appropriate goal for some individuals, and repeated weight loss attempts may have detrimental effects on health. One longitudinal study found that the probability of individuals with obesity to attain normal body weight status was just 0.8% for women (and only 0.5% for men) (48). For these individuals and others, prevention of additional weight gain may be a more appropriate treatment plan and target for treatment programs and interventions. Prevention of weight gain can be achieved through lifestyle and behavior changes which may result in improved health status, even in the absence of weight change.

Weight gain prevention studies

Primary prevention has been identified as the most affordable and effective method to prevent obesity and other chronic diseases (11), yet low quality evidence supports the effectiveness of interventions in reducing and/or maintaining body weight (44). One reason for the low quality of evidence is the fact that little work has targeted primary prevention of

obesity (Table 2.1). Instead, much of the current literature has focused on weight loss and weight loss maintenance, otherwise known as secondary weight maintenance (38). The first weight gain prevention trial was conducted nearly 30 years ago in normal weight adults (<115% of ideal body weight) between the ages of 25-74 years (49). This study compared the effects of low-intensity nutrition education and financial incentives on weight change to a no-contact control over a 12-month period. Newsletters that included general nutrition information regarding nutrition, physical activity and other lifestyle components were sent to participants who were randomized to the treatment group each month. With each newsletter, participants were asked to self-report body weight and any strategies used to prevent weight gain. A financial incentive was also included in the treatment intervention. For participants wanting or needing greater assistance in weight gain prevention, four intensive nutrition education sessions were offered around the 6-month mark of the intervention. At the end of 12 months, significantly more individuals in the treatment group (82%) maintained or lost weight as compared to the no-contact control group (56%), and the difference in weight loss between the two groups was significantly different (-2.1 pounds vs. -0.3 pounds).

Extending this early work, Jeffery and French investigated weight gain prevention over three years in adults between the ages of 20-45 years (50,51). Body weight was not used as entry criteria for the study; however, participants were overweight, on average. Participants were randomized to one of two low-intensity nutrition education groups or to a no-contact control group. Participants in both of the nutrition education groups received monthly newsletters that emphasized regular self-weighing, increased consumption of fruits and vegetables, decreased consumption of high-fat foods, and increased participation in exercise, specifically walking. Similar to the first weight gain prevention study, participants were asked to self-report body weight and answer questions related to the major themes emphasized by the newsletters. Semiannual nutrition and exercise classes were offered to education intervention participants as well. The education groups were identical in materials and messages, except that one group received a lottery incentive for returning the monthly postcard. After one year (50) and three years (51), there were no significant differences in weight change between groups; however, participants in the nutrition education groups reported a significant increase

in frequency of self-weighing (50,51). Of the entire study population, just 37% of participants maintained (or lost) weight over the 3-year period (51). Although weight-related behaviors were positively impacted by nutrition education and this was related to a decrease in weight gain over time, weight gain was not significantly different between individuals randomized to the treatment or control groups (51). An additional cross-sectional and prospective analysis of participants in the Pound of Prevention study that aimed to identify behaviors that predicted and/or accompanied weight change over time concluded that the determinants of weight regulation are multifactorial (52). Dietary fat intake and lack of physical activity were behaviors most consistently related to weight gain over time in women (52). Even though overall results of the Pound of Prevention study were not significant, the authors recommended several suggestions for future weight gain prevention programs, including more interactive interventions that include increased frequency of messaging, tailored information for participants and attention to motivational issues (51).

Levine and colleagues examined weight gain prevention over a 3-year period (2-year intervention plus 1-year follow-up) in normal weight and overweight women between the ages of 25-44 years (53). Participants were randomized to an information-only control group, a clinic-based nutrition education intervention or a correspondence-based nutrition education intervention. Women randomized to the clinic-based group attended 15 group meetings led by trained nutritionists and behavioral interventionists over the 2-year intervention; sessions were held biweekly for the first two months and bimonthly for the remaining 22 months of the study. The correspondence-based group received the same 15 lessons as the clinic-based group by mail along with a brief homework assignment that included a question about current self-reported weight. Upon randomization, women in the control group received a booklet (54) containing general wellness information. Over three years, there were no significant differences between groups in changes in body weight and body mass index (BMI), and nearly 40% of all participants, regardless of group assignment, lost or maintained body weight. More women in the clinic-based and control groups remained weight stable when compared to the correspondence-based group, but these differences were not significant. Overall, older age,

lower feelings of hunger and dieting status at baseline predicted weight gain prevention in this sample of normal weight and overweight women. Since neither intervention was found to be superior to the information booklet, the authors called for more innovative and intensive approaches, in addition to stronger public health messages, to aid in weight gain (and obesity) prevention.

The Groningen Overweight and Lifestyle (GOAL) study compared the effects of computerguided lifestyle counseling by a nurse practitioner to usual care by a general practitioner on weight gain prevention and health status in overweight and obese men and women, between the ages of 40-70 years, with hypertension and/or dyslipidemia (55-57). Lifestyle counseling by the nurse practitioner included four individual visits and one telephone feedback session during the first year of the study (55,56) and one individual visit and two telephone feedback sessions during years two and three (57). A standardized computer software program that followed nutritional and behavioral guidelines was used by nurse practitioners to guide sessions with participants. Participants randomized to the control group received usual care defined as one, 10-minute visit with the general practitioner.

After one year, approximately 70% of all participants maintained or lost weight (55,56); this number decreased to 60% after three years (57). Significantly more individuals in the intervention group successfully maintained or lost weight after one year when compared to the control group (55,56), but these differences were not present three years later (57). After one year and three years, there were no significant differences in weight change between groups (55-57); however, men randomized to the intervention group significantly reduced body weight and waist circumference when compared to the control group after one year (55,56), but these differences significant effects were not seen in women, the authors hypothesized that more intensive interventions may be necessary to prevent weight gain in women (55,56). Lifestyle counseling by the nurse practitioner did result in a significant decrease in fasting blood glucose levels, but had no effect on blood pressure or lipid levels (57).

A medium-intensity behavioral intervention designed to prevention weight gain was evaluated in overweight and class 1 obese premenopausal black women between the ages of 25-44 years as part of the Shape Program (58). Usual care that included biannual general

wellness newsletters was compared to a weight gain prevention intervention that included weekly self-monitoring, monthly counseling calls with a registered dietitian, skills training tailored to behavior change goals and a 12-month gym membership. At one year, significantly more women in the intervention group maintained or lost weight as compared to women in the usual care group. Weight loss was significantly greater in the intervention group at the end of the intervention, and these significances were still present six months following the intervention. However, there were no significant differences between groups for changes in other measures including blood pressure, waist circumference, or glucose or lipid levels at 12 or 18 months.

Final results of the most recent weight gain prevention study, the Study of Novel Approaches to Weight Gain Prevention (SNAP), are not yet available (59). The SNAP program targeted weight gain prevention in normal weight and overweight young adults between the ages of 18-35 years. A minimal intervention control was compared to one of two novel selfregulation approaches to weight management over an average of three years. Individuals in both self-regulation intervention groups attended eight weekly group sessions and two monthly group sessions as part of the initial treatment, followed by monthly contact via telephone and/or email, but with one group focused on small, manageable behavior changes and the other group focused on periodic, large behavior changes. While final results are not yet available, it is hypothesized that both self-regulation approaches will be more successful in promoting weight gain prevention as compared to the control group, and that the large changes self-regulation approach will have the greatest influence on weight gain prevention.

In summary, few studies have targeted primary prevention of weight gain in adults, and the results of studies that have examined weight gain prevention are not encouraging. Just two of the six studies included in this review found significant impacts of interventions on weight gain over time (49,58). Interventions ranged from low-intensity nutrition education via newsletters to medium-intensity interventions that included group sessions with registered dietitians or trained nutritionists. Additionally, the amount of contact or education given to the control group varied between studies, ranging from absolutely no contact to general wellness information provided through a booklet or newsletter.

Even though body weight was not substantially impacted by interventions, researchers reported positive changes in dietary and lifestyle behaviors. Not all studies measured additional health outcomes, so it is unclear if these changes had positive effects on biomarkers or disease risk even in the absence of weight change. The studies examining weight gain prevention varied in participant characteristics. Four studies (49,50,55,59) included men and women, while two studies included only women (53,58). Age ranges were similar for most studies; however, participants in the GOAL study (55-57) were older, and the age range was wide in A Pound of Prevention (49) when compared to other studies. One limitation of the current body of weight gain prevention evidence is that previous studies have focused on either normal weight *or* overweight/obese populations (not both). More research is clearly needed to understand the determinants of weight gain prevention in order to effectively develop strategies for long-term weight management (38,60). Without such an understanding, individuals remain at risk for increased healthcare costs as well as other health complications related to excess body weight.

Efficacy of registered dietitians in adult weight management

The 2013 American Heart Association (AHA)/American College of Cardiology (ACC)/The Obesity Society (TOS) Guideline for the Management of Overweight and Obesity in Adults recommend a comprehensive lifestyle intervention that includes referrals to nutrition professionals, such as registered dietitians, to treat overweight and obesity (61). Registered dietitians are viewed as credible sources of nutrition information (10,62) and have been shown to positively impact health- and weight-related outcomes (63-70) while also decreasing healthcare costs (65,67). When compared to usual medical care, treatment that included registered dietitians decreased diabetes risk and improved glucose control over a 12-week period in adults with prediabetes (70). Over a 12-month period, treatment by registered dietitians decreased anthropometric measurements, increased quality of life and reduced usage of prescription medications in individuals with obesity and type 2 diabetes (64), resulting in a nearly \$4,000 decrease in healthcare costs per person (65). In men with hyperlipidemia, an average of three, 1-hour sessions with a registered dietitian over an 8-week period significantly reduced BMI, total cholesterol, low-density lipoprotein cholesterol and triglycerides which

resulted in a decrease in use of medications to manage hyperlipidemia and translated to an approximate saving of three dollars for every one dollar spent on medical nutrition therapy (67). While overweight and obese individuals who did not receive medical nutrition therapy were able to significantly reduce anthropometric measures and increase exercise frequency over a 2-year period, individuals who participated in medical nutrition therapy achieved greater decreases in body weight and BMI (69). An observational study of cardiac rehab patients found that nutrition education and counseling by a registered dietitian significantly decreased triglycerides, total cholesterol, low-density lipoprotein cholesterol, BMI and waist circumference when compared to general nutrition education delivered by cardiac rehabilitation staff members (63). Taken together, findings from previous research studies demonstrate that registered dietitians are an efficacious component of programs and interventions.

The use of peer educators

Although registered dietitians are food and nutrition experts who have completed education and experience requirements related to food, nutrition, health and behavior, the inclusion of registered dietitians in programs or interventions may prove to be a barrier. Limited time and lack of reimbursement have been cited as issues that prevent physicians from being more involved in the treatment of overweight and obesity (71); these reasons also may explain why the use of registered dietitians in programs and interventions is not widespread. However, the use of lay health educators or peers may be an alternative to registered dietitians, as inclusion of these individuals in programs and interventions has been associated with a decrease in chronic disease risk factors for diabetes and cardiovascular disease (72-76). In individuals with prediabetes, a 24-month lifestyle intervention delivered by community health workers resulted in significant reductions in body weight, BMI, waist circumference, insulin, insulin resistance and glucose as compared to enhanced usual care that included two visits with a registered dietitian (74). A unique aspect of this study was that the community health workers were recruited, trained and monitored throughout the intervention by registered dietitians who also provided support (74). West and colleagues compared lay health educators trained by

research staff with a wait-list control group in one, 12-session intervention for older adults modeled on the Diabetes Prevention Program (75). Compared to the control group, individuals in the intervention group experienced significantly greater weight loss and improved lifestyle behaviors (75). Similar findings were reported in a peer-led lifestyle intervention, although training of peers was not explained (76) and in a pilot study that evaluated the use of trained peer coaches to deliver weight loss treatment in African Americans with diabetes or prediabetes (72). Although peer coaches received in-depth training, findings from Dutton (72) should be interpreted with caution, as this pilot study did not include a control group. Another study that compared the facilitation of a 6-month weight loss trial by professional, peer and mentor coaches, found that individuals led by professional and peer coaches had the largest decrease in body weight but that all groups were able to significantly decrease body weight (73). Even without a substantial nutrition or health background, lay professionals are able to promote significant health improvements in a variety of populations with adequate training by trained health professionals. Registered dietitians can expand their field of reach and utilize their expertise by adequately recruiting, training, monitoring and supporting lay educators. As the availability of registered dietitians may be limited in certain areas, the use of lay educators can help to fill this gap while meeting the needs of communities and individuals and serve as a cost-effective alternative (73,74).

Next steps in weight gain prevention

With limited research addressing primary weight gain prevention, strategies for preventing weight gain over time remain largely unknown. As different strategies are required to achieve weight loss and weight loss maintenance, strategies for successful weight gain prevention most likely differ as well (10). Without more weight gain prevention programs and interventions, the determinants of weight maintenance cannot be identified and an impact on public health and obesity prevalence cannot be made. Therefore, more programs and interventions are necessary to examine the feasibility of weight gain prevention and the underlying determinants that lead to gradual weight gain over time. These interventions should include individuals of all weight status categories, and gender should be separated as

determinants and strategies most likely differ between men and women (10,77). The intensity of future interventions should also be considered, as should the mode of delivery and who delivers the information. More trials including registered dietitians and adequately trained peer educators are necessary.

Due to the limited research in this area, there is an incomplete understanding of weight gain prevention. In order to gain a more comprehensive view, it is necessary to build upon past findings to develop optimal approaches to adult weight management. With potential benefits such as reduced healthcare costs, decreased obesity prevalence and improvement of chronic disease risk factors, weight gain prevention is a topic worthy of future exploration.

References

- 1. Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. JAMA. 2012;307:491-7.
- Wang Y, Boydoun MA, Liang L, Caballero B, Kumanyika SK. Will all Americans become overweight or obese? Estimating the progression and cost of the US obesity epidemic. Obesity (Silver Spring). 2008;16:2323-30.
- 3. Mozaffarian D, Hao T, Rimm EB, Willett WC, Hu FB. Changes in diet and lifestyle and long-term weight gain in women and men. N Engl J Med. 2011;364:2392-404.
- 4. Williamson DF, Kahn HS, Remington PL, Anda RF. The 10-year incidence of overweight and major weight gain in US adults. Arch Intern Med. 1990;150:665-72.
- Truesdale KP, Stevens J, Lewis CE, Schreiner PJ, Loria CM, Cai J. Changes in risk factors for cardiovascular disease by baseline weight status in young adults who maintain or gain weight over 15 years: the CARDIA study. Int J Obes (Lond). 2006;30:1397-407.
- Sheehan TJ, DuBrava S, DeChello LM, Fang Z. Rates of weight change for black and white Americans over a twenty year period. Int J Obes Relat Metab Disord. 2003;27:498-504.
- Wing RR. Changing diet and exercise behaviors in individuals at risk for weight gain. Obes Res. 1995;3:277s-82s.
- 8. Ball K, Brown W, Crawford D. Who does not gain weight? Prevalence and predictors of weight maintenance in young women. Int J Obes Relat Metab Disord. 2002;26:1570-8.
- 9. Wane S, van Uffelen JG, Brown W. Determinants of weight gain in young women: a review of the literature. J Womens Health (Larchmt). 2010;19:1327-40.
- Metzgar CJ, Preston AG, Miller DL, Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. J Hum Nutr Diet. 2015;28:593-603.
- Slawson DL, Fitzgerald N, Morgan KT. Position of the Academy of Nutrition and Dietetics: the role of nutrition in health promotion and chronic disease prevention. J Acad Nutr Diet. 2013;113:972-9.

- 12. Withrow D, Alter DA. The economic burden of obesity worldwide: a systematic review of the direct costs of obesity. Obes Rev. 2011;12:131-41.
- 13. Cawley J, Meyerhoefer C, Biener A, Hammer M, Wintfeld N. Savings in medical expenditures associated with reductions in body mass index among US adults with obesity, by diabetes status. Pharmacoeconomics. 2015;33:7707-22.
- 14. Finkelstein EA, Khavjou OA, Thompson H, et al. Obesity and severe obesity forecasts through 2030. Am J Prev Med. 2012;42:563-70.
- 15. Wolf AM, Colditz GA. Current estimates of the economic costs of obesity in the United States. Obes Res. 1998;6:97-106.
- Wing RR, Phelan S. Long-term weight loss maintenance. Am J Clin Nutr. 2005;82:2225-5S.
- 17. Goodrick GK, Poston WS, Foreyt JP. Methods for voluntary weight loss and control: update 1996. Nutrition. 1996;12:672-6.
- 18. Jeffery RW, Drewnowski A, Epstein LH, et al. Long-term maintenance of weight loss: current status. Health Psychol. 2000;19:5-16.
- 19. Byrne S, Cooper Z, Fairburn C. Weight maintenance and relapse in obesity: a qualitative study. Int J Obes Relat Metab Disord. 2003;27:955-62.
- 20. Mann T, Tomiyama J, Westling E, Lew AM, Samuels B, Chatman J. Medicare's search for effective obesity treatments: diets are not the answer. Am Psychol. 2007;62:220-33.
- 21. McGuire MT, Wing RR, Klem ML, Lang W, Hill JO. What predicts weight regain in a group of successful weight losers? J Consult Clin Psychol. 1999:67;177-85.
- Fontaine KR, Barofsky I. Obesity and health-related quality of life. Obes Rev. 2001;2;173-82.
- 23. Kushner RF, Foster GD. Obesity and quality of life. Nutrition. 2000;16:947-52.
- 24. Fontaine KR, Barofsky I, Andersen RE, et al. Impact of weight loss on health-related quality of life. Qual Life Res. 1999;8:275-7.
- Rothberg AE, McEwen LN, Kraftson AT, et al. The impact of weight loss on healthrelated quality-of-life: implications for cost-effective analyses. Qual Life Res. 2014;23:1371-6.

- 26. Fontaine KR, Barofsky I, Bartlett SJ, Franckowiak SC, Andersen RE. Weight loss and health-related quality of life: results at 1-year follow-up. Eat Behav. 2004;5:85-8.
- 27. Hill JO, Thompson H, Wyatt H. Weight maintenance: what's missing? J Am Diet Assoc.2005;105:S63-6.
- 28. Dansinger ML, Gleason JA, Griffith JL, Selker HP, Schaefer EJ. Comparison of the Atkins, Ornish, Weight Watchers, and Zone diets for weight loss and heart disease risk reduction: a randomized trial. JAMA. 2005;293:43-53.
- 29. Gardner CD, Kiazand A, Alhassan S, et al. Comparison of the Atkins, Zone, Ornish, and LEARN diets for change in weight and related risk factors among overweight premenopausal women: the A TO Z Weight Loss Study: a randomized trial. JAMA. 2007;297:969-77.
- Shai I, Schwarzfuchs D, Henkin Y, et al. Weight loss with a low-carbohydrate, Mediterranean, or low-fat diet. N Engl J Med. 2008;359:229-41.
- Makris A, Foster GD. Dietary approaches to the treatment of obesity. Psychiatr Clin North Am. 2011;34:813-27.
- 32. Sacks FM, Bray GA, Carey VJ, et al. Comparison of weight-loss diets with different compositions of fat, protein, and carbohydrates. N Engl J Med. 2009;360:859-73.
- 33. McGuire MT, Wing RR, Klem ML, Seagle HM, Hill JO. Long-term maintenance of weight loss: do people who lose weight through various weight loss methods use different behaviors to maintain their weight? Int J Obes Relat Metab Disord. 1998;22:572-7.
- 34. Abildso CG, Schmid O, Byrd M, Zizzi S, Quartiroli A, Fitzpatrick SJ. Predictors of weight loss maintenance following an insurance-sponsored weight management program. J Obes. 2014; 2014:736080. doi: 10.1155/2014/736080.
- 35. Dombrowski SU, Knittle K, Avenell A, Araújo-Soares V, Sniehotta FF. Long term maintenance of weight loss with non-surgical interventions in obese adults: systematic review and meta-analyses of randomised controlled trials. BMJ. 2014;348:g2646. doi: 10.1136/bmj.g2646.

- Stelmach-Mardas M, Mardas M, Walkowiak J, Boeing H. Long-term weight status in regains after weight loss by intervention: status and challenges. Proc Nutr Soc. 2014;73:509-18.
- Middleton KM, Patidar SM, Perri MG. The impact of extended care on long-term maintenance of weight loss: a systematic review and meta-analysis. Obes Rev. 2012;13:509-17.
- 38. Lindvall K, Jenkins P, Emmelin M, et al. Primary weight maintenance: an observational study exploring candidate variables for intervention. Nutr J. 2013;12:97.
- 39. Weiss EC, Galuska DA, Khan LK, Gillespie C, Serdula MK. Weight regain in U.S. adults who experienced substantial weight loss, 1999-2002. Am J Prev Med. 2007;33:34-40.
- 40. Kayman S, Bruvold W, Stern JS. Maintenance and relapse after weight loss in women: behavioral aspects. Am J Clin Nutr. 1990:52:800-7.
- 41. Jeffery RW. Does weight cycling present a health risk? Am J Clin Nutr. 1996;63:452S-5S.
- 42. Vergnaud AC, Bertrais S, Oppert JM, et al. Weight fluctuations and risk for metabolic syndrome in an adult cohort. Int J Obes Relat Metab Disord. 2008;32:315-21.
- 43. St Jeor ST, Brunner RL, Harrington ME, et al. A classification system to evaluate weight maintainers, gainers, and losers. J Am Diet Assoc. 1997;97:481-8.
- 44. Peirson L, Fitzpatrick-Lewis D, Ali MU, et al. Prevention of overweight/obesity in adult populations: a systematic review with meta-analyses. Ottawa: Canadian Task Force on Preventive Health Care; 2014. <u>http://canadiantaskforce.ca/files/guidelines/2015-</u> <u>obesity-adults-treatment-systematic-review-en.pdf</u>
- 45. Fabbrini E, Yoshino J, Yoshino M, et al. Metabolically normal obese people are protected from adverse effects following weight gain. J Clin Invest. 2015;125:787-98.
- Durward CM, Hartman TJ, Nickols-Richardson SM. All-cause mortality of metabolically healthy obese individuals in NHANES III. J Obes. 2012;2012:460321. doi: 10.1155/2012/460321.
- 47. van Genugten L, van Empelen P, Oenema A. From weight management goals to action planning: identification of a logical sequence from goals to actions and underlying determinants. J Hum Nutr Diet. 2012;25:354-64.

- 48. Fildes A, Charlton J, Rudisill C, Littlejohns P, Prevost AT, Gulliford MC. Probability of an obese person attaining normal body weight: cohort study using electronic health records. Am J Public Health. 2015;105:e54-9.
- 49. Forster JL, Jeffery RW, Schmid TL, Kramer FM. Preventing weight gain in adults: a pound of prevention. Health Psychol. 1988;7:515-25.
- 50. Jeffery RW, French SA. Preventing weight gain in adults: design, methods and one year results from the Pound of Prevention study. Int J Obes Relat Metab Disord. 1997;21:457-64.
- Jeffery RW, French SA. Preventing weight gain in adults: the pound of prevention study. Am J Public Health. 1999;89:747-51.
- 52. Sherwood NE, Jeffery RW, French SA, Hannan PF, Murrary DM. Predictors of weight gain in the Pound of Prevention study. Int J Obes Relat Metab Disord. 2000;24:395-403.
- 53. Levine MD, Klem ML, Kalarchian MA, et al. Weight gain prevention among women. Obesity (Silver Spring). 2007;15:1267-77.
- 54. Koop CE. On your way to fitness: a practical guide to achieving and maintain healthy weight and physical fitness. Connecticut Marketing Association, Inc.; Wilton, CT: 1995.
- 55. ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing weight gain: one-year results of a randomized lifestyle intervention. Am J Prev Med. 2009;37:270-7.
- 56. ter Bogt NC, Milder IE, Bemelmans WJ, et al. Changes in lifestyle habits after counselling by nurse practitioners: 1-year results of the Groningen Overweight and Lifestyle study. Public Health Nutr. 2011;14:995-1000.
- 57. ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing weight gain by lifestyle intervention in a general practice setting: three-year results of a randomized controlled trial. Arch Intern Med. 2011; 171:306-13.
- 58. Bennett GG, Foley P, Levine E, et al. Behavioral treatment for weight gain prevention among black women in primary care practice: a randomized clinical trial. JAMA Intern Med. 2013;173:1770-7.

- 59. Wing RR, Tate D, Espeland M, et al. Weight gain prevention in young adults: design of the study of novel approaches to weight gain prevention (SNAP) randomized controlled trial. BMC Public Health. 2013:13;300. doi: 10.1186/1471-2458-13-300.
- 60. Gill TP. Key issues in the prevention of obesity. Br Med Bull. 1997;53:359-88.
- 61. Jensen MD, Ryan DH, Apovian CM, et al. 2013 AHA/ACC/TOS guideline for the management of overweight and obesity in adults: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines and The Obesity Society. J Am Coll Cardiol. 2014;63:2985-3023.
- 62. Seagle HM, Strain GW, Makris A, Reeves, RS, American Dietetic Association. Position of the American Dietetic Association: weight management. J Am Diet Assoc. 2009;10:330-46.
- 63. Holmes AL, Sanderson B, Maisiak R, Brown A, Bittner V. Dietitian services are associated with improved patient outcomes and the MEDFICTS dietary assessment questionnaire is a suitable outcome measure in cardiac rehabilitation. J Am Diet Assoc. 2005;105:1533-40.
- 64. Wolf AM, Conaway MR, Crowther JQ, et al. Translating lifestyle intervention to practice in obese patients with type 2 diabetes: Improving Control with Activity and Nutrition (ICAN) study. Diabetes Care. 2004;27:1570-6.
- Wolf AM, Siadaty M, Yaeger B, et al. Effects of lifestyle intervention on healthcare costs: Improving Control with Activity and Nutrition (ICAN) study. J Am Diet Assoc. 2007;107:1365-73.
- 66. Eilat-Adar S, Eldar A, Goldbourt U. Association of intentional changes in body weight with coronary heart disease event rates in overweight subjects who have an additional coronary risk factor. Am J Epidemiol. 2005;161:352-8.
- 67. Sikand G, Kashyap ML, Wong ND, Hsu JC. Dietitian intervention improves lipid values and saves medication costs in men with combined hyperlipidemia and a history of niacin noncompliance. J Am Diet Assoc. 2000;100:218-24.

- 68. Lavie CJ, Morshedi-Meibodi A, Milani RV. Impact of cardiac rehabilitation on coronary risk factors, inflammation, and the metabolic syndrome in obese coronary patients. J Cardiometab Syndr. 2008;3:136-40.
- 69. Bradley DW, Murphy G, Snetselaar LG, Myers EF, Qualls LG. The incremental value of medical nutrition therapy in weight management. Manag Care. 2013;22:40-5.
- 70. Parker AR, Byham-Gray L, Denmark R. Winkle PJ. The effect of medical nutrition therapy by a registered dietitian nutritionist in patients with prediabetes participating in randomized controlled clinical research trial. J Acad Nutr Diet. 2014;114:1739-48.
- 71. Dutton GR, Phillips JM. Is there a role for lay peer coaches in treatment of obesity in primary care? Obesity: Open Access. 2015;1. doi http://dx.doi. org/10.16966/2380-5528.e102
- 72. Dutton GR, Phillips JM, Kukkamalla M, Cherrington AL, Safford MM. Pilot study evaluating the feasibility and initial outcomes of a primary care weight loss intervention with peer coaches. Diabetes Educ. 2015;41:361-8.
- Table TM, Wing RR. A randomized controlled pilot study testing three types of health coaches for obesity treatment: professional, peer, and mentor. Obesity (Silver Spring). 2013;21:928-34.
- 74. Katula JA, Vitolins MZ, Morgan TM, et al. The Healthy Living Partnerships to Prevent Diabetes study: 2-year outcomes of a randomized controlled trial. Am J Prev Med. 2013;44:S324-32.
- 75. West DS, Bursac Z, Cornell CE, et al. Lay health educators translate a weight-loss intervention in senior centers: a randomized controlled trial. Am J Prev Med. 2011;41:385-91.
- 76. Parikh P, Simon EP, Fei K, et al. Results of a pilot diabetes prevention intervention in East Harlem, New York City: Project HEED. Am J Public Health. 2010;100:S232-9.
- 77. Tiggerman M, Rothblum ED. Gender differences in internal beliefs about weight and negative attitudes towards self and others. Pyschol Women Q. 1997;21:581-93.

Table 2.1 Studies of weight gain prevention in adults.

Authors	Population	Intervention	Duration	Results
Forster JL, Jeffery RW,	Men and women	Treatment group: Received	1 year	Mean weight change was
Schmid TL, Kramer FM.	(n=219)	monthly newsletters relating to		significantly greater in the
Preventing weight gain in		weight management and		treatment group compared to
adults: a pound of	25-74 years	participated in a financial		the control group.
prevention. Health Psychol.		incentive system; offered an		
1988;7:515-25.	Normal weight (<115%	optional 4-session education		Treatment had a stronger effect
	of ideal body weight)	course in the sixth month of the		in men, older individuals (51+
A Pound of Prevention		program.		years) and nonsmokers.
		No-contact control.		
Jeffery RW, French SA.	Men and women	Education only: Received	3 years	Weight gain did not differ
Preventing weight gain in	(n=809)	monthly newsletters and		significantly by treatment
adults: design, methods and		optional low-cost intervention		group.
one year results from the	20-45 years	activities every 6 months.		
Pound of Prevention study.				Individuals in intervention
Int J Obes Relat Metab	No weight criteria	Education + incentive: Received		groups reported favorable
Disord. 1997;21:457-64.		the same newsletters plus a		changes over time in frequency
		lottery incentive for		of weighing and healthy dieting
Jeffery RW, French SA.		participation.		practices compared to those in
Preventing weight gain in				the control group.
adults: the pound of		No-contact control.		
prevention study. Am J Public				
Health. 1999;89:747-51.				

Pound of Prevention

Table 2.1 (cont.)

Levine MD, Klem ML, Kalarchian MA, et al. Weight gain prevention among	Women (n=284) 25-44 years	Clinic-based group: Attended 15 group meetings over a 24-month period.	2 year intervention	Intervention had no significant effect on weight over time.
women. Obesity (Silver Spring). 2007;15:1267-77.	Body mass index <30 kg/m²	Correspondence course group: Received 15 lessons by mail, identical in content to the clinic- based group, over a 24-month period.	1 year follow- up	Predictors of weight gain prevention: older age, low hunger at baseline, not on a diet at baseline, increased dietary restraint and decreased dietary disinhibition over time.
		Information-only control: Received a booklet containing information about the benefits of weight maintenance, low-fat eating and regular physical activity.		
ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing	Men and women (n=457)	General practitioner (GP) group: Usual care.	3 years	Mean weight loss was significantly greater in NP men compared to GP men at one
weight gain: one-year results of a randomized lifestyle	40-70 years	Nurse practitioner (NP) group: Lifestyle counseling consisting of		year; there were no significant differences between women in
intervention. Am J Prev Med. 2009;37:270-7.	Body mass index 25-40 kg/m ²	4 individual visits and 1 telephone feedback session		the NP and GP groups.
ter Bogt NC, Milder IE, Bemelmans WJ, et al. Changes in lifestyle habits after counselling by nurse practitioners: 1-year results of the Groningen Overweight and Lifestyle study. Public Health Nutr. 2011;14:995- 1000.	Hypertension and/or dyslipidemia	(year 1) and 1 individual visit and 2 feedback sessions (years 2 and 3).		After three years, there were no significant differences in weight change between groups.

Table 2.1 (cont.)

ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing weight gain by lifestyle intervention in a general practice setting: three-year results of a randomized controlled trial. Arch Intern Med. 2011; 171:306-13.

Groningen Overweight and

Lifestyle (GOAL) Study

Bennett GG, Foley P, Levine	African American	Intervention group: Received	1 year	Mean weight and body mass
E, et al. Behavioral treatment	women (n=194)	tailored behavior change goals,	intervention	index changes were significantly
for weight gain prevention		weekly self-monitoring via		larger in the intervention group
among black women in	25-44 years	interactive voice response,	6 month	compared to the control group
primary care practice: a		monthly counseling calls,	follow-up	at one year and these changes
randomized clinical trial.	Body mass index 25-	tailored skills training materials		were sustained at 18 months.
JAMA Intern Med.	34.9 kg/m ²	and a gym membership.		
2013;173:1770-7.				No differences between groups
		Usual care control group:		in changes in waist
Shape Program		Received newsletters that		circumference, blood pressure,
		covered general wellness topics		glucose or lipid levels.
		every 6 months.		
Wing RR, Tate D, Espeland M,	Men and women	Self-regulation + small behavior	1 year	Not yet published.
et al. Weight gain prevention	(n=600)	changes: Attended 8 weekly	intervention	
in young adults: design of the		group sessions and 2 monthly		
study of novel approaches to	18-35 years	sessions (year 1) and 2 annual 4-	3 year	
weight gain prevention		week refresher courses (years 2	average	
(SNAP) randomized	Body mass index 21-30	and 3). Taught to make small	follow-up	
controlled trial. BMC Public	kg/m ²	daily changes (~100 kcal) and to		

Health. 2013:13;300. doi: 10.1186/1471-2458-13-300.

Study of Novel Approaches to Weight Gain Prevention (SNAP) accumulate 2000 additional steps per day.

Self-regulation + large behavior changes: Attended 8 weekly group sessions and 2 monthly sessions (year 1) and 2 annual 4week refresher courses (years 2 and 3). Taught to create a weight loss buffer of 5-10 pounds once per year to protect against weight gain.

Self-guided behavior change (control): Received 1 face-toface group session and an overview of Large and Small Changes approaches. Participants self-selected their desired approach.
In this chapter, findings from preliminary research are presented. Completed studies focus on weight loss and weight loss maintenance in women, and these findings have provided the foundation for the current dissertation research. The first study discusses results from an original 18-week weight loss intervention (1), while the second presents findings from a follow-up qualitative study of participants that completed the original weight loss trial in its entirety (2).

The first study describes a weight loss intervention that was designed to address issues related to energy restriction, discretionary calories and the inclusion of portion-controlled sweet and savory snacks within an energy-restricted diet. It was hypothesized that the inclusion of, rather than total elimination of, portion-controlled sweet and savory snacks would result in better adherence to the energy-restricted diet while promoting beneficial changes in body weight, blood pressure and other metabolic outcomes. While findings from this initial weight-loss intervention demonstrated that weight loss and other favorable metabolic changes are possible while consuming an energy-restricted diet that includes portion-controlled snacks (1), this study did not provide clarity about emotional and/or cognitive factors that may influence weight loss success or why some individuals were more successful in achieving weight loss goals when compared to others.

A qualitative follow-up study was conducted to gain a more comprehensive and deeper understanding of facilitators and barriers related to weight loss and weight loss maintenance in women participating in the initial intervention. Focus groups revealed that planning ahead, portion control, basic nutrition education, self-motivation, accountability to others, social support and mindfulness and awareness of food choices are major facilitators to weight loss and weight loss maintenance success. Major barriers included lack of accountability and social support, life transitions, environmental pressures, internal factors and health status changes (2) Attention to these barriers and facilitators has been given in developing the current weight gain prevention trial described in Chapter 4.

Collectively, these two studies demonstrate that flexible dietary approaches can promote beneficial changes in body weight; however, maintenance of weight loss remains a challenge for many women. As a majority of female participants were unable to maintain weight loss over time, these studies support the need for weight management strategies to prevent inappropriate weight gain over time. In combination with findings from other research studies, weight gain prevention may be a more sustainable and appropriate target than weight loss alone. Following the presentation of this preliminary work, Chapter 4 further describes the methodology for the current dissertation research.

References

- This article previously appeared in its entirety as Sharon M. Nickols-Richardson, Kathryn E. Piehowski, Catherine J. Metzgar, Debra L. Miller and Amy G. Preston. Changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr Res Pract. 2014;8(6):695-704; doi:10.4162/nrp.2014.8.6.695.[Epub ahead of print]. This is an Open Access article distributed under the terms of the Creative Commons Attribution Non-Commercial License (<u>http://creativecommons.org/licenses/by-nc/3.0/</u>) which permits unrestricted non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.
- 2. This article previously appeared in its entirety as Metzgar CJ, Preston AG, Miller DL. Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. *J Hum Nutr Diet.* 2014 Sept 18. doi: 10.1111/jhn.12273. [Epub ahead of print]. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is noncommercial and no modifications or adaptations are made.

Changes in body weight, blood pressure and selected metabolic biomarkers with an energyrestricted diet including twice daily sweet snacks and once daily sugar-free beverage¹

BACKGROUND/OBJECTIVES: The type of sweet snack incorporated into an energy-restricted diet (ERD) may produce differential effects on metabolic improvements associated with body weight (BW) loss. This study compared effects of incorporating either twice daily energy-controlled dark chocolate snacks plus once daily sugar-free cocoa beverage (DC) to non-chocolate snacks plus sugar-free non-cocoa beverage (NC) into an ERD on BW loss and metabolic outcomes.

MATERIALS/METHODS: In an 18-week randomized comparative trial, 60 overweight/obese premenopausal women were assigned to DC (n = 30) or NC group (n = 30). Dietary intake was measured at baseline and week 18, and BW, anthropometrics, blood pressure (BP) and serum glucose, insulin and lipid concentrations were measured at baseline, and weeks 6, 12 and 18. Data were analyzed using repeated measures ANOVA.

RESULTS: Using intention-to-treat analysis, women in DC and NC groups reduced energy intake (both P < 0.001) and lost 4.4 ± 0.6 kg and 5.0 ± 0.9 kg (both P < 0.001), respectively. Both groups lowered systolic and diastolic BP [DC = 2.7 (P < 0.05), 2.7 (P < 0.01); NC = 3.4 (P < 0.01), 4.2 (P < 0.01) mmHg, respectively]. Glucose and insulin concentrations decreased by 0.72 mmol/L (P < 0.01) and 13.20 pmol/L (P < 0.01) in DC group and by 0.83 mmol/L (P < 0.001) and 13.20 pmol/L (P < 0.01) in DC group. Total cholesterol increased in NC group (P < 0.05), with no significant lipid changes in DC group. There were no significant differences in biomarker outcomes between groups.

CONCLUSIONS: Overweight/obese premenopausal women following an 18-week ERD that included either DC or NC sweet snack and sugar-free beverage lost equivalent amounts of BW and improved BP measurements and glucose and insulin concentrations.

¹This article previously appeared in its entirety as Sharon M. Nickols-Richardson, Kathryn E. Piehowski, Catherine J. Metzgar, Debra L. Miller and Amy G. Preston. Changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr Res Pract. 2014;8(6):695-704; doi:10.4162/nrp.2014.8.6.695.[Epub ahead of print]. This is an Open Access article distributed under the terms of the Creative Commons Attribution Non-Commercial License (<u>http://creativecommons.org/licenses/by-nc/3.0/</u>) which permits unrestricted non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

INTRODUCTION

Overweight and obesity have become major public health concerns as 67 to 75% of adults in the United States (U.S.) are now affected [1]. Individuals with excess body weight (BW) spend 30% more on healthcare than those of normal BW, due to the increased incidence of comorbidities such as hypertension, diabetes and heart disease [2]. To improve current health status and prevent future complications, the primary treatment for these individuals is reduction of excess adiposity through moderate BW loss [3]. The key dietary objective for inducing BW loss is a reduction in total daily energy intake below energy needs [4].

Many weight-loss diets and diet programs restrict all high-fat and/or high-sugar snack foods [5,6,7,8]. Allowing individuals to consume snacks that are normally enjoyed in energyand portion-controlled amounts as part of an energy-restricted diet (ERD) may make adherence easier and potentially increase diet satisfaction, because habitual eating patterns are emphasized, and there is less dramatic alteration in food choices [9]. Chocolate is one of the most commonly liked and widely consumed sweet snacks among women in the U.S. and around the world [10,11], with approximately one-half of women reporting weekly consumption [12]. In addition, U.S. women are more likely than men to consume sweet foods such as ice cream, pastries and non-chocolate candy on a regular basis [11].

Chocolate as a sweet snack food is of particular interest, due to its volume of consumption [13], likeability [10] and pleasing sensory properties [12]. Epidemiologic studies suggest that chocolate intake is inversely related to body mass index (BMI) [14] and reduced risk of low high-density lipoprotein-cholesterol (HDL-C) and other metabolic syndrome indicators [13,15]. Further, previous studies suggest that consumption of chocolate and cocoa, specifically dark chocolate, may have beneficial effects on blood pressure (BP) [16,17,18,19,20,21], fasting blood glucose [20,22], insulin sensitivity [16,23,24] and blood lipids [17,19,25,26]. Cocoa is rich in minerals and phytonutrients, namely flavanols, including catechin, epicatechin and proanthocyanidins (PACs), and due to the higher cocoa content, dark chocolate may confer the greatest metabolic benefits when compared to milk or white chocolate [27,28].

Dark chocolate is commonly regarded as an energy-dense food [29] and excess consumption of any energy-dense food may have adverse metabolic effects, including weight gain. Therefore, women attempting BW loss often withhold chocolate and other sweet snacks from their diet. While several short-term (i.e., 2 to 8 weeks) studies have examined changes in BW following consumption of dark chocolate [30], only one study has compared the effects of dark chocolate against non-chocolate intake specifically on changes in BW and body composition during energy restriction in overweight/obese women [9]. In this feasibility study, inclusion of a dark chocolate or non-chocolate sweet snack as part of an ERD resulted in losses in BW, fat mass (FM) and body fat percentage (BF%), with no significant differences between the two snack groups. The sample size of this pilot study was small, and outcomes were limited to body composition without further exploration of biomarkers of metabolic health [9].

The aim of the current 18-week randomized intervention was to compare effects of incorporating twice daily dark chocolate snacks plus once daily sugar-free cocoa beverage (DC group) to twice daily non-chocolate snacks plus once daily sugar-free non-cocoa beverage (NC group) into an ERD with a typical macronutrient distribution [31] on BW loss and metabolic outcomes in free-living premenopausal women with overweight/obesity. Outcome measures included estimated energy intake, BW, anthropometric and BP measurements as well as serum glucose, insulin and lipid concentrations. Based on emerging evidence that dark chocolate and cocoa may modulate obesity, it was hypothesized that inclusion of dark chocolate and cocoa into an ERD would result in a significantly greater decrease in BW and more favorable improvements in metabolic indicators of health, compared to inclusion of non-chocolate and non-cocoa products.

SUBJECTS AND METHODS

Participants

Participants were recruited from central Pennsylvania, U.S., by word-of-mouth, newsletter and newspaper advertisements, electronic-mail notices and flyers posted in the local community. Two-hundred three women provided verbal consent for an initial telephone screening and were assessed for study eligibility; 85 were enrolled in the study. Due to a

relatively lengthy recruitment interval, 25 women withdrew before baseline testing was completed for a final sample size of 60 women (Figure 3.1).

The current dietary intervention included women ages 25 to 45 years with a BMI of \geq 25.0 and < 43.0 kg/m². Women were moderately physically active (\leq 5 hours of planned exercise/week), eumenorrheic (\geq 8 menstrual cycles/year) and of self-reported stable BW (< 5% change in BW for at least six months before study participation). Further inclusion criteria included a score of < 50 on the Zung Self-Rating Depression Scale/Status Inventory and no intolerance, aversion or allergy to chocolate. Exclusion criteria included women who currently smoked, were pregnant or attempting to become pregnant, had a hysterectomy and/or ovariectomy without hormone replacement therapy and those who used oral contraceptives for < 2 years in duration (if used). Women who used medications, including steroid or thyroid hormones, bisphosphonates, anticonvulsants and glucocorticoids, or consumed \geq 40 grams of chocolate per day (i.e., equivalent of one standard chocolate bar or more/day) were also excluded. All study participants underwent medical examinations by their personal healthcare providers to obtain measured and reliable values for height, BW and BMI to ascertain inclusion criteria was met.

Written informed consent was provided by all participants before entry into the study. The Institutional Review Board (IRB) for Research Involving Human Subjects at The Pennsylvania State University (PSU; University Park, PA, U.S.) conducted a full review of the study procedures and approved the study protocol (PSU-IRB#29543).

Study design

This was an 18-week randomized, parallel-arm, comparative dietary intervention in which participants were enrolled in two cohorts (July-December 2009 and March-July 2010). After enrollment of each cohort, women were stratified by baseline age, BMI and physical activity and then randomly assigned to either the DC group (n = 30) or the NC group (n = 30).

Dietary intervention

Participants in both groups followed an ERD with a macronutrient composition of 50% carbohydrate, 30% fat and 20% protein designed to induce approximately 0.91 kg of BW loss per week by consuming 2,092 fewer kJ per day than required for energy balance. Baseline energy levels were set between 5,439 and 7,531 kJ per day as determined using the Harris-Benedict equation [32].

Women in both groups were administered portion-controlled and energy-matched snacks and beverages. Women in the DC group consumed one, 236 mL sugar-free natural cocoa beverage (The Hershey Company, Hershey, PA, U.S.) each day (272 kJ/day) and one 1.45 oz dark chocolate tasting square (Hershey's® Extra Dark® dark chocolate, The Hershey Company) at two intervals each day (377 kJ/day). Women in the NC group drank one, 236 mL sugar-free cocoa-free vanilla beverage (The Hershey Company; 272 kJ/day) each day and consumed two non-chocolate sweet snacks (fruit-flavored licorice stick; The Hershey Company) each day (377 kJ/day) at the same daily intervals as the DC group. Participants in both groups were instructed to not consume additional cocoa or chocolate products throughout the 18-week intervention beyond the snack and beverage assignments. Women in the DC group consumed 270 mg of flavanols (PACs 1-10) per day from dark chocolate snacks and the sugar-free cocoa beverage [16,17], and women in the NC group consumed 0 mg of flavanols per day from non-chocolate snacks and the sugar-free non-cocoa beverage.

A registered dietitian educated participants on how to follow the ERD, which was based on a food exchange system. Women were assigned a certain number of servings from each of six exchange groups to promote a flexible dietary approach that approximated usual intake except for energy restriction and inclusion of assigned beverages and twice daily snacks. Three main principles were emphasized, including portion control [33,34], planning ahead for meals [35] and consuming vegetables to satisfy an acute hunger sensation [36]. Handouts that contained food options, dietary patterns and individualized meal plans specific to energy levels were provided.

All women attended weekly nutrition education classes that covered topics such as general nutrition information, dining in restaurants, food selection, food preparation and recipe modification. Problem-solving and motivational concerns also were discussed and addressed.

Education sessions were specific to DC or NC group; however, topics were the same for both groups, and one registered dietitian led all of the education sessions to maintain consistency between groups. Snacks and beverage mixes were dispensed at education sessions, and compliance with snack and beverage intervention was assessed by participant self-report and concurrent investigator-conducted snack counts; compliance was defined as intake of \geq 85% of weekly snacks and beverages based on group assignment. Upon completion of the study, participants received monetary compensation of \$80 (U.S.).

Outcome measures

Dietary intake was evaluated at baseline and week 18. Anthropometric, physical activity and BP measurements were completed and whole blood samples were collected at baseline, week 6, week 12 and week 18.

Dietary intake assessment

Dietary intake was estimated using 4-day food records. Women recorded all foods and beverages consumed on three non-consecutive weekdays and one weekend day in the week before measurement sessions at baseline and week 18. Handouts containing pictures of standard serving sizes of different foods and beverages were provided to aid in recording intake. Food records were evaluated using Food Processor® dietary analysis software (version 10.6.0, 2010, *esha* Research, Salem, OR, U.S.) for estimated average daily intake of total energy (kJ); carbohydrate, fat, protein and alcohol (% of total kJ); total sugar (g), fiber (g), saturated fat (g) and cholesterol (g); and sodium (mg).

Anthropometric measurements

Standing height (cm) was measured using a stadiometer (Seca 700, Hamburg, Germany), and BW (kg) and BF% were measured using an electronic scale (410GS, Tanita Corporation, Arlington Heights, IL, U.S.); height and BW measurements were used to calculate BMI for each participant. Using a spring-calibrated measuring tape (Gulik II, Country Technology, Gay Mills, WI, U.S.), two measurements each of the waist at the narrowest point above the belly button

and hips at the widest part of the buttocks were taken to the nearest 0.1 cm and averaged. For all measurements, women were dressed in lightweight clothing without shoes.

Physical activity

Physical activity was estimated using the Stanford 7-day Physical Activity Recall Scale [37]. For seven consecutive days before a measurement session, participants recorded number of hours slept, spent in front of a television or computer screen and engaged in moderate, hard and very hard activity. Total hours of moderate, hard and very hard activity were summed from the recall scale and divided by seven to estimate hours of physical activity per day.

Blood pressure

Seated systolic and diastolic BP (mmHg) was measured by a registered nurse using a standard sphygmomanometer (Baumanometer[®] Desk Model, Copiague, NY, U.S.). Two BP measurements were recorded with a 2- to 3-minute rest period between readings; values were averaged.

Sample collection

Venous blood samples were obtained by a registered nurse between the hours 0700-0930 after a 12-hour fast. Samples were centrifuged at 810 × P for 12 minutes, after clotting. Serum was pipetted into cryovials and stored at -80°C until completion of bioassays.

Metabolic profile including serum glucose, insulin and lipids

Serum glucose (Kit #1070, Stanbio Labs, Boerne, TX, U.S.) was measured (mmol/L) using ultra-violet (UV) spectrophotometry (version 3.0, Simple Reads Software, Varian, Santa Clara, CA, U.S.), and serum insulin (Catalog #IS130D, CalBiotech, Spring Valley, CA, U.S.) was measured (pmol/L) using enzyme-linked immunosorbent assay (GEN5 version 1.10, Epoch, BioTek, Winooski, VT, U.S.). Insulin resistance was estimated by homeostasis model of assessmentinsulin resistance (HOMA-IR) using the formula: fasting insulin concentration (μ IU/mL) × fasting glucose concentration (mg/dL) × 0.0555/22.5 [38].

Serum total cholesterol (mmol/L), HDL-C (mmol/L) and triacylglycerides (mmol/L) concentrations (Kits #1010, #0599, and #2100, respectively, Stanbio Labs) were measured using UV spectrophotometry (Varian). Low-density lipoprotein-cholesterol (LDL-C) was calculated using the equation: LDL-C = total cholesterol-HDL-C-(triacylglycerides/5) [39].

All serum samples were analyzed in duplicate. Intra-assay coefficients of variation (CVs) for glucose and insulin were 7.4 and 6.0%, respectively. Intra-assay CVs for serum total cholesterol, HDL-C and triacylglycerides were 6.0, 5.9 and 7.9%, respectively.

Statistical analyses

Using BW change from baseline to week 18 as the primary outcome, 21 participants per group were required to detect a treatment difference with 80% power when using t-tests and a 2-sided type I error of 5%. Using data from the 60 women who completed baseline measurements, data were first analyzed using intention-to-treat model. The nine women who did not complete the study (i.e., non-completers) were included in intention-to-treat analyses by replacing missing data with the last available measurement value. A secondary efficacy analysis was conducted by including only the 51 women who completed the 18-week intervention.

Data are presented as means ± SEM unless otherwise indicated. Differences between the two cohorts were analyzed using independent *t*-tests. Differences in characteristics at baseline between DC and NC groups and between study-completers and those who withdrew also were analyzed using independent *t*-tests.

Using intention-to-treat data, a 2 × 4 ANOVA with repeated measures on the time factor was performed to assess differences in anthropometrics and BP measurements and metabolic indicators between DC and NC groups over four intervals. The interaction of group (Treatment) by interval (Time) also was assessed. Data were analyzed using the Statistical Package for the Social Sciences (version 17.0, 2008, SPSS Inc., Chicago, IL, U.S.). All tests were two-sided with significance set at P < 0.05.

RESULTS

Statistically significant differences were not observed between the two cohorts of women (July-December, March-July) in baseline characteristics or estimated dietary intakes, with the exception of self-reported physical activity. Therefore, data from both cohorts were combined and used in analyses. Sixty women (one Native American, two African American and 57 Caucasian), with a mean \pm SEM age of 35.9 ± 0.8 years and BMI of 31.0 ± 0.6 kg/m² began the intervention. Fifty-one of the women (85%) completed the intervention with no difference between DC and NC groups in discontinuation rate. No statistically significant differences in race, age, height, BW, BMI, waist and hip circumferences and physical activity between the DC (n = 30) and NC (n = 30) groups were found at baseline. There were no differences in these same characteristics at baseline for women randomly assigned to DC or NC group who completed (n = 51) the study compared to those who withdrew (n = 9).

Snack compliance and class attendance

Snack and beverage compliance was 90 and 90%, respectively, for the DC group and 92 and 94%, respectively, for the NC group. Attendance at nutrition education classes was 74 and 75% for the DC group and NC group, respectively. Neither snack and beverage compliance nor class attendance differed between groups.

Intention-to-treat analysis

Dietary intake assessment

Fifty-nine women completed 4-day food records at baseline. Table 3.1 displays estimated dietary intake of these participants at baseline and week 18 and changes over time. Within the DC and NC groups, women decreased estimated total energy, total sugar, saturated fat and cholesterol intakes, supporting that participants were successful in complying with energy restriction. Women in the DC group also reduced estimated sodium intake, and women in the NC group increased estimated dietary protein intake. Significant differences in estimated dietary intake variables were not found between groups at baseline and week 18 or in changes in nutrient intakes over time.

Anthropometric measurements

Women in the DC group lost 5.3% (P < 0.001) of BW, while women in the NC group lost 5.9% (P < 0.001) of BW from baseline to week 18 (Table 3.1). The rate of BW change over time did not differ between groups. Women within both groups significantly reduced BMI, waist and hip circumferences and BF% over time (Table 3.2), again suggesting compliance with energy restriction. Changes in these anthropometric measurements did not differ between groups at any interval or over time.

Physical activity

Self-reported physical activity (hr/day) was greater in the first cohort of women compared to second cohort (P < 0.01) at baseline. Therefore, a physical activity change variable was created for each treatment group- (DC and NC) by-cohort (1 and 2). The change variable was then compared among these four groups using ANOVA. No significant difference in the change in physical activity from baseline to week 18 among treatment group-by-cohort was observed. The effect of time on physical activity was assessed within each group using repeated measures ANOVA. Self-reported physical activity did not differ within the four treatment groupby-cohort categories or between treatment groups over time.

Blood pressure

From baseline to week 18, women in both DC and NC groups experienced significant reductions in systolic and diastolic BP (Table 3.3). For women in the DC group, the significant change in diastolic BP occurred by week 12. For women in the NC group, significant changes in systolic and diastolic BP occurred by week 12. Differences in BP measurements between groups were not found, and the changes within DC and NC groups over time were not different between groups.

Metabolic profile

Women in the DC group had decreases in serum glucose concentration by week 6 (7.5% \downarrow , *P* < 0.001), and at week 12 (5.3% \downarrow , *P* < 0.01) and week 18 (12.9% \downarrow , *P* < 0.001) compared

to baseline (Table 3.3). Women in the NC group also experienced decreases in serum glucose by week 6 (4.9% \downarrow , *P* < 0.05), at week 12 (4.0% \downarrow , *P* < 0.05) and week 18 (15.2% \downarrow , *P* < 0.001) compared to baseline. Serum glucose concentrations did not differ between groups at any interval, and the change over time within each group did not differ between groups.

Serum insulin concentration decreased in the DC group (26.4% \downarrow , *P* < 0.01) and in the NC group (26.8% \downarrow , *P* < 0.01) from baseline to week 18 (Table 3.3). Differences between groups at any interval or for the change over time between groups in serum insulin were not statistically significant. HOMA-IR followed a pattern similar to serum insulin for both DC and NC groups.

Serum lipid concentrations did not differ between groups at any interval (Table 3.3). In the DC group, serum total and LDL-C increased by week 12 but returned to baseline levels at week 18. HDL-C decreased by week 6 but returned to baseline level at week 18 for women in the DC group. Women in the NC group had an increase in serum total cholesterol from baseline that persisted to week 18 (8.5% \uparrow , *P* < 0.05). Women in the NC group also had a decrease in HDL-C from baseline to week 6 (9.5% \downarrow , *P* < 0.001) that returned to baseline level at week 18. Changes over time in serum lipids were not different between DC and NC groups.

Efficacy analysis

Dietary intake assessment

For those participants who completed the study (n = 51), women in the DC (n = 26) and NC (n = 25) groups reduced estimated energy intake by 1,900 kJ/day (P < 0.001) and 2,213 kJ/day (P < 0.001), respectively. Macronutrient intake did not change in the DC group, while percent of energy from protein increased for women in the NC group (P < 0.01). Changes in total sugar, fiber, saturated fat, cholesterol and sodium intakes within and between groups over time were similar to results previously reported using the intention-to-treat analysis.

Anthropometric and blood pressure measurements and metabolic profile

Due to the low number of dropouts from each group, changes in anthropometric and BP measurements and glucose, insulin, HOMA-IR and lipid concentrations in the efficacy analysis

were greater but had similar statistical significance to changes observed in intention-to-treat analysis. From baseline to week 18, BW decreased by 5.1 ± 1.4 kg (P < 0.001) and 5.9 ± 0.9 kg (P < 0.001) in the DC group and NC group, respectively. In the DC group, BMI, waist circumference, hip circumference and BF% decreased by 1.9 ± 0.2 kg/m² (P < 0.001), 5.8 ± 0.9 cm (P < 0.001), 6.0 ± 0.7 cm (P < 0.001) and $3.7 \pm 0.4\%$ (P < 0.001), respectively, from baseline to week 18. In the NC group, BMI, waist circumference, hip circumference and BF% decreased by 2.2 ± 0.3 kg/m² (P < 0.001), 6.1 ± 1.0 cm (P < 0.001), 5.8 ± 0.7 cm (P < 0.001) and $3.3 \pm 0.6\%$ (P < 0.001), respectively, from baseline to week 18. These changes over time were not significantly different between DC and NC groups.

Systolic and diastolic BP, respectively, decreased by $3.6 \pm 1.4 \text{ mmHg}$ (P < 0.05) and $3.5 \pm 1.1 \text{ mmHg}$ (P < 0.01) and $5.3 \pm 1.4 \text{ mmHg}$ (P < 0.01) in the DC group and NC group, respectively, from baseline to week 18. Serum glucose and insulin concentrations and HOMA-IR, respectively, decreased by 14.7% (P < 0.001), 30.6% (P < 0.01) and 40.2% (P < 0.001) and by 17.9% (P < 0.001), 28.2% (P < 0.05) and 39.6% (P < 0.001) in the DC group and NC group, respectively, from baseline to week 18. Values for serum lipid concentrations for participants who completed the study by DC or NC group were within 15% of values for intention-to-treat analysis. Changes over time in BP measurements and metabolic biomarkers did not differ between groups.

DISCUSSION

Premenopausal women with overweight/obesity who followed an 18-week ERD that included twice daily dark chocolate or non-chocolate sweet snacks plus once daily sugar-free beverage were able to achieve an energy deficit, reduce BW and significantly improve BP and glucose and insulin concentrations. Participants in both groups were compliant with snack and beverage intake, as well as reducing overall energy intake as evidenced by improvements in BW and other anthropometric measurements. Although the hypothesis was not supported, and differential effects between the DC group and NC group were not found in this randomized comparative study, the two snack and beverage assignments were equally effective in promoting significant BW loss and improvements in metabolic parameters while following a

dietary approach that did not appreciably alter macronutrient composition of the habitual diet or completely eliminate sweet snacks. Because the current study did not include a non-snack and non-beverage control group, further evaluation is needed.

Golomb et al. [14] found an inverse relationship between frequency of chocolate consumption and BMI among nearly 1,000 adults, ages 20 to 85 years, even after adjusting for age, gender, physical activity, dietary components and energy intake. Using National Health and Nutrition Examination Survey, 1999-2004, data, O'Neil et al. [13] reported that chocolate consumers had significantly lower BW and waist circumference compared to non-consumers. Conversely, a prospective analysis by Greenberg and Buijsse [40] found more frequent consumption of chocolate to be significantly associated with greater weight gain over the long term. These epidemiological studies cannot draw causal inferences regarding effects of chocolate intake on BW or BMI but rather provide direction for future research to confirm these results. Further, these studies rely on self-reported data and not all distinguished the type of chocolate consumed. Randomized clinical trials exploring the impact of dark chocolate and cocoa on anthropometric measurements as the primary outcome measure are few; however, Desch et al. [41] observed a slight weight gain after three months of consuming 25 g of dark chocolate per day. Conversely, Taubert et al. [18] did not detect a change in BW after 18 weeks of daily consumption of 6.3 g of dark chocolate.

In an experimental study, Matsui et al. [42] demonstrated that cocoa intake for three weeks led to lower BW and white adipose tissue weight in male Wistar rats fed a high-fat diet compared to rats fed mimetic cocoa and high-fat diet. Cocoa consumption in these animals suppressed fatty acid synthase and other liver enzymes required for fatty acid synthesis. In addition, fatty acid binding protein and fatty acid synthase were lowered in white adipose tissue of cocoa-fed rats, suggesting altered lipid metabolism with cocoa intake in the presence of high dietary fat (-50% of total energy) [42]. Min et al. [43] cultured 3T3-L1 preadipocytes with cocoa polyphenol extract and found that signaling systems for cell proliferation were blunted. In a complementary 5-week whole animal study, Min et al. [43] further showed that C57BL/6N mice fed a high-fat diet and cocoa polyphenol extract had lesser BW gain and adiposity compared to mice fed only the high-fat diet. These experimental studies suggest biologically

plausible mechanisms by which cocoa may moderate BW and FM. Results of the current comparative trial are somewhat inconsistent with human epidemiologic and clinical trials [13,14,18,40,41], experimental animal [42,43] and cell culture [43] studies, in that women in the NC group also experienced benefits to BW and BF%. Discordant results are likely due to the design of the current study conducted in free-living women, where the total diet was not controlled and the lack of a control group. Nonetheless significant BW loss and improved BF% was achieved by incorporating preferred sweet snacks within an ERD.

Effects on BP, glucose, insulin, HOMA-IR, total cholesterol, HDL-C, triacylglycerides and LDL-C did not differ between the DC group and NC group. Several meta-analyses have indicated that consumption of 500 to 1,000 mg of cocoa flavanols per day results in acute (i.e., 2 to 12 weeks, with one 18-week trial) health benefits, including lowering of SBP [21,22,28] and DBP [21,24,28] as well as moderating insulin [24], HOMA-IR [22,24], total cholesterol [44] and LDL-C [22,24,44] and improving HDL-C [22,24], although one meta-analysis did not find any consistent effects on HDL-C [44]. One systematic review has reported no effect of cocoa intake on total cholesterol [22], and two reviews have indicated a lack of effect of cocoa on triacylglycerides [22,44]. Changes in the DC group were consistent with previous findings for SBP (-2.77 mmHg) [21] and DBP (-2.20 mmHg) [21], although mean differences for SBP and DBP were slightly better in the NC group. The change in HOMA-IR for the DC group was similar to the finding (-0.67) reported in the meta-analysis by Hooper et al. [24]. Although these benefits also were apparent for women in the NC group, the flavanol dosage for the DC group (270 mg) was less than the 500 to 1,000 mg used in previous studies. Blood lipids did not change or changed in expected directions when compared to other studies [16,20,45]. BW reduction has been shown to produce favorable changes in metabolic risk factors for disease [46,47]. Therefore, the weight loss achieved by women in the DC group may have overshadowed any differential effect or potentially additive effect of dark chocolate and cocoa intake on metabolic markers, given that comparable BW loss occurred in the NC group with similar metabolic outcomes.

This study is limited to premenopausal women of specific age and BMI ranges and cannot be generalized to others. The duration of 18 weeks was shorter than the ideal length of \geq 24 weeks for a short-term weight-loss study; however, 18 weeks was of sufficient duration to

detect glucose, insulin and blood lipid changes measured in the current study and this duration was consistent or longer than previous trials examining dark chocolate and cocoa. Nutrition education classes included in the current intervention make it difficult to distinguish between positive outcomes attributed to weekly visits with a registered dietitian from benefits of dietary components. The current study was not a metabolic feeding trial and consequently relied on self-reported dietary intakes that have previously been shown to be underreported in overweight/obese individuals [48]. The current intervention was conducted with free-living women to mimic a real world scenario, and positive changes in BW, FM and metabolic markers demonstrated compliance with the dietary intervention, including energy restriction. Long-term intervention studies are needed, and future studies should compare inclusion of cocoa-based sweet snacks and sugar-free cocoa beverages into an ERD against a control ERD that does not include snacks or beverages to make appropriate recommendations for weight management programs. While the lack of a control group limits the generalizability of results, the design of the current study was to compare a sweet, widely consumed snack to one that is not as widely consumed. However, future interventions should compare an ERD that includes sweet snacks and beverages as well as sugar-free snacks and beverages to an ERD that does not.

While emerging evidence from animal, human and *in vitro* studies suggests dark chocolate and cocoa may have beneficial effects on body weight and other anthropometric measures [30], the current study found that an 18-week ERD that included twice daily dark chocolate snacks plus once daily sugar-free cocoa beverage resulted in a magnitude of BW loss and changes in BP and metabolic markers that were comparable to an ERD that included a sweet snack without dark chocolate or cocoa and sugar-free cocoa-free beverage. Premenopausal women with overweight/obesity and without established hypertension or hyperglycemia may experience clinically significant improvements in BP and glucose and insulin concentrations with a moderate 6.5% decrease in BW facilitated by an ERD that includes two energy- and portion-controlled sweet snacks and one sugar-free beverage daily. Portion- and energy-controlled sweet snacks, including dark chocolate or non-chocolate snacks, may be included in diets intended for BW modification.

ACKNOWLEDGMENTS

Authors thank the participants for their engagement in this study and undergraduate student research assistants who contributed to participant recruitment and retention. This research was supported by a grant from The Hershey Company, including provision of dark chocolate and non-chocolate snacks and cocoa and non-cocoa beverage mixes in addition to Graduate Research Assistant support. Graduate Research Fellowships were provided through the American Dietetic Association, The American Association of Family and Consumer Sciences, and The Pennsylvania State University.

REFERENCES

- Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. JAMA.2012;307:491–497.
- Withrow D, Alter DA. The economic burden of obesity worldwide: a systematic review of the direct costs of obesity. Obes Rev. 2011;12:131–141.
- Hankey CR. Session 3 (joint with the British Dietetic Association): management of obesity: weight-loss interventions in the treatment of obesity. Proc Nutr Soc. 2010;69:34–38.
- Wu T, Gao X, Chen M, van Dam RM. Long-term effectiveness of diet-plus-exercise interventions vs. diet-only interventions for weight loss: a meta-analysis.Obes Rev. 2009;10:313–323.
- Sears B, Lawren B. Enter the Zone: A Dietary Road Map. New York (NY): Harper Collins; 1995.
- Ornish D. Eat More, Weigh Less: Dr. Dean Ornish's Program for Losing Weight Safely While Eating Abundantly. New York (NY): Harper Collins; 2001.
- 7. Atkins RC. Dr. Atkins' New Diet Revolution. New York (NY): Harper Collins; 2002.
- Agatston A. The South Beach Diet : The Delicious, Doctor-Designed, Foolproof Plan for Fast and Healthy Weight Loss. Emmaus (PA): Rodale; 2003.
- Piehowski KE, Preston AG, Miller DL, Nickols-Richardson SM. A reduced-calorie dietary pattern including a daily sweet snack promotes body weight reduction and body composition improvements in premenopausal women who are overweight and obese: a pilot study. J Am Diet Assoc. 2011;111:1198–1203.
- 10. Rozin P, Levine E, Stoess C. Chocolate craving and liking. Appetite.1991;17:199–212.
- 11. Zellner DA, Garriga-Trillo A, Rohm E, Centeno S, Parker S. Food liking and craving: a cross-cultural approach. Appetite. 1999;33:61–70.
- Lewis JR, Prince RL, Zhu K, Devine A, Thompson PL, Hodgson JM. Habitual chocolate intake and vascular disease: a prospective study of clinical outcomes in older women. Arch Intern Med. 2010;170:1857–1858.

- O'Neil CE, Fulgoni VL, 3rd, Nicklas TA. Candy consumption was not associated with body weight measures, risk factors for cardiovascular disease, or metabolic syndrome in US adults: NHANES 1999-2004. Nutr Res. 2011;31:122–130.
- 14. Golomb BA, Koperski S, White HL. Association between more frequent chocolate consumption and lower body mass index. Arch Intern Med.2012;172:519–521.
- 15. Khawaja O, Gaziano JM, Djoussé L. Chocolate and coronary heart disease: a systematic review. Curr Atheroscler Rep. 2011;13:447–452.
- 16. Grassi D, Lippi C, Necozione S, Desideri G, Ferri C. Short-term administration of dark chocolate is followed by a significant increase in insulin sensitivity and a decrease in blood pressure in healthy persons. Am J Clin Nutr.2005;81:611–614.
- Grassi D, Desideri G, Necozione S, Lippi C, Casale R, Properzi G, Blumberg JB, Ferri C. Blood pressure is reduced and insulin sensitivity increased in glucose-intolerant, hypertensive subjects after 15 days of consuming high-polyphenol dark chocolate. J Nutr. 2008;138:1671–1676.
- Taubert D, Roesen R, Lehmann C, Jung N, Schomig E. Effects of low habitual cocoa intake on blood pressure and bioactive nitric oxide: a randomized controlled trial. JAMA. 2007;298:49–60.
- 19. Allen RR, Carson L, Kwik-Uribe C, Evans EM, Erdman JW., Jr Daily consumption of a dark chocolate containing flavanols and added sterol esters affects cardiovascular risk factors in a normotensive population with elevated cholesterol. J Nutr. 2008;138:725–731.
- 20. Almoosawi S, Fyfe L, Ho C, Al-Dujaili E. The effect of polyphenol-rich dark chocolate on fasting capillary whole blood glucose, total cholesterol, blood pressure and glucocorticoids in healthy overweight and obese subjects. Br J Nutr.2010;103:842–850.
- 21. Ried K, Sullivan TR, Fakler P, Frank OR, Stocks NP. Effect of cocoa on blood pressure. Cochrane Database Syst Rev. 2012;8:CD008893.
- 22. Shrime MG, Bauer SR, McDonald AC, Chowdhury NH, Coltart CE, Ding EL. Flavonoid-rich cocoa consumption affects multiple cardiovascular risk factors in a meta-analysis of short-term studies. J Nutr. 2011;141:1982–1988.

- Davison K, Coates AM, Buckley JD, Howe PR. Effect of cocoa flavanols and exercise on cardiometabolic risk factors in overweight and obese subjects. Int J Obes (Lond) 2008;32:1289–1296.
- 24. Hooper L, Kay C, Abdelhamid A, Kroon PA, Cohn JS, Rimm EB, Cassidy A. Effects of chocolate, cocoa, and flavan-3-ols on cardiovascular health: a systematic review and meta-analysis of randomized trials. Am J Clin Nutr.2012;95:740–751.
- 25. Mursu J, Voutilainen S, Nurmi T, Rissanen TH, Virtanen JK, Kaikkonen J, Nyyssönen K, Salonen JT. Dark chocolate consumption increases HDL cholesterol concentration and chocolate fatty acids may inhibit lipid peroxidation in healthy humans. Free Radic Biol Med. 2004;37:1351–1359.
- 26. Jia L, Liu X, Bai YY, Li SH, Sun K, He C, Hui R. Short-term effect of cocoa product consumption on lipid profile: a meta-analysis of randomized controlled trials. Am J Clin Nutr. 2010;92:218–225.
- 27. Corti R, Flammer AJ, Hollenberg NK, Lüscher TF. Cocoa and cardiovascular health. Circulation. 2009;119:1433–1441.
- Fernández-Murga L, Tarín JJ, García-Perez MA, Cano A. The impact of chocolate on cardiovascular health. Maturitas. 2011;69:312–321.
- 29. Tey SL, Brown RC, Gray AR, Chisholm AW, Delahunty CM. Long-term consumption of high energy-dense snack foods on sensory-specific satiety and intake. Am J Clin Nutr. 2012;95:1038–1047.
- Farhat G, Drummond S, Fyfe L, Al-Dujaili EA. Dark chocolate: an obesity paradox or a culprit for weight gain? Phytother Res. 2014;28:791–797.
- 31. Panel on Macronutrients; Panel on the Definition of Dietary Fiber; Subcommittee on Upper Reference Levels of Nutrients; Subcommittee on Interpretation and Uses of Dietary Reference Intakes; Standing Committee on the Scientific Evaluation of Dietary Reference Intakes; Food and Nutrition Board; Institute of Medicine (US) Dietary Reference Intakes for Energy, Carbohydrate, Fiber, Fat, Fatty Acids, Cholesterol, Protein and Amino Acids. Washington, D.C.: National Academies Press; 2005.

- Harris JA, Benedict FG. A Biometric Study of Basal Metabolism in Man.Washington, D.C.: Carnegie Institute of Washington; 1919.
- Hill JO, Peters JC. Environmental contributions to the obesity epidemic.Science. 1998;280:1371–1374.
- Stroebele N, Ogden LG, Hill JO. Do calorie-controlled portion sizes of snacks reduce energy intake? Appetite. 2009;52:793–796.
- 35. Sciamanna CN, Kiernan M, Rolls BJ, Boan J, Stuckey H, Kephart D, Miller CK, Jensen G, Hartmann TJ, Loken E, Hwang KO, Williams RJ, Clark MA, Schubart JR, Nezu AM, Lehman E, Dellasega C. Practices associated with weight loss versus weight-loss maintenance results of a national survey. Am J Prev Med.2011;41:159–166.
- Rolls BJ. Plenary lecture 1: dietary strategies for the prevention and treatment of obesity. Proc Nutr Soc. 2010;69:70–79.
- 37. Sallis JF, Haskell WL, Wood PD, Fortmann SP, Rogers T, Blair SN, Paffenbarger RS., Jr Physical activity assessment methodology in the Five-City Project. Am J Epidemiol. 1985;121:91–106.
- 38. Tsai MC, Chang CM, Huang TL. Changes in high-density lipoprotein and homeostasis model assessment of insulin resistance in medicated schizophrenic patients and healthy controls. Chang Gung Med J. 2010;33:613–618.
- 39. Friedewald WT, Levy RI, Fredrickson DS. Estimation of the concentration of low-density lipoprotein cholesterol in plasma, without use of the preparative ultracentrifuge. Clin Chem. 1972;18:499–502.
- 40. Greenberg JA, Buijsse B. Habitual chocolate consumption may increase body weight in a dose-response manner. PLoS One. 2013;8:e70271.
- Desch S, Kobler D, Schmidt J, Sonnabend M, Adams V, Sareban M, Eitel I, Bluher M, Schuler G, Thiele H. Low vs. higher-dose dark chocolate and blood pressure in cardiovascular high-risk patients. Am J Hypertens. 2010;23:694–700.
- 42. Matsui N, Ito R, Nishimura E, Yoshikawa M, Kato M, Kamei M, Shibata H, Matsumoto I, Abe K, Hashizume S. Ingested cocoa can prevent high-fat diet-induced obesity by

regulating the expression of genes for fatty acid metabolism. Nutrition. 2005;21:594–601.

- 43. Min SY, Yang H, Seo SG, Shin SH, Chung MY, Kim J, Lee SJ, Lee HJ, Lee KW. Cocoa polyphenols suppress adipogenesis in vitro and obesity in vivo by targeting insulin receptor. Int J Obes (Lond) 2013;37:584–592.
- 44. Tokede OA, Gaziano JM, Djoussé L. Effects of cocoa products/dark chocolate on serum lipids: a meta-analysis. Eur J Clin Nutr. 2011;65:879–886.
- 45. Engler MB, Engler MM, Chen CY, Malloy MJ, Browne A, Chiu EY, Kwak HK, Milbury P, Paul SM, Blumberg J, Mietus-Snyder ML. Flavonoid-rich dark chocolate improves endothelial function and increases plasma epicatechin concentrations in healthy adults. J Am Coll Nutr. 2004;23:197–204.
- 46. Nordmann AJ, Nordmann A, Briel M, Keller U, Yancy WS, Jr, Brehm BJ, Bucher HC. Effects of low-carbohydrate vs low-fat diets on weight loss and cardiovascular risk factors: a meta-analysis of randomized controlled trials. Arch Intern Med. 2006;166:285–293.
- 47. Makris A, Foster GD. Dietary approaches to the treatment of obesity. Psychiatr Clin North Am. 2011;34:813–827.
- Schoeller DA. How accurate is self-reported dietary energy intake? Nutr Rev.1990;48:373–379.



Figure 3.1. Flow diagram of participant enrollment in a study of premenopausal women with overweight/obesity, designed to evaluate changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily dark chocolate snacks plus once daily sugarfree cocoa beverage or twice daily non-chocolate snacks plus once daily sugar-free non-cocoa beverage.

Table 3.1. Estimated dietary intake of premenopausal women with overweight/obesity at baseline, week 18 and change over time in a study evaluating changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily dark chocolate snacks plus once daily sugar-free cocoa beverage or twice daily non-chocolate snacks plus once daily sugar-free non-cocoa beverage

Dietary variableIntervalplus sugar-free cocoa beverage group ($n=20$)1 beverage group ($n=20$)1Total energy intake (kl/d)Baseline8473 t348 ²)8807 t 406 ³ Week 186887 t 2756979 t 302Change1586 t 348***1528 t 360***Carbohydrate intake (% of total energy)Baseline49.1 t 1.750.5 t 1.4Week 1848.5 t 0.850.7 t 1.4Change-0.6 t 1.7+0.2 t 1.1Fat intake (% of total energy)Baseline34.7 t 1.133.1 t 1.2Fat intake (% of total energy)Baseline33.8 t 1.031.5 t 1.4Protein intake (% of total energy)Baseline15.8 t 0.514.9 t 0.4Week 1815.8 t 0.516.5 t 0.61.2 t 0.5Alcohol intake (% of total energy)Baseline1.3 t 0.41.5 t 0.7Alcohol intake (% of total energy)Baseline1.3 t 0.41.5 t 0.7Alcohol intake (% of total energy)Baseline1.3 t 0.41.5 t 0.7Alcohol intake (% of total energy)Baseline1.3 t 0.41.5 t 0.7Total sugar (g/d)Baseline84.4 t 6.1103.9 t 8.2Total sugar (g/d)Baseline84.4 t 6.1103.9 t 8.2Total fibre (g/d)Baseline21.2 t 1.918.9 t 1.3Saturated fat (g/d)Baseline25.7 t 1.925.5 t 3.8 t 0.6Saturated fat (g/d)Baseline25.7 t 2.123.5 t 1.8Saturated fat (g/d)Baseline25.7 t 2.123.5 t 1.8Saturated fat (g/d)Baseline25.7			Dark-chocolate snacks	Non-chocolate snacks plus
Dietary variableIntervalbeverage $group (n=29)^{11}$ beverage $group (n=30)$ Total energy intake (kl/d)Baseline 8473 ± 348^{21} 8807 ± 406^{21} Week 18 6887 ± 275 6979 ± 302 Change $1586 \pm 348^{***}$ $1828 \pm 360^{***}$ Carbohydrate intake (% of total energy)Baseline 49.1 ± 1.7 50.5 ± 1.4 Fat intake (% of total energy)Baseline 49.1 ± 1.7 50.5 ± 1.4 Fat intake (% of total energy)Baseline 34.7 ± 1.1 31.5 ± 1.4 Fat intake (% of total energy)Baseline 34.7 ± 1.1 31.5 ± 1.4 Protein intake (% of total energy)Baseline 15.8 ± 0.5 14.9 ± 0.4 Protein intake (% of total energy)Baseline 15.8 ± 0.5 14.9 ± 0.4 Alcohol intake (% of total energy)Baseline 1.3 ± 0.4 1.5 ± 0.6 Alcohol intake (% of total energy)Week 18 1.5 ± 0.6 1.2 ± 0.5 Alcohol intake (% of total energy)Baseline 84.4 ± 6.1 103.9 ± 8.2 Total sugar (g/d)Baseline 84.4 ± 6.1 103.9 ± 8.2 Total sugar (g/d)Baseline 81.4 ± 0.4 $75.4 \pm 8.3 \pm 0.5$ Total fibre (g/d)Baseline 25.2 ± 1.5 27.3 ± 2.0 Saturated fat (g/d)Baseline 25.2 ± 1.5 27.3 ± 2.0 Change $-6.0 \pm 1.7^{****}$ -0.6 ± 1.3 Change $-6.0 \pm 1.7^{****}$ -0.6 ± 1.3 Saturated fat (g/d)Baseline 25.7 ± 1.5 27.3 ± 2.0 Change $-6.0 \pm 1.7^{****}$ </td <td></td> <td></td> <td>plus sugar-free cocoa</td> <td>sugar-free non-cocoa</td>			plus sugar-free cocoa	sugar-free non-cocoa
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Dietary variable	Interval	beverage group (<i>n</i> =29) ¹⁾	beverage group (<i>n</i> =30)
Week 18 6887 ± 275 6979 ± 302 Change 1586 ± 348*** 1828 ± 360*** Carbohydrate intake (% of total energy) Baseline 49.1 ± 1.7 50.5 ± 1.4 Week 18 48.5 ± 0.8 50.7 ± 1.4 Change -0.6 ± 1.7 +0.2 ± 1.1 Fat intake (% of total energy) Baseline 34.7 ± 1.1 33.1 ± 1.2 Week 18 38.8 ± 1.0 31.5 ± 1.4 Protein intake (% of total energy) Baseline 15.8 ± 0.5 14.9 ± 0.4 Week 18 16.2 ± 0.5 16.5 ± 0.6 Protein intake (% of total energy) Baseline 15.8 ± 0.5 14.9 ± 0.4 Week 18 15.2 ± 0.5 Alcohol intake (% of total energy) Baseline 1.3 ± 0.4 1.5 ± 0.7 Week 18 15.9 ± 0.7 Total sugar (g/d) Baseline 8.3 ± 1.0 1.2 ± 0.5 1.2 ± 0.5 Total sugar (g/d) Baseline 8.4 ± 6.1 103.9 ± 8.2 Total sugar (g/d) Baseline 21.2 ± 1.9 18.3 ± 1.0 Week 18 19.6 ± 1.3 0.6 ± 1.3 0.6 ± 1.3 Saturated fat (g/d) Baseline </td <td>Total energy intake (kJ/d)</td> <td>Baseline</td> <td>8473 ± 348²⁾</td> <td>8807 ± 406²⁾</td>	Total energy intake (kJ/d)	Baseline	8473 ± 348 ²⁾	8807 ± 406 ²⁾
Change1586 ± 348***1828 ± 360***Carbohydrate intake (% of total energy)Baseline49.1 ± 1.750.5 ± 1.4Week 1848.5 ± 0.850.7 ± 1.4Change-0.6 ± 1.7-0.2 ± 1.1Fat intake (% of total energy)Baseline34.7 ± 1.133.1 ± 1.2Week 1833.8 ± 1.031.5 ± 1.4Change-0.9 ± 1.1-1.6 ± 1.1Protein intake (% of total energy)Baseline15.2 ± 0.514.9 ± 0.4Meek 1816.2 ± 0.514.9 ± 0.4-0.6 ± 1.7Alcohol intake (% of total energy)Baseline13.5 ± 0.514.9 ± 0.4Meek 181.5 ± 0.514.9 ± 0.4-0.6 ± 1.7Alcohol intake (% of total energy)Baseline1.3 ± 0.41.5 ± 0.7Meek 181.5 ± 0.51.2 ± 0.5-0.3 ± 0.7Total sugar (g/d)Baseline8.4 ± 6.1103.9 ± 8.2Meek 181.5 ± 0.5-0.3 ± 0.7-0.5 ± 0.3 ± 0.7Total sugar (g/d)Baseline21.2 ± 1.918.9 ± 1.3Meek 181.5 ± 0.5-0.3 ± 0.7-0.5 ± 0.5Total fibre (g/d)Baseline21.2 ± 1.918.9 ± 1.3Saturated fat (g/d)Baseline21.2 ± 1.918.3 ± 1.0Meek 1819.6 ± 1.3-0.6 ± 1.3-0.6 ± 1.3Saturated fat (g/d)Baseline25.2 ± 1.527.3 ± 2.0Week 1819.2 ± 1.4-0.3 ± 1.8-0.6 ± 1.3Cholesterol (mg/d)Baseline25.7 ± 1.523.5 ± 1.8Meek 1819.2 ± 1.4-0.3 ± 1.8-0.6		Week 18	6887 ± 275	6979 ± 302
$\begin{array}{llllllllllllllllllllllllllllllllllll$		Change	1586 ± 348***	1828 ± 360***
$\begin{array}{ c c c c } Week 18 & 48.5 \pm 0.8 & 50.7 \pm 1.4 \\ Change & -0.6 \pm 1.7 & 40.2 \pm 1.1 \\ \hline \begin{tabular}{ c c c c } \hline \begin{tabular}{ c c c c c } \hline \begin{tabular}{ c c c c c } \hline \begin{tabular}{ c c c c c } \hline \end{tabular} \hline \begin{tabular}{ c c c c c c } \hline \end{tabular} \hline tab$	Carbohydrate intake (% of total energy)	Baseline	49.1 ± 1.7	50.5 ± 1.4
$\begin{array}{ c c c c } \begin{tabular}{ c c c } \hline Change & -0.6 \pm 1.7 & +0.2 \pm 1.1 & 33.1 \pm 1.2 & 33.1 \pm 1.2 & 33.1 \pm 1.2 & 33.8 \pm 1.0 & 31.5 \pm 1.4 & 0.6 & 33.8 \pm 1.0 & 31.5 \pm 1.4 & 0.6 & 0.9 \pm 1.1 & -1.6 \pm 1.1 & 0.6 & 0.9 \pm 1.1 & -1.6 \pm 1.1 & 0.6 & 0.9 \pm 1.1 & -1.6 \pm 1.1 & 0.6 & 0.9 \pm 1.1 & 0.6 \pm 1.1 & 0.6 & 0.9 \pm 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.4 & 0.6 & 0.5 \pm 0.6 & 0.6$		Week 18	48.5 ± 0.8	50.7 ± 1.4
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		Change	-0.6 ± 1.7	+0.2 ± 1.1
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Fat intake (% of total energy)	Baseline	34.7 ± 1.1	33.1 ± 1.2
Image -0.9 ± 1.1 -1.6 ± 1.1 Protein intake (% of total energy) Baseline 15.8 ± 0.5 14.9 ± 0.4 Week 18 16.2 ± 0.5 16.5 ± 0.6 Change +0.4 ± 0.4 +1.6 ± 0.5** Alcohol intake (% of total energy) Baseline 1.3 ± 0.4 1.5 ± 0.7 Veek 18 1.5 ± 0.6 1.2 ± 0.5 -0.3 ± 0.7 Total sugar (g/d) Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 0.0 75.4 ± 4.8 -0.8 ± 8.3 Total sugar (g/d) Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 0.0 75.4 ± 4.8 -28.5 ± 8.3* Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 -0.6 ± 1.3 Cholesterol (mg/d) Baseline 257 ± 21 235 ± 1.8 Week 18 181 ± 15 184 ± 16 -51 ± 18** <		Week 18	33.8 ± 1.0	31.5 ± 1.4
Protein intake (% of total energy) Baseline 15.8 ± 0.5 14.9 ± 0.4 Week 18 16.2 ± 0.5 16.5 ± 0.6 Change +0.4 ± 0.4 +1.6 ± 0.5** Alcohol intake (% of total energy) Baseline 1.3 ± 0.4 1.5 ± 0.7 Week 18 1.5 ± 0.6 1.2 ± 0.5 -0.3 ± 0.7 Total sugar (g/d) Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 4.0 75.4 ± 4.8 Change -13.5 ± 5.9* -28.5 ± 8.3* Total sugar (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 -0.6 ± 1.3 Change -1.8 ± 2.3 -0.6 ± 1.3 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 -0.6 ± 1.3 Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 1		Change	-0.9 ± 1.1	-1.6 ± 1.1
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Protein intake (% of total energy)	Baseline	15.8 ± 0.5	14.9 ± 0.4
Image Change +0.4 ± 0.4 +1.6 ± 0.5** Alcohol intake (% of total energy) Baseline 1.3 ± 0.4 1.5 ± 0.7 Week 18 1.5 ± 0.6 1.2 ± 0.5 -0.3 ± 0.7 Total sugar (g/d) Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 4.0 75.4 ± 4.8 Change -13.5 ± 5.9* -28.5 ± 8.3* Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Yweek 18 19.6 ± 1.3 18.3 ± 1.0 -0.6 ± 1.3 Total fibre (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 -7.0 ± 2.1** Cholesterol (mg/d) Baseline 25.7 ± 21 25.5 ± 1.8 Week 18 19.2 ± 1.4 20.3 ± 1.8 -7.0 ± 2.1** Cholesterol (mg/d) Baseline 25.7 ± 21 25.5 ± 1.8 Week 18 181 ± 15 184 ± 16 -1.1 ± 1.4* Cholesterol (mg/d) Baseline 3407 ± 174		Week 18	16.2 ± 0.5	16.5 ± 0.6
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		Change	+0.4 ± 0.4	+1.6 ± 0.5**
Week 18 1.5 ± 0.6 1.2 ± 0.5 Change +0.2 ± 0.5 -0.3 ± 0.7 Total sugar (g/d) Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 4.0 75.4 ± 4.8 Change -13.5 ± 5.9* -28.5 ± 8.3* Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 Change -1.8 ± 2.3 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215	Alcohol intake (% of total energy)	Baseline	1.3 ± 0.4	1.5 ± 0.7
Change +0.2 ± 0.5 -0.3 ± 0.7 Total sugar (g/d) Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 4.0 75.4 ± 4.8 Change -13.5 ± 5.9* -28.5 ± 8.3* Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Saturated fat (g/d) Baseline 25.2 ± 1.4 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 -11.1** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 -346 ± 215		Week 18	1.5 ± 0.6	1.2 ± 0.5
Total sugar (g/d)Baseline 84.4 ± 6.1 103.9 ± 8.2 Week 18 70.9 ± 4.0 75.4 ± 4.8 Change $-13.5 \pm 5.9^*$ $-28.5 \pm 8.3^*$ Total fibre (g/d)Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 18.3 ± 1.0 Change -1.8 ± 2.3 -0.6 ± 1.3 Saturated fat (g/d)Baseline 25.2 ± 1.5 27.3 ± 2.0 Saturated fat (g/d)Baseline 25.2 ± 1.5 27.3 ± 2.0 Cholesterol (mg/d)Baseline 257 ± 21.4 20.3 ± 1.8 Cholesterol (mg/d)Baseline 257 ± 21 235 ± 18 Sodium (mg/d)Baseline 3407 ± 174 3520 ± 235 Sodium (mg/d)Baseline 3407 ± 174 3520 ± 235 Week 18 1928 ± 152 3174 ± 171 -346 ± 215		Change	+0.2 ± 0.5	-0.3 ± 0.7
Week 18 70.9 ± 4.0 75.4 ± 4.8 Change -13.5 ± 5.9* -28.5 ± 8.3* Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 18.3 ± 1.0 Change -1.8 ± 2.3 -0.6 ± 1.3 18.3 ± 1.0 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Saturated fat (g/d) Baseline 25.2 ± 1.4 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 184 ± 16 Change -76 ± 23** -51 ± 18** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 -346 ± 215	Total sugar (g/d)	Baseline	84.4 ± 6.1	103.9 ± 8.2
Change -13.5 ± 5.9* -28.5 ± 8.3* Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 Change -1.8 ± 2.3 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215		Week 18	70.9 ± 4.0	75.4 ± 4.8
Total fibre (g/d) Baseline 21.2 ± 1.9 18.9 ± 1.3 Week 18 19.6 ± 1.3 18.3 ± 1.0 Change -1.8 ± 2.3 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 20.3 ± 1.8 Cholesterol (mg/d) Baseline 25.7 ± 2.1 23.5 ± 1.8 Cholesterol (mg/d) Baseline 25.7 ± 2.1 23.5 ± 1.8 Sodium (mg/d) Baseline 25.7 ± 2.1 23.5 ± 1.8 Sodium (mg/d) Baseline 26.7 ± 2.1 23.5 ± 1.8 Week 18 18.1 ± 1.5 18.4 ± 1.6 1.4 Change -76 ± 2.3** -51 ± 1.8** Sodium (mg/d) Baseline 3407 ± 1.74 3520 ± 2.35 Week 18 292.8 ± 1.52 31.74 ± 1.71 Change -47.9 ± 1.81* -34.6 ± 2.15		Change	-13.5 ± 5.9*	-28.5 ± 8.3*
Week 18 19.6 ± 1.3 18.3 ± 1.0 Change -1.8 ± 2.3 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 184 ± 16 Change -76 ± 23** -51 ± 18** 50 ± 1.7** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215	Total fibre (g/d)	Baseline	21.2 ± 1.9	18.9 ± 1.3
Change -1.8 ± 2.3 -0.6 ± 1.3 Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215		Week 18	19.6 ± 1.3	18.3 ± 1.0
Saturated fat (g/d) Baseline 25.2 ± 1.5 27.3 ± 2.0 Week 18 19.2 ± 1.4 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215		Change	-1.8 ± 2.3	-0.6 ± 1.3
Week 18 19.2 ± 1.4 20.3 ± 1.8 Change -6.0 ± 1.7*** -7.0 ± 2.1** Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215	Saturated fat (g/d)	Baseline	25.2 ± 1.5	27.3 ± 2.0
Change -6.0±1.7*** -7.0±2.1** Cholesterol (mg/d) Baseline 257±21 235±18 Week 18 181±15 184±16 Change -76±23** -51±18** Sodium (mg/d) Baseline 3407±174 3520±235 Week 18 2928±152 3174±171 Change -479±181* -346±215		Week 18	19.2 ± 1.4	20.3 ± 1.8
Cholesterol (mg/d) Baseline 257 ± 21 235 ± 18 Week 18 181 ± 15 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215		Change	-6.0 ± 1.7***	-7.0 ± 2.1**
Week 18 181 ± 15 184 ± 16 Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215	Cholesterol (mg/d)	Baseline	257 ± 21	235 ± 18
Change -76 ± 23** -51 ± 18** Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215		Week 18	181 ± 15	184 ± 16
Sodium (mg/d) Baseline 3407 ± 174 3520 ± 235 Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215		Change	-76 ± 23**	-51 ± 18**
Week 18 2928 ± 152 3174 ± 171 Change -479 ± 181* -346 ± 215	Sodium (mg/d)	Baseline	3407 ± 174	3520 ± 235
Change -479 ± 181* -346 ± 215		Week 18	2928 ± 152	3174 ± 171
		Change	-479 ± 181*	-346 ± 215

¹⁾Missing data for n = 1 participant.

²⁾Values are means ± SEM. *P*-values from intention-to-treat analysis, using analysis of covariance with repeated measures on the time factor.

 ${}^{*}P < 0.05$, ${}^{**}P < 0.01$, ${}^{***}P < 0.001$ within group change from baseline. There were no significant differences in estimated dietary intake variables between snack/beverage groups at any interval or for the change over time between snack/beverage groups.

Table 3.2. Anthropometric measurements of premenopausal women with overweight/obesity at baseline and change from baseline at weeks 6, 12 and 18 in a study evaluating changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily dark chocolate snacks plus once daily sugar-free cocoa beverage or twice daily non-chocolate snacks plus once daily sugar-free non-cocoa beverage

		Dark-chocolate snacks	Non-chocolate snacks plus
		plus sugar-free cocoa	sugar-free non-cocoa
Measurement	Interval	beverage group (<i>n</i> =30)	beverage group (<i>n</i> =30)
Age (y)	Baseline	$36.0 \pm 1.1^{1)}$	$35.9 \pm 1.1^{1)}$
Height (cm)	Baseline	164.9 ± 0.8	164.9 ± 1.2
Weight (kg)	Baseline	83.7 ± 2.5	85.1 ± 2.3
	Week 6	-2.7 ± 0.4***	-2.6 ± 0.4***
	Week 12	-4.2 ± 0.5***	-4.0 ± 0.7***
	Week 18	-4.4 ± 0.6***	-5.0 ± 0.9***
Body mass index (kg/m ²)	Baseline	30.8 ± 0.9	31.2 ± 0.7
	Week 6	$-1.0 \pm 0.1^{***}$	$-1.0 \pm 0.2^{***}$
	Week 12	$-1.6 \pm 0.2^{***}$	-1.5 ± 0.2***
	Week 18	-1.6 ± 0.2 ***	$-1.8 \pm 0.3^{**}$
Waist circumference (cm)	Baseline	89.4 ± 1.9	92.0 ± 1.9
	Week 6	-3.0 ± 0.4***	$-3.4 \pm 0.6^{***}$
	Week 12	-4.2 ± 0.5***	-4.8 ± 0.7***
	Week 18	-5.0 ± 0.9***	-5.1 ± 1.0 ***
Hip circumference (cm)	Baseline	115.4 ± 1.8	116.4 ± 1.6
	Week 6	-3.9 ± 0.5***	-3.6 ± 0.6***
	Week 12	-4.7 ± 0.7***	-4.7 ± 0.7***
	Week 18	-5.2 ± 0.8***	-4.8 ± 0.8***
Body fat (%)	Baseline	40.4 ± 1.0	41.2 ± 1.0
	Week 6	-1.4 ± 0.2 ***	$-1.0 \pm 0.3^{***}$
	Week 12	-2.7 ± 0.4***	-2.1 ± 0.4 ***
	Week 18	-3.2 ± 0.5***	-2.7 ± 0.6***

¹⁾Values are means ± SEM for baseline values and mean ± SEM for change from baseline values. *P*-values from intention-to-treat analysis, using analysis of covariance with repeated measures on the time factor. **P* < 0.05, ***P* < 0.01, ****P* < 0.001 within group change from baseline. There were no significant differences in measurements between snack/beverage groups at any interval or for the change over time between snack/beverage groups. **Table 3.3.** Blood pressure measurements and selected metabolic biomarkers of premenopausal women with overweight/obesity at baseline and change from baseline at weeks 6, 12 and 18 in a study evaluating changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily dark chocolate snacks plus once daily sugar-free cocoa beverage or twice daily non-chocolate snacks plus once daily sugar-free non-cocoa beverage

		Dark-chocolate snacks	Non-chocolate snacks plus
		plus sugar-free cocoa	sugar-free non-cocoa
Measurement	Interval	beverage group (<i>n</i> =30)	beverage group (<i>n</i> =30)
Systolic blood pressure (mm Hg)	Baseline	$118.8 \pm 1.4^{1)}$	$119.7 \pm 1.7^{1)}$
	Week 6	-1.7 ± 1.0	-2.0 ± 1.2
	Week 12	-0.6 ± 1.0	-1.8 ± 0.8*
	Week 18	-2.7 ± 1.2*	-3.4 ± 1.0**
Diastolic blood pressure (mm Hg)	Baseline	72.8 ± 1.4	74.9 ± 1.4
	Week 6	-1.6 ± 0.9	-1.6 ± 1.3
	Week 12	-2.2 ± 1.0*	-2.7 ± 1.2*
	Week 18	$-2.7 \pm 1.0^{**}$	-4.2 ± 1.3**
Glucose (mmol/L)	Baseline	5.58 ± 0.11	5.44 ± 0.10
	Week 6	-0.42 ± 0.12***	-0.27 ± 0.11*
	Week 12	-0.29 ± 0.08**	-0.22 ± 0.09*
	Week 18	-0.72 ± 0.13***	-0.83 ± 0.12***
Insulin (pmol/L)	Baseline	50.00 ± 5.33	49.31 ± 5.20
	Week 6	+2.78 ± 8.75	-1.39 ± 3.30
	Week 12	-6.95 ± 3.81	-2.78 ± 4.06
	Week 18	-13.20 ± 4.06**	-13.20 ± 4.56**
Homeostasis model of assessment – insulin	Baseline	1.80 ± 0.21	1.71 ± 1.01
resistance	Week 6	+0.04 ± 0.38	-0.08 ± 0.72
	Week 12	-0.33 ± 0.16	-0.13 ± 0.78
	Week 18	-0.63 ± 0.15***	-0.62 ± 0.88***
Total cholesterol (mmol/L)	Baseline	4.10 ± 0.21	4.15 ± 1.17
	Week 6	-0.16 ± 0.08	-0.05 ± 0.56
	Week 12	+0.19 ± 0.08*	+0.04 ± 0.61
	Week 18	+0.12 ± 0.09	+0.35 ± 0.59*
High-density lipoprotein-cholesterol (mmol/L)	Baseline	1.33 ± 0.05	1.29 ± 0.05
	Week 6	-0.08 ± 0.03*	-0.12 ± 0.03***
	Week 12	-0.04 ± 0.02*	-0.05 ± 0.03
	Week 18	+0.04 ± 0.03	+0.05 ± 0.03
Low-density lipoprotein-cholesterol (mmol/L)	Baseline	2.43 ± 0.11	2.49 ± 0.20
	Week 6	-0.11 ± 0.08	+0.08 ± 0.11
	Week 12	+0.21 ± 0.07*	+0.05 ± 0.11
	Week 18	+0.04 ± 0.10	+0.23 ± 0.12
Triacylglycerides (mmol/L)	Baseline	0.73 ± 0.04	0.82 ± 0.07
	Week 6	+0.06 ± 0.07	-0.02 ± 0.05
	Week 12	+0.06 ± 0.05	+0.06 ± 0.05
	Week 18	+0.05 ± 0.07	+0.06 ± 0.10

¹⁾Values are means ± SEM for baseline values and mean ± SEM for change from baseline values. *P*-values from intention-to-treat analysis, using analysis of covariance with repeated measures on the time factor. **P* < 0.05, ***P* < 0.01, ****P* < 0.001 within group change from baseline. There were no significant differences in measurements between snack/beverage groups at any interval or for the change over time between snack/beverage groups. Homeostasis model of assessment-insulin resistance calculated by fasting insulin concentration (μ IU/mL) × fasting glucose concentration (mg/dL) × 0.0555/22.5.

Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration²

Abstract

Background: The present study aimed to explore facilitators and barriers to weight loss (WL) and weight loss maintenance (WLM) in women who participated in a primary, 18-week comparative trial that promoted WL with an energy-restricted diet.

Methods: Twenty-three women participated in seven focus groups conducted by a moderator and co-facilitator using open-ended questions and probes. Focus groups were held in a private room and audio tape-recorded. Tapes were transcribed verbatim and thematic analysis was used to evaluate transcripts for common themes.

Results: Accountability to others, social support, planning ahead, awareness and mindfulness of food choices, basic nutrition education, portion control, exercise, and self-motivation were perceived as key facilitators for WL and WLM by women. Identified barriers included life transitions, health status changes, internal factors, environmental pressures, lack of accountability and an absence of social support.

Conclusions: Future interventions should address these salient facilitators and barriers to promote sustainable changes in women across their WL and WLM journeys.

²This article previously appeared in its entirety as Metzgar CJ, Preston AG, Miller DL. Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. *J Hum Nutr Diet*. 2014 Sept 18. doi: 10.1111/jhn.12273. [Epub ahead of print]. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is noncommercial and no modifications or adaptations are made.

Introduction

Following weight loss (WL) success, weight loss maintenance (WLM) remains a challenge (Stunkard & McLaren-Hume, 1959; Lyznicki et al., 2001; Stevens et al., 2006). Dietary programmes and related interventions that produce WL in the short term rarely promote WLM over the long term (Jeffery et al., 2000). Indeed, approximately half of any weight lost is regained within 1 year, and most individuals return to their baseline weight within 3–5 years (Goodrick et al., 1996; Jeffery et al., 2000; Byrne et al., 2003).

Although interventions and programmes provide strategies for short-term WL, few include WLM strategies, which may differ from those used to achieve WL (Byrne et al., 2003; Reyes et al., 2012). Additionally, studies examining effective long-term WL strategies (Klem et al., 1997; Jeffery et al., 2000; Ogden, 2000; Elfhag & Rossner, 2005; Wing & Phelan, 2005; Wing et al., 2006; Svetkey et al., 2008; Burke et al., 2011) have not always captured the subtleties of WLM that can only be uniquely described by individuals engaging in such efforts (Kayman et al., 1990; Byrne et al., 2003; Barnes et al., 2007; Reyes et al., 2012). The National Weight Control Registry, with over 10 000 members, is the largest database of individuals successful at longterm WLM (Wing & Phelan, 2005). To be eligible to join the National Weight Control Registry, individuals must have maintained a WL of \geq 13.6 kg (\geq 30 lbs) for \geq 1 year. Behavioural and psychological characteristics as well as WLM strategies of National Weight Control Registry members have been prospectively investigated using detailed questionnaires and follow-up surveys. Six key strategies for WLM have been identified and include: engaging in high levels of physical activity; consuming a diet low in energy and fat; eating breakfast; maintaining a consistent eating pattern; self-monitoring of weight on a frequent basis; and reversing slight weight regains quickly (Wing & Phelan, 2005). These behaviours do not fully reflect the cognitive and emotional aspects of WLM strategies or real and perceived structural support systems that influence WLM success. Additionally, National Weight Control Registry members each have their own WL journey to long-term WLM. Strategies for and barriers and facilitators to WL and WLM need to be examined further in groups of individuals who have undergone the same WL intervention.

The purpose of the present qualitative study was to use focus groups to explore facilitators and barriers to WL and WLM in women who previously participated in a randomised comparative trial that promoted WL using an energy-restricted diet (Nickols-Richardson et al., 2014) and to capture insights into their perceptions regarding WL during and WLM following completion of a single intervention. No hypotheses were established a priori because an inductive approach was used to analyse data.

Materials and methods

Participants

Women were recruited from a pool of 51 participants who completed an 18-week comparative trial that promoted WL using an energy-restricted diet including portion-controlled daily sweet snacks; diets were identical in macronutrient composition, differing only in snack group assignment (Nickols-Richardson et al., 2014). The primary intervention included a nutrition education component delivered by a registered dietitian. Interactive, group education classes were held for 1 h each week during the 18-week trial and included strategies for portion control, planning ahead for meals and consuming vegetables for WL (Nickols-Richardson et al., 2014). Class topics included general nutrition principles, eating away from home, food selection, preparation methods and recipe modification.

All women who fully completed the primary study (n = 51) were invited to participate in the present qualitative study to obtain a homogenous sample of individuals with similar backgrounds and experiences related to WL (Morse & Field, 1995; Stewart et al., 2007; Krueger & Casey, 2009; James et al., 2012), with no additional inclusion or exclusion criteria. Original inclusion and exclusion criteria are described elsewhere (Nickols-Richardson et al., 2014). Briefly, the primary study included overweight and obese women who were moderately physically active, eumenorrheic, with a self-reported stable body weight, who scored <50 on the Zung Self-Rating Depression Scale/Status Inventory (Zung, 1986) and who reported no intolerance, aversion or allergy to chocolate. Women who currently smoked, were pregnant or attempting to become pregnant, had a hysterectomy and/or ovariectomy without hormone replacement therapy and those who used oral contraceptives for <2 years in duration (if used)

were excluded, in addition to women who used medications, including steroid or thyroid hormones, bisphosphonates, anticonvulsants and glucocorticoids, or consumed \geq 40 g of chocolate per day (i.e. the equivalent of one standard chocolate bar or more day-1).

Packets explaining the purpose of the present study were sent to women, followed by telephone or e-mail messages confirming the receipt of study materials. By completing and returning study questionnaires, participants provided their implied informed consent for the study, and interested women were invited to participate in one focus group session. Recruitment, enrollment activities and focus group sessions were conducted from 16 April until 9 July 2012, approximately ≥20 months following the completion of the primary intervention. The present study and primary study were approved by the Institutional Review Board for Research Involving Human Subjects at The Pennsylvania State University.

Anthropometric measurements

On the day of the focus group session, standing height was measured to the nearest 0.1 cm using a stadiometer (Seca 700; Seca North America East, Hanover, MD, USA) and body weight was measured to the nearest 0.1 kg using a calibrated balance-beam scale (Seca 700). Height and body weight were used to calculate body mass index (BMI; kg m–2). A retractable measuring tape (Gulik II; Country Technology, Inc., Gay Mills, WI, USA) was used to measure waist and hip circumferences to the nearest 0.1 cm. Two measurements at both the waist and the hip were obtained and averaged to result in a single measurement for each site. Waist circumference was measured at the narrowest point of the waist, approximately 1 inch (2.54 cm) above the navel; hip circumference was measured at the widest part of the buttocks (Nickols-Richardson et al., 2014). One investigator completed all height and body weight measurements; a second investigator conducted all waist and hip circumference measurements.

Focus group sessions

Focus groups promoted discussion among participants and encouraged sharing of perceptions, experiences and opinions related to WL and WLM. As a result of a limited pool of participants, it

was preferable to conduct more groups with fewer participants, as recommended by Greenbaum (1998) and Krueger (1998) and employed by Reyes et al. (2012). Seven mini-focus groups, including dyads and one triad, were conducted (Greenbaum, 1998; Krueger, 1998; Krueger & Casey, 2009). Groups included four, three, two, four, four, four and two women per session. The same moderator (CJM), who was not involved in the primary intervention, guided participant discussions, according to standard focus group methods (Creswell, 1998; Stewart et al., 2007; Harris et al., 2009; Krueger & Casey, 2009). A purpose statement was read at the beginning of each focus group for consistency of context and orientation to discussion. Participants were asked to respond to 11 open-ended questions (Table 3.4) that were obtained from or modelled after those developed by Hindle & Carpenter (2011). Questions, including probing statements, covered topics including WL and WLM success and limitations, facilitators and barriers to WLM, strategies for weight regulation, eating patterns and perceptions of snacking. Questions also addressed constructs of the theory of planned behaviour, social cognitive theory and the health belief model (Table 3.4).

One co-facilitator (SNR) attended all focus groups and maintained written records. Focus groups were conducted in a private room and audio tape-recorded. Digital audiotapes were transcribed verbatim by the facilitator using written records from the co-facilitator to clarify and/or verify transcribed audiotapes as needed. Each focus group lasted approximately 1 h. Upon completion of the session, each participant received a \$50 gift card.

Statistical analysis

Descriptive statistics [mean (SD)] were used to characterise study participants. Paired t-tests were used to compare current anthropometric measurements of women participating in focus groups (n = 23) with their post-primary-intervention measurements. Independent t-tests were used to compare current anthropometric measurements of women who participated in focus groups (n = 23) with post-primary-intervention measurements of women who did not participate in focus groups (n = 28). Quantitative data analyses were conducted using SPSS, version 20.0 (IBM Corp., Armonk, NY, USA). P < 0.05 was considered statistically significant.

Focus group transcripts were evaluated using thematic analysis (Braun & Clarke, 2006). The moderator and co-facilitator independently reviewed transcripts and generated codes that were later collated into themes. To be considered, themes must have been stated by at least three participants across groups. Following independent analysis, themes were confirmed by these investigators through a question-by-question review of the major themes identified. Discrepancies were reconciled through discussion and further review of transcripts. As common themes were identified, the meanings of these phenomena were interpreted and considerations for future programmes were proposed. Selected verbatim quotes that captured participants' insights, perceptions and opinions have been included.

Results

Subject characteristics

Anthropometric measurements were obtained from twenty-three women, with a mean (SD) age of 38.8 (6.0) years and mean (SD) BMI of 30.0 (6.2) kg m–2 who engaged in one of seven focus group sessions. Twelve women declined to participate, and 16 women were lost to follow-up after the primary study. Characteristics of participants are presented in Table 3.5. Twenty-one women were Caucasian and two women were African American. There were no significant differences in anthropometric measurements of focus group participants (n = 23) and women who did not participate (n = 28). Within focus group participants, BMI, waist circumference and hip circumference measurements were significantly lower after the 18-week primary intervention compared to current measurements, demonstrating that participants were not successful at WLM.

Theme identification and common themes

Nine major themes were identified through data analyses and included: (i) the weight loss journey; (ii) accountability and support; (iii) planning ahead, mindfulness and awareness; (iv) nutrition education; (v) portion control; (vi) exercise; (vii) motivation; (viii) total lifestyle change; and (ix) eating patterns and snacking. The meanings of these themes and considerations for future interventions are summarised in Table 3.6.

Women described their WL journeys as continual bouts of weight gain, WL, WLM and prevention of weight regain that were related to life transitions. One woman stated, 'It's [WL journey] definitely going to be an ongoing process'. Another participant reflected that 'it has been this up and down type of journey for me, for a long time'. All women self-reported at least one attempt at losing weight throughout their lifetimes, with many women reporting multiple episodes and methods for WL. Specific methods used by participants included Weight Watchers®, The South Beach Diet®, over-the-counter medications and various energy-restricted diets. Although these approaches were successful in producing initial WL, reports of long-term success with WLM were scarce among these participants.

Accountability to others and support from friends, family members and coworkers were verbalised as key facilitators that allowed women to maintain or continue their WL following the intervention. Lack of accountability to someone else and an absence of support from others were commonly reported as barriers to achieving further WL and WLM following completion of the primary study. Compared to previous attempts with WL, accountability to study investigators was perceived as a key component of the WL intervention, motivating women to achieve WL. Women also identified weekly educational group meetings during the primary comparative trial as being useful for facilitating accountability, support and motivation. Without an individual (i.e. study investigator) monitoring their progress after the primary intervention ended, women perceived a drastic drop in motivation for WLM. One woman's statement ('1 mean for me, hands down, accountability. I have no one looking over my shoulder') represented this idea. Women in the present study did not mention self-accountability, but only accountability to someone else.

Although some women perceived others to be supportive and complimentary of their WL efforts, many women reported negative reactions. Women indicated that friends, family members and coworkers tempted them with high-energy, savoury foods or made 'snide' remarks regarding their healthy food choices when eating in restaurants, at social gatherings and during family mealtimes. Comments such as: 'You look ill.'; 'You don't need to lose weight'; 'You are having a salad again today?'; 'I don't know why you have to eat all that [healthy] stuff, just eat less'; 'You should stop losing weight'; and 'Go ahead and eat that [dessert]. You deserve

that. You work hard' were reported. Several women remarked that their spouses hid and consumed snack foods and candies in secret. Women considered that such comments and practices were forms of sabotage that were often unintentional, yet unhelpful in facilitating WL and WLM. One participant expressed this concern by stating, 'See my support group means to be supportive, but comes off as not. Like my husband had made a comment one time. He's like, "Oh I hate when you're on a diet, because then I can't eat what I want". One woman stated, 'When you don't have the support group, it makes [WLM] very difficult'.

Planning ahead emerged as a common thread throughout focus group discussions. Although planning ahead was identified as a strategy that women used to maintain their WL, women also recognised that such preparation took a substantial amount of time. Women who planned their meals in advance or took their own food to social events perceived greater success with WL and WLM. For example, one woman stated, 'Plan. If you don't plan, it's ... uh ... for me, it's just so hard'. Planning was reported as particularly helpful for maintaining the dietary intervention during social activities, holidays, travel and when dining away from home. The terms 'mindfulness' and 'awareness' were commonly repeated by women during focus group discussions. These terms were often used in conjunction with planning ahead, as demonstrated by one woman who stated, 'And being planful and being mindful, like paying attention'. Another participant commented that 'You know, that sort of speaks to, like what you [other participant] were saying, like making healthier choices, just about what you're eating and recognising like where the calories come from and just really being a bit more mindful of that'.

Many women reported that the addition of weekly sessions with a 'credible resource' (registered dietitian) differed from past attempts to lose weight using commercial WL programmes and approaches. During focus group discussions, most women reported that they gained valuable nutrition knowledge and perceived an enhanced ability to make healthier food choices during the WL intervention. One woman stated, 'Yeah, for me, it was the accountability mixed with the, um, the education'. A majority of women admitted that they did not continue applying nutrition education principles during WLM because they no longer had the group support system. Women reported that weekly sessions assisted them with incorporating more
fruits and vegetables, whole grains, lean protein and low-fat dairy in their eating patterns by having a greater sense of awareness of nutritional value of foods and an understanding of the relationship between dietary intake and body weight changes. After the intervention, loss of this resource was a barrier to WLM, as expressed by one woman who noted, 'And having that educational background from those sessions just really helped me know what I have to do, it's just whether or not I'm following it on that day [after the trial]'.

Many women in the present study indicated that they became acutely aware of their own 'portion distortion' during the WL trial. Furthermore, some women reported having an 'inner voice' reminding them to avoid large portion sizes in the interval between the end of the intervention and focus group session. One woman stated, 'I kept hearing 'that's portion distortion, that's portion distortion' in my mind'. Another participant commented, 'I think measuring my cereal in the mornings is a big eye opener'. Women perceived that, when they implemented portion control during social events, holidays and dining away from home, they were better able to maintain WL. Portion control was identified as the key facilitator to body weight regulation by many women during the focus group sessions.

Women who perceived greater success with additional WL and WLM post-intervention reported developing an exercise routine and continuously engaging in regular exercise. For example, one woman reported that, 'exercise for me is the key'. Another participant stated, 'I think my biggest thing, it's not what I'm eating or what I'm not eating, it's exercising'. Many women who self-reported an inability to maintain WL and subsequently experienced weight gain considered that a lack of exercise was a substantial barrier to their success. Finding 'time' and 'enjoyable activities' were cited as issues for not engaging in exercise, whereas other women expressed 'feeling guilty' about taking time for themselves to exercise.

Self-motivation or a lack thereof was an additional facilitator or barrier, respectively, to WL and WLM reported by women during focus group discussions. Other internal factors, including a lack of self-control, willpower and self-regulation, were identified as barriers. Readiness for change was perceived as necessary for self-motivation; one woman commented, 'Readiness. You have to be ready'. Waning self-motivation also was commonly cited as a barrier, as represented by one woman's comment, 'I have trouble maintaining focus and

motivation, even though I am successful in the beginning'. Renewing motivation on a daily basis was perceived by women as a facilitator to WL and WLM. Women referred to this as a 'freshslate mentality' or the ability to self-forgive 'slip-ups' or 'bad days' of poor compliance with healthy dietary patterns. This perception was expressed by a participant who stated, 'For me, the, the notion that each day is, is a new day instead of, oh well, yesterday was shot so I might as well continue the downward spiral. That has been helpful to say, okay, I'll start again today'.

Several women commented that a 'total lifestyle change', encompassing diet and exercise modifications, was required to achieve and maintain WL. One woman supported this idea by stating, 'Exercise, eating right. Just a better whole lifestyle'.

A distinct feature of the primary WL trial was the use of a dietary approach that emphasised vegetables and portion control, at the same time as including sweet snacks and a sugar-free beverage on a daily basis. Women reported that they found this approach 'quite flexible' and 'accommodating' for WL and WLM. Almost all of the women condered that inclusion of portion-controlled sweet snacks during the 18-week intervention altered their attitudes, beliefs and perceptions of eating and snacking patterns for the long term. One woman represented these feelings by stating, 'So I always thought it was bad to snack, and since, I realised that it's okay, and that it's actually easier for me to keep my weight in check if I snack'. During focus group discussions, women reported an overall improvement in eating patterns as a result of 'more frequent breakfast consumption,' 'better portion control' and 'healthy snack choices'. Women experienced a shift in their perceptions of snacks from 'nonnutritious', 'easily accessible food items' that were generally 'consumed without planning' and snack foods that were used to satisfy cravings ('vending machine food items') from snacks consumed to control or manage hunger and meet overall dietary intake recommendations ('fruits, vegetables, low-fat dairy foods'). For example, one woman stated, 'Snack foods and snacks are two different things. Snacks are – you can be purposeful about your snacks. Snack foods are the easy way out'. Most women expressed a positive view of snacks, considering that they are beneficial and necessary to dietary intake during WL and WLM.

Discussion

Few qualitative studies examining facilitators and barriers to WL and WLM have been conducted (Byrne et al., 2003; Barnes et al., 2007; Befort et al., 2008; Herriot et al., 2008; Thomas et al., 2008; Ely et al., 2009; Green et al., 2009; Hardcastle & Hagger, 2011; Hindle & Carpenter, 2011; Reyes et al., 2012), specifically in a homogenous sample of individuals with similar experiences related to WL and WLM. Even so, many of the themes identified by women in the present study are consistent with strategies reported in previous qualitative and quantitative studies (Kayman et al., 1990; Klem et al., 1997; Jeffery et al., 2000; Ogden, 2000; Wing & Hill, 2001; Byrne et al., 2003; Elfhag & Rossner, 2005; Wing & Phelan, 2005; Wing et al., 2006; Barnes et al., 2007; Befort et al., 2008; Herriot et al., 2008; Svetkey et al., 2008; Thomas et al., 2008; Ely et al., 2009; Green et al., 2009; Hardcastle & Hagger, 2011; Hindle & Carpenter, 2011; Sciamanna et al., 2011; Reyes et al., 2012).

Approaches used to induce WL by women in the present study are consistent with those reported by others (Barnes et al., 2007; Herriot et al., 2008; Thomas et al., 2008; Hindle & Carpenter, 2011; Reyes et al., 2012) as are reports of multiple WL attempts throughout the lifetime (Hindle & Carpenter, 2011). Women in the present qualitative study specifically acknowledged certain points of the lifespan to be problematic and stressors that precipitated weight gain or made weight maintenance difficult throughout their weight management journeys. Specifically, women identified pregnancy and post-pregnancy years as particularly challenging times, consistent with a qualitative study of African American women in whom motherhood was cited as a precursor to weight gain (Befort et al., 2008) and the initiation of WL efforts as a result of this weight gain (Goodrick et al., 1996). Life transitions related to student status, employment, family structure and health status also were reported by women as critical periods that hindered weight management. Both young adulthood and midlife appear to be crucial intervals for weight gain (Ball et al., 2002; Wane et al., 2010), and life stress and significant life events also have been associated with weight gain (Elfhag & Rossner, 2005). Women perceived that their capacity for achieving and maintaining WL was lower than for men, consistent with evidence demonstrating that women are twice as likely as men to experience major weight gain over the life course (Williamson et al., 1990). Consistent with other studies, women described their desires to improve appearance, self-esteem and/or

health as motivations for losing body weight (Goodrick et al., 1996; Barnes et al., 2007; Thomas et al., 2008; Hindle & Carpenter, 2011; Reyes et al., 2012). Therefore, future interventions should be tailored to the individual needs of each participant with attention to a woman's biology, physiology, life stage and previous experiences.

Social support is widely regarded as an integral element of WL and WLM programmes, regardless of whether this support comes from friends, family members, coworkers or a support group (Elfhag & Rossner, 2005; Barnes et al., 2007). Additionally, social support has been identified as both a facilitator and barrier to making dietary and behaviour changes in previous research (Tessaro et al., 2006). Therefore, it is not surprising that women in the present qualitative study reported that support or a lack thereof from these individuals and groups served to facilitate or hinder WL and WLM, respectively. Saboteurs have been identified as a barrier to WLM in several studies (Befort et al., 2008; Thomas et al., 2008; Hardcastle & Hagger, 2011; Hindle & Carpenter, 2011), as also reported by women in the present study. Evidence suggests that social support may enhance motivation for weight loss (Jeffery et al., 2000), offering an explanation as to why women found social support so crucial. Previous research has identified group support as being critical to weight management as a means of problem-solving and experience sharing (Herriot et al., 2008; Ely et al., 2009), facets that were shared by participants in the present study. Unlike members of the National Weight Control Registry, women did not report self-accountability as a means for motivation but, instead, found external accountability necessary to provide motivation and support. Focus group participants in a previous study of WL maintainers and regainers (Reyes et al., 2012) perceived that a lack of external accountability and low levels of structured support resulted in decreased motivation for weight regulation. Similarly, Hardcastle & Hagger (2011) found that obese men and women desired an external source of accountability to serve as an incentive to maintain lifestyle changes and provide support and motivation. Internal factors appear to be key components for successful weight management in some (Herriot et al., 2008) and maintaining motivation appeared to be a struggle for many of the women in the present study. Sciamanna et al. (2011) found motivation to engage in weight control practices to diminish over time and suggested different practices should be encouraged throughout the weight management

process to maintain motivation and increase effectiveness. Along these lines, renewing motivation on a daily basis was perceived by women as a facilitator to WL and WLM, a finding supported by Green et al. (2009). Programmes and interventions that provide a source of external accountability, as well as a support network, may be beneficial in helping individuals achieve their WL and WLM goals.

The primary WL intervention emphasised planning for food intake, whereas previous WL interventions have not always included this skill (Wing et al., 2006; Svetkey et al., 2008; Foster-Schubert et al., 2012). Consequently, the theme of planning ahead and being mindful and aware of food choices has not been identified in previous qualitative work. A cross-sectional study of US adults who were successful at WLM did find that planning was a crucial component in their success (Sciamanna et al., 2011). Although the primary intervention did not employ mindful eating strategies, previous research has shown that a conscious awareness of foods and energy actually consumed plays a role in WL and WLM (Wansink, 2010). Future interventions should incorporate planning skills to help individuals enhance mindful awareness of food intake.

Women in the present study reported that they gained valuable general nutrition knowledge as a result of nutrition education classes and appreciated that a registered dietitian was involved in the primary intervention because few of their previous WL attempts involved a registered dietitian or even a credible source of dietary information. Registered dietitians are a credible source of reliable and truthful nutrition education information regarding weight management (Seagle et al., 2009) and possess the skills and knowledge to encourage individuals throughout their weight management efforts. Including registered dietitians in future programmes may provide additional means of support, as well as access to accurate nutrition information.

Portion distortion was an overwhelmingly common phenomenon in the present study sample, as well as in other segments of the US population. Once women recognised their own 'portion distortion', implementing reduced portion sizes and portion control of foods promoted weight loss (Ello-Martin et al., 2005; Seagle et al., 2009; Sciamanna et al., 2011). Additionally, individuals who consume smaller portion sizes are generally more successful in maintaining WL

and achieving WLM (Wing & Hill, 2001; Elfhag & Rossner, 2005; Sciamanna et al., 2011). These results are consistent with those of Jeffery et al. (1993), who documented that consuming portion-controlled servings aided in WLM and with those of Hannum et al. (2004) who observed significant WL in individuals consuming portion-controlled entrees. Furthermore, individuals in a previous qualitative study reported portion control as a common strategy for WLM (Reyes et al., 2012). Along with consuming appropriate portions, regularly consuming breakfast and making healthy snack choices were noted as positive changes in the eating patterns of women in the present study. These findings are consistent with others (Elfhag & Rossner, 2005; Wing & Phelan, 2005; Seagle et al., 2009). Although a standard definition for snack (Gregori & Maffeis, 2007; Gregori et al., 2011) has not been agreed upon, being able to distinguish between snacks and snack foods proved to be beneficial in controlling weight and satisfying nutrient and food group requirements. A recent cross-sectional analysis found snacking to be positively associated with overall diet quality and a more nutrient-dense diet (Zizza & Xu, 2012). Along with portion control strategies, flexible approaches to eating should be encouraged in future interventions and dietary programmes.

Regular participation in exercise has been associated with long-term WLM and has been identified as a common strategy of National Weight Control Registry participants for WL and WLM (Kayman et al., 1990; Wing & Hill, 2001; Elfhag & Rossner, 2005; Sciamanna et al., 2011). The primary intervention did not include an exercise component; however, following the intervention, women recognised the importance of physical activity in maintaining WL. Women who reported the most success at WLM adopted some form of exercise, whereas women who were not successful recognised that a lack of exercise may have played a role in weight regain. Barriers to exercise, including guilt, a lack of time and unenjoyable activities, are consistent with those reported by Thomas et al. (2008) and represent factors that should be considered in future interventions.

Weight loss is generally a function of time, whereas WLM is the result of sustained actions over time (Reyes et al., 2012). Women recognised that 'quick fixes' and fad diets were not sustainable and did not produce long-term results; rather, women recognised that lifestyle changes were necessary to achieve long-term WLM. This finding is consistent with the study of

Herriot et al. (2008), who reported that shifting from a dieting attitude to a permanent lifestyle change was predictive of successful WL. Lifestyle modifications that include sustainable diet, exercise and behaviour therapy components have been shown to produce significant WL and aid in long-term weight control (Lyznicki et al., 2001; Wadden et al., 2004) and should be included in future trials.

A novel finding of the present study is that women never reported that selfaccountability was important to them or that self-monitoring of body weight was central to WLM. This finding is in contrast to practices of National Weight Control Registry members, who use self-weighing as a form of self-monitoring and self-accountability as a means for motivation to achieve long-term WLM (Wing et al., 2006). In the present study, all women used the same approach to induce WL and self-monitoring was not emphasised. Inclusion in the National Weight Control Registry and the study by Wing et al. (2006) was based on WL and WLM standards, regardless of the approach used to achieve WL and WLM. Some of these interventions may have encouraged self-weighing and monitoring and self-accountability. Group-based interventions, such as in the present study, appear to provide an important source of support for healthy eating and physical activity behaviours (Wang et al., 2014); thus, social support from study participants and accountability to external others may have been more important than self-accountability or self-monitoring of body weight. Only a few women in the present study reported using clothing fit as a means of self-monitoring of body weight, a practice commonly reported in other studies (Klem et al., 1997; Wing & Phelan, 2005; Hindle & Carpenter, 2011; Reyes et al., 2012). Different self-regulation and self-monitoring techniques may be necessary to assist individuals in their WL and WLM efforts.

The limitations of the present study include the use of a convenience sample and limited generalisability to other populations. Although these results may not be generalisable to other populations, they do provide understanding and insight into experiences and perceptions of participants. Large focus groups were not conducted as a result of the restricted number of participants; however, the topic of interest was still explored in-depth among this small homogenous sample of women who participated in the same WL intervention. Although successful at achieving WL during the primary intervention, most participants were not

successful at WLM. Additional research is warranted to further explore psychosocial and behavioural characteristics of optimal WL and WLM strategies through programmes and interventions that will assist women with their weight management journeys.

In conclusion, the qualitative findings of the present study indicate that perceived accountability to others, social support, planning ahead, awareness and mindfulness of food choices, basic nutrition education, portion control, exercise and self-motivation all facilitated WL and WLM. Barriers included life transitions, health status changes, internal factors, environmental pressures, lack of accountability to others and an absence of social support. Because women considered WLM to be more difficult than WL, as demonstrated by the fact that the overall sample was not successful at WLM, the different strategies necessary to achieve WL and WLM should be acknowledged. Therefore, future interventions should incorporate WLM strategies for the prevention of undesired weight gain or weight regain following WL. The identification of facilitators and barriers to WL and WLM provides direction for future interventions and programmes that promote robust and sustainable behaviour changes for women during their weight management journeys.

Acknowledgments

This work was performed at The Pennsylvania State University, University Park, PA, USA.

Conflict of interests, source of funding and authorship

SMN-R discloses research funding from The Hershey Company; the Bell Institute of Health and Nutrition, General Mills, Inc.; Dairy Research Institute; and the United States Department of Agriculture. Research funding provided to SMN-R is unrelated to the present study, with the exception of a grant from The Hershey Company, which provided support for the present study. AGP and DLM are employed by The Hershey Company.

The Hershey Company (Hershey, PA, USA) funded this qualitative study through a research grant provided to the senior author and a Graduate Research Fellowship to the primary author.

CJM and SNR designed the study, collected data and drafted the paper. All authors analysed and/or interpreted data, and also critically reviewed the manuscript and approved the final version submitted for publication.

References

- Ball, K., Brown, W. & Crawford, D. (2002) Who does not gain weight? Prevalence and predictors of weight maintenance in young women. Int. J. Obes. Relat. Metab. Disord. 26, 1570–1578.
- Barnes, A.S., Goodrick, G.K., Pavlik, V., Markesino, J., Laws, D.Y. & Taylor, W.C. (2007) Weight loss maintenance in African-American women: focus group results and questionnaire development. J. Gen. Intern. Med. 22, 915–922.
- Befort, C.A., Thomas, J.L., Daley, C.M., Rhode, P.C. & Ahluwalia, J.S. (2008) Perceptions and beliefs about body size, weight, and weight loss among obese African American women: a qualitative inquiry. Health Educ. Behav. 35, 410–426.
- Braun, V. & Clarke, V. (2006) Using thematic analysis in psychology. Qual. Res. Psychol. 3, 77– 101.
- Burke, L., Wang, J. & Sevick, M.A. (2011) Self-monitoring in weight loss: a systematic review of the literature. J. Am. Diet. Assoc. 111, 92–102.
- Byrne, S., Cooper, Z. & Fairburn, C. (2003) Weight maintenance and relapse in obesity: a qualitative study. Int. J. Obes. Relat. Metab. Disord. 27, 955–962.
- Creswell, J.W. (1998) Qualitative Inquiry and Research Design: Choosing Among Five Traditions. Thousand Oaks, CA: Sage Publications.
- Elfhag, K. & Rossner, S. (2005) Who succeeds in maintaining weight loss? A conceptual review of factors associated with weight loss maintenance and weight regain. Obes. Rev. 6, 67–85.
- Ello-Martin, J.A., Ledikwe, J.H. & Rolls, B.J. (2005) The influence of food portion size and energy density on energy intake: implications for weight management. Am. J. Clin. Nutr. 81, 236S–241S.
- Ely, A.C., Befort, C., Banitt, A., Gibson, C. & Sullivan, D. (2009) A qualitative assessment of weight control among rural Kansas women. J. Nutr. Educ. Behav. 41, 207–211.
- Foster-Schubert, K.E., Alfano, C.M., Duggan, C.R., Xiao, L., Campbell, K.L., Kong, A., Bain, C.E., Wang, C.Y., Blackburn, G.L. & McTiernan, A. (2012) Effect of diet and exercise, alone or combined, on weight and body composition in overweight-to-obese postmenopausal women. Obesity (Silver Spring) 20, 1628–1638.

- Goodrick, G.K., Poston, W.S. & Foreyt, J.P. (1996) Methods for voluntary weight loss and control: update. Nutrition 12, 672–676.
- Green, A.R., Larkin, M. & Sullivan, V. (2009) Oh stuff it! The experience and explanation of diet failure: an exploration using interpretative phenomenological analysis. J. Health Psychol. 14, 997–1008.
- Greenbaum, T.L. (1998) The Handbook for Focus Group Research. Thousand Oaks, CA: Sage Publications.
- Gregori, D. & Maffeis, C. (2007) Snacking and obesity: urgency of a definition to explore such a relationship. J. Am. Diet. Assoc. 107, 562–563.
- Gregori, D., Foltran, F., Ghidina, M. & Berchialla, P. (2011) Understanding the influence of the snack definition on the association between snacking and obesity: a review. Int. J. Food Sci. Nutr. 62, 270–275.
- Hannum, S.M., Carson, L., Evans, E.M., Canene, K.A., Petr, E.L., Bui, L. & Erdman, J.W. Jr (2004) Use of portion-controlled entrees enhances weight loss in women. Obes. Res. 12, 538–546.
- Hardcastle, S. & Hagger, M.S. (2011) 'You can't do it on your own': experiences of a motivational interviewing intervention on physical activity and dietary behaviour. Psychol. Sport Exerc. 12, 314–323.
- Harris, J.E., Gleason, P.M., Sheean, P.M., Boushey, C., Beto, J.A. & Bruemmer, B. (2009) An introduction to qualitative research for food and nutrition professionals. J. Am. Diet. Assoc. 109, 80–90.
- Herriot, A.M., Thomas, D.E., Hart, K.H., Warren, J. & Truby, H. (2008) A qualitative investigation of individuals' experiences and expectations before and after completing a trial of commercial weight loss programmes. J. Hum. Nutr. Diet. 21, 72–80.
- Hindle, L. & Carpenter, C. (2011) An exploration of the experiences and perceptions of people who have maintained weight loss. J. Hum. Nutr. Diet. 24, 342–350.
- James, D.C., Pobee, J.W., Oxidine, D., Brown, L. & Joshi, G. (2012) Using the health belief model to develop culturally appropriate weight-management materials for African-American women. J. Acad. Nutr. Diet. 112, 664–670.

- Jeffery, R.W., Wing, R.R., Thorson, C., Burton, L.R., Raether, C., Harvey, J. & Mullen, M. (1993) Strengthening behavioral interventions for weight loss: a randomized trial of food provision and monetary incentives. J. Consult. Clin. Psychol. 61, 1038–1045.
- Jeffery, R.W., Drewnowski, A., Epstein, L.H., Stunkard, A.J., Wilson, G.T., Wing, R.R. & Hill, D.R. (2000) Long-term maintenance of weight loss: current status. Health Psychol. 19, 5–16.
- Kayman, S., Bruvold, W. & Stern, J.S. (1990) Maintenance and relapse after weight loss in women: behavioral aspects. Am. J. Clin. Nutr. 52, 800–807.
- Klem, M.L., Wing, R.R., McGuire, M.T., Seagle, H.M. & Hill, J.O. (1997) A descriptive study of individuals successful at long-term maintenance of substantial weight loss. Am. J. Clin. Nutr. 66, 239–246.
- Krueger, R.A. (1998) Analyzing and Reporting Focus Group Results. Thousand Oaks, CA: Sage Publications.
- Krueger, R.A. & Casey, M.A. (2009) Focus Groups: A Practical Guide for Applied Research.Thousand Oaks, CA: Sage Publications.
- Lyznicki, J.M., Young, D.C., Riggs, J.A., Davis, R.M. & Council on Scientific Affairs, American Medical Association (2001) Obesity: assessment and management in primary care. Am. Fam. Physician 63, 2185–2196.
- Morse, J.M. & Field, P.A. (1995) Qualitative Research Methods for Health Professionals, 2nd edn. Thousand Oaks, CA: Sage Publications.
- Nickols-Richardson, S.M., Piehowski, K.E., Metzgar, C.J., Miller, D.L. & Preston, A.G. (2014) Changes in body weight, blood pressure and selected metabolic biomarkers with an energyrestricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr. Res. Pract. 8.
- Ogden, J. (2000) The correlates of long-term weight loss: a group comparison study of obesity. Int. J. Obes. Relat. Metab. Disord. 24, 1018–1025.
- Reyes, N.R., Oliver, T.L., Klotz, A.A., Lagrotte, C.A., Vander Veur, S.S., Virus, A., Bailer, B.A. & Foster, G.D. (2012) Similarities and differences between weight loss maintainers and regainers: a qualitative analysis. J. Acad. Nutr. Diet. 112, 499–505.

- Sciamanna, C.N., Kiernan, M., Rolls, B.J., Boan, J., Stuckey, H., Kephart, D., Miller, C.K., Jensen, G., Hartman, T.J., Loken, E., Hwang, K.O., Williams, R.J., Clark, M.A., Schubart, J.R., Nezu, A.M., Lehman, E. & Dellasega, C. (2011) Practices associated with weight loss versus weight-loss maintenance: results of a national survey. Am. J. Prev. Med. 41, 159–166.
- Seagle, H.M., Strain, G.W., Makris, A. Reeves, R.S., American Dietetic Association (2009)
 Position of the American Dietetic Association: weight management. J. Am. Diet. Assoc. 10, 330–346.
- Stevens, J., Truesdale, K.P., McClain, J.E. & Cai, J. (2006) The definition of weight maintenance. Int. J. Obes. (Lond) 30, 391–399.
- Stewart, D.W., Shamdasani, P.N. & Rook, D.W. (2007) Focus Groups: Theory and Practice, 2nd edn. Thousand Oaks, CA: Sage Publications.
- Stunkard, A. & McLaren-Hume, M. (1959) The results of treatment for obesity: a review of the literature and report of a series. AMA Arch. Intern. Med. 103, 79–85.
- Svetkey, L.P., Stevens, V.J., Brantley, P.J., Appel, L.J., Hollis, J.F., Loria, C.M., Vollmer, W.M.,
 Gullion, C.M., Funk, K., Smith, P., Samuel-Hodge, C., Myers, V., Lien, L.F., Laferriere, D.,
 Kennedy, B., Jerome, G.J., Heinith, F., Harsha, D.W., Evans, P., Erlinger, T.P., Dalcin, A.T.,
 Coughlin, J., Charleston, J., Champagne, C.M., Bauck, A., Ard, J.D. Aicher, K. & Weight Loss
 Maintenance Collaborative Research Group (2008) Comparison of strategies for sustaining
 weight loss: the weight loss maintenance randomized controlled trial. JAMA 299, 1139–1148.
- Tessaro, I., Rye, S., Parker, L., Trangsrud, K., Mangone, C., McCrone, S. & Leslie, N. (2006)
 Cookin' Up Health: developing a nutrition intervention for a rural Appalachian population.
 Health Promot. Pract. 7, 252–257.
- Thomas, S.L., Hyde, J., Karunaratne, A., Kausman, R. & Komesaroff, P.A. (2008) 'They all work ... when you stick to them': a qualitative investigation of dieting, weight loss, and physical exercise, in obese individuals. Nutr. J. 7, 34.
- Wadden, T.A., Butryn, M.L. & Byrne, K.J. (2004) Efficacy of lifestyle modification for long-term weight control. Obes. Res. 12, 151S–162S.

- Wane, S., van Uffelen, J.G. & Brown, W. (2010) Determinants of weight gain in young women: a review of the literature. J. Women's Health (Larchmt) 19, 1327–1340.
- Wang, M.L., Pbert, L. & Lemon, S.C. (2014) Influence of family, friend, and coworker social support and social undermining on weight gain prevention among adults. Obesity (Silver Spring) 22, 1973–1980 Jun 19. doi: 10.1002/oby.20814.
- Wansink, B. (2010) From mindless eating to mindlessly eating better. Physiol. Behav. 100, 454–463.
- Williamson, D.F., Kahn, H.S., Remington, P.L. & Anda, R.F. (1990) The 10-year incidence of overweight and major weight gain in US adults. Arch. Intern. Med. 150, 665–672.

Wing, R.R. & Hill, J.O. (2001) Successful weight loss maintenance. Annu. Rev. Nutr. 21, 323–341.

- Wing, R.R. & Phelan, S. (2005) Long-term weight loss maintenance. Am. J. Clin. Nutr. 82, 222S– 225S.
- Wing, R.R., Tate, D.F., Gorin, A.A., Raynor, H.A. & Fava, J.L. (2006) A self-regulation program for maintenance of weight loss. N. Engl. J. Med. 355, 1563–1571.

Zizza, C.A. & Xu, B. (2012) Snacking is associated with overall diet quality among adults. J. Acad. Nutr. Diet. 112, 291–296.

Zung, W.W.K. (1986) Zung self-rating depression scale and depression status inventory. In Assessment of Depression. eds N. Sartorius & T.A. Ban, pp. 221–231. Berlin: Springer.

Table 3.4 Focus group questions used during a qualitative study exploring weight loss andweight-loss maintenance in 23 women from a primary weight-loss study

Question	Theoretical model	Behavior change variable
 Very briefly, would anyone like to share about their weight loss journey?* 		
2. How was your experience with weight loss in this study different from previous experiences of managing your weight?*	Theory of planned behavior	Attitudes
3. If you learned any strategies during this weight loss experience that have helped you maintain weight loss, please describe these strategies.	Theory of planned behavior Social cognitive theory	Skills Self-efficacy
4. Explain your success in maintaining weight loss or experiencing further weight loss.	Theory of planned behavior Social cognitive theory	Skills Self-efficacy
5. Explain any difficulties in terms of maintaining your weight loss or achieving further weight loss.*	Health belief model	Barriers
6. How do you think a close friend would describe you/your experience with weight management/weight loss/weight regain?*	Theory of planned behavior	Subjective norms
7. If you were writing a self-help book about managing your weight, or advising a friend, what would be the key points you would want to make?*	Theory of planned behavior	Skills; Subjective norms
8. How has your change in lifestyle and/or weight affected those close to you?*	Theory of planned behavior	Subjective norms
9. Have your eating patterns changed since participation in the study?	Theory of planned behavior Social cognitive theory Health belief model	Attitudes Outcome expectations Barriers
10. How do you feel about snacks since your experience with weight loss?	Social cognitive theory	Outcome expectations

*Hindle & Carpenter (2011)

Table 3.5 Descriptive statistics of women participating in focus group discussions compared to nonparticipants

Characteristic	Focus group participants: current [†]	Focus group participants: post- intervention [‡]	Nonparticipants§
Age (years)	38.8 (6.0)	35.7 (5.8)	39.0 (5.6)
Height (cm)	164.6 (4.9)	164.6 (4.9)	165.0 (6.6)
Weight (kg)	80.7 (17.4)	77.9 (16.7)	79.0 (12.4)
Body mass index (kg/m ²)	30.0 (6.2)	28.7 (5.5)*	29.0 (4.1)
Waist circumference (cm)	88.4 (15.0)	83.0 (12.0)***	85.7 (9.5)
Hip circumference (cm)	113.3 (11.5)	110.1 (11.9)*	110.0 (9.5)

Data are reported as the mean ± SD.

*P < 0.05; **P < 0.01; ***P < 0.001 for comparison of current to week 18 measurements within focus group participants, using paired t-tests.

[†]Data collected during the day of focus group session; participants (n = 23) included those engaging in focus group discussions.

^{*}Data collected at week 18 of the primary intervention study (post-intervention); participants (n = 23) included those engaging in focus group discussions.

[§]Data collected at week 18 of the primary intervention study (post-intervention); nonparticipants (n = 28) included those who declined to participate (n = 12) and those lost to follow-up after the primary study (n = 16).

Table 3.6 Summary of major themes (mentioned by at least three participants across groups), interpretation of meanings and considerations for future interventions regarding facilitators and barriers to weight loss and weight loss maintenance as perceived by a group of 23 women who participated in focus group sessions after completing a primary weight loss study

Themes	Interpretations	Considerations for future interventions
Weight loss journey	Women perceive their weight loss journeys to be continuous processes that include the use of various approaches to induce weight loss or to manage weight. Biological and physiological challenges to weight loss and weight loss maintenance exist for women. Women desire to prevent weight gain across life transitions.	Tailor programs to individual needs, given a woman's biology, physiology and life stage.
Accountability and support	External accountability is preferred during weight loss and weight loss maintenance. A high level of support is needed from many social sectors during weight loss maintenance.	Incorporate external accountability strategies and protocols into interventions. Provide and/or build social support networks in programs for weight loss and weight loss maintenance.
Planning ahead, mindfulness and awareness	Planning is an important aspect for achieving successful weight loss and weight loss maintenance.	Incorporate planning skills into interventions that are practical, simple and provide structure.
	Planning skills take time to develop and maintain. Planning is inherently linked to mindfulness and awareness of food choices and energy intake.	Practice and reinforce planning skills during programs to enhance mindfulness of food intake and awareness of energy and nutritional content of foods as part of daily caloric intake.

Table 3.6 (cont.)

Nutrition education	Group nutrition education sessions and discussions provide valuable knowledge, necessary to making healthier food choices. Credibility of the registered dietitian as an expert is valued; access to a credible source of information post-intervention is desired.	Include routine education classes, led by a registered dietitian, for dissemination of accurate nutrition information. Offer registered dietitian services during weight loss and weight loss maintenance interventions.
Portion control	Regulation of portion sizes and recognition and avoidance of portion distortion is a major facilitator of weight loss and weight loss maintenance.	Incorporate portion control strategies and education into interventions.
Exercise	Regular exercise facilitates weight loss maintenance. Lack of exercise is a key barrier to success with weight loss maintenance.	Incorporate preferred and tailored exercise routines into interventions; provide appropriate technical supervision as needed.
Motivation	Preventing minor dietary indiscretions from turning into large relapses of weight gain is critical for weight loss and weight loss maintenance.	Create and discuss scenarios that test self-control, willpower and self-regulation for relapse prevention. Discuss self- forgiveness for noncompliance.
	Self-motivation plays a key role in weight loss maintenance. Identification of readiness for change is helpful to predict weight loss and weight loss maintenance success.	Include skills and strategies for maintaining motivation and sustaining change.
Total lifestyle change	Avoiding a "dieting" or "quick fix" mentality and shifting toward a total lifestyle approach for weight loss and weight loss maintenance is crucial to success.	Include strategies for sustainable dietary and physical activity modifications in interventions.

Table 3.6 (cont.)

Eating patterns and	Women perceive healthy snacks to	Encourage flexible and
snacking	be a necessary component of	accommodating approaches to
	dietary approaches for weight	eating, including regular
	management.	consumption of breakfast,
		healthy snacks and a consistent
	A consistent eating pattern is	eating pattern.
	found to be valuable in weight loss	
	and weight loss maintenance.	

Chapter 4 describes the materials and methods for the current dissertation research which focuses on weight gain prevention in young adult and midlife women. The current research emphasizes prevention of weight gain, rather than treatment of excess body mass by weight loss, as few individuals are successfully able to lose and maintain weight loss over the long term.

These materials and methods are grounded in findings from preliminary research as well as methodology from previous studies examining weight gain prevention in adult populations (1). The current study is unique in that it focuses only on women, but includes normal weight, overweight and obese women. Previous weight gain prevention studies that have included women have studied normal weight and overweight women *or* overweight and obese women.

This study is also unique in that it employs interactive nutrition education classes as the intervention to provide basic nutrition education and knowledge to assist participants in preventing weight gain over the course of one year. Additionally, this study compares the education classes delivered by a trained nutrition professional (i.e. a registered dietitian) with individuals matched for gender and age with no formal nutrition training.

Expected results of the study are presented in this chapter; major findings and results from this research are presented in Chapters 5, 6 and 7.

Reference

 This chapter previously appeared in its entirety as Metzgar CJ, Nickols-Richardson SM. Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial. JMIR Res Protoc. 2015;4(1):e36. doi:10.2196/resprot.4008. This is an open-access article (http://www.researchprotocols.org/2015/1/e36/) distributed under the terms of the Creative Commons Attribution License and the authors retain copyright.

Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial³

Abstract

Background: Treatment of overweight and obesity through body weight reduction has been monumentally ineffective as few individuals are able to sustain weight loss. Rather than treating weight gain once it has become problematic, prevention of weight gain over time may be more effective.

Objective: The aim of this research is to preclude the burden of adult obesity in women by identifying the determinants of weight gain prevention. The objective of this randomized controlled trial (RCT) is to compare a weight gain prevention intervention delivered by the registered dietitian versus counselor.

Methods: This is a 12-month parallel-arm weight gain prevention RCT designed to increase selfefficacy, self-regulation, outcome expectations and family and social support through the use of a nutrition education intervention in women, aged 18-45 years, from the Urbana-Champaign (Illinois, USA) area. Women have been randomized to registered dietitian, counselor or wait-list control groups (August 2014) and are undergoing weekly nutrition education sessions for four months, followed by monthly sessions for eight months (through August 2015). Outcome measures, including: (1) dietary intake, (2) physical activity, (3) anthropometric and blood pressure measurements, (4) biochemical markers of health, (5) eating behaviors and health perceptions, and (6) mediators of behavior change, were collected before the intervention began (baseline) and will be collected at 3, 6, 9, and 12 months of the study. **Results:** In total, 87 women have been randomized to intervention groups, and 81 women have completed first week of the study. Results are expected in early 2016. **Conclusions:** This RCT is one of the first to examine weight gain prevention in women across

normal, overweight, and obese body mass index categories. Results of this research are

³This chapter previously appeared in its entirety as Metzgar CJ, Nickols-Richardson SM. Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial. *JMIR Res Protoc.* 2015;4(1):e36. doi:10.2196/resprot.4008. This is an open-access article (http://www.researchprotocols.org/2015/1/e36/) distributed under the terms of the Creative Commons Attribution License and the authors retain copyright.

expected to have application to evidence-based practice in weight gain prevention for women and possibly have implication for policy regarding decreasing the encumbrance of overweight and obesity in the United States.

(JMIR Res Protoc 2015;4(1):e36) doi:10.2196/resprot.4008

KEYWORDS

body weight; weight gain prevention; weight maintenance; women

Introduction

Adult Weight Management

Small weight gains over time, around 1-2 pounds per year [1,2], contribute to the development of overweight and obesity. Once established, obesity is difficult to treat [3], as reduction of excess body weight is rarely effective in the long term. Short-term weight loss can be achieved by a variety of methods, but few of these approaches are sustainable and effective in facilitating permanent weight loss [4-10]. On average, individuals adhere to weight loss programs for approximately six months [11]; following weight loss, most individuals regain half of the weight lost within one year, and return to baseline weight within 3-5 years [11-13]. Weight gain prevention, on the other hand, avoids the difficulties that may accompany weight loss and its maintenance and offers an alternative option for weight management.

To reduce disease risk and improve overall health, effective weight gain prevention is essential; however, few interventions have successfully examined weight gain prevention and little is known about the determinants of and strategies for preventing weight gain over the long term. Much of the existing research has focused on treatment of overweight and obesity through reduction of excess body weight [14,15] or prevention of weight regain following weight loss [16-20].

Weight Gain Prevention

In the first weight gain prevention trial, normal weight adults, aged 25-74 years, were randomized to an untreated control group or a treatment group that received monthly newsletters plus a financial incentive for weight maintenance for one year [21]. The treatment group experienced an average weight loss of 1 kg, which was significantly different from the control group; with the treatment effect being stronger in men than women [21]. Building upon the Pound of Prevention (POP) work, 3-year weight gain prevention in adults, aged 20-45 years, was investigated [22,23]. Participants were randomized to a no-contact control group or to one of two education groups that received nutrition education via monthly newsletters and semiannual nutrition and exercise classes. One education group received a lottery incentive for

participation. Significant differences in weight gain between the control and education groups were not found, although weight-related behaviors did improve in participants receiving education [22,23].

The Shape Program was a medium-intensity behavioral intervention in overweight and class I obese premenopausal black women that included weekly self-monitoring, monthly counseling calls, tailored skills training and a YMCA gym membership and was compared to usual care that included newsletters covering general wellness topics every six months during the 18-month study [24]. After one year, weight loss was significantly greater in the intervention group; these changes were sustained at 18 months. No significant differences in waist circumference, blood pressure, glucose or lipid levels between the intervention and usual care groups were observed at any measurement point during the study [24]. Levine and colleagues [25] randomized normal weight and overweight women to a clinic-based group, a correspondence group or a control group for 24 months. During three years, the intervention had no influence on weight gain in either group; however, age, dieting status, and feelings of hunger were found to be predictive of weight gain.

The Groningen Overweight and Lifestyle (GOAL) study examined weight gain prevention in overweight and obese men and women with hypertension and/or dyslipidemia in the Netherlands by comparing the effects of lifestyle counseling by a nurse practitioner to usual care from a general practitioner during a 1-year period [26]. No significant differences were observed in weight change between groups at one year or after three years [26,27]. Study of Novel Approaches to Weight Gain Prevention (SNAP) is the most recently published intervention [28]. Two novel self-regulation approaches to weight gain prevention—small consistent changes and large periodic changes—were compared to a minimal treatment control for an average of three years of follow-up. Results of this study have not yet been published [28]. Without complete knowledge of the determinants of and strategies for weight gain prevention, public health will remain at risk for complications and costs related to overweight

and obesity. Weight gain prevention offers a primary strategy for weight management and obesity prevention [24,26].

Women who previously participated in a weight-loss intervention identified gender-specific life transitions and stressors, including pregnancy, post pregnancy, family responsibilities, health status changes, and aging as precipitators of weight gain [29]. Young adulthood and perimenopause appear to be critical intervals for weight gain [30-32]; therefore, weight gain prevention efforts should target these lifespan stages, specifically in women.

Aims and Objectives

The current study aims to identify determinants of weight gain prevention in young adult and midlife premenopausal women through a 1-year weight gain prevention intervention that includes nutrition education. We hypothesize that compared to a wait-list control group, women who participate in a weight gain prevention intervention designed to increase self-efficacy, self-regulation, outcome expectations, and family and social support will maintain current body weight during a 12-month period. It is further hypothesized that women in an intervention group led by registered dietitians will have less weight gain during 12 months compared to women in an intervention group led by counselors.

Methods

Recruitment, Screening, and Enrollment

Participants were recruited by word-of-mouth, electronic mail messages, and posted flyers from the University of Illinois campus and the Urbana-Champaign (IL, USA) communities. A flow diagram of response, screening, and randomization steps is displayed in Figure 4.1. A total of 330 women responded to recruitment methods, between June and August 2014. Of these, 266 women met prescreening criteria (appropriate age, body mass index [BMI], and desire to prevent weight gain) and received screening materials including a medical history form, Zung Self-Rating Depression Scale/Status Inventory [33], and informed consent. A total of 146 women returned screening materials, which were reviewed by investigators. One hundred two

women met eligibility criteria for participation, and 87 women were randomized, with 81 women completing baseline testing.

The current study included premenopausal women between the ages of 18-45 years with a BMI of >18.5 kg/m². There were no additional criteria for body weight and BMI to ensure participation by women from a range of weight status categories. Further inclusion criteria included eumenorrhea (≥8 menstrual cycles/year), score of <50 on the Zung Self-Rating Depression Scale/Status Inventory [33], and no self-reported metabolic, cardiovascular or musculoskeletal diseases or use of medications or supplements to manage a chronic health condition. Exclusion criteria included women who currently smoked, were pregnant or attempting to become pregnant or were currently lactating. Women using medications influencing weight regulation, such as steroid or thyroid hormones or oral contraceptives, were excluded if use was for <2 months before the start of the study. Gastric bypass surgery was also an exclusion criterion.

The Institutional Review Board (IRB) for the protection of human subjects at the University of Illinois at Urbana-Champaign (UIUC) approved the study protocol (UIUC IRB#14397). Each participant provided written informed consent before study participation.

Study Design

The current study is a 12-month parallel-arm weight gain prevention randomized controlled trial. After enrollment, women were randomized to one of three intervention groups: (1) weight gain prevention intervention delivered by a registered dietitian (RDG); (2) weight gain prevention intervention delivered by a counselor (CSG), or (3) wait-list control (CON) group. The RDG and CSG weight gain prevention interventions are identical in materials and content; the only difference is the credentialing of the individuals leading the intervention. Women in the CON group receive no intervention; upon completion of the 12-month waiting period, these women will be randomized to the RDG group or CSG group and will receive the respective intervention for the next 12-month period.

Intervention

During the 1-year study, women randomized to the RDG and CSG groups will attend a total of 24, 1-hour nutrition education sessions that are based on effective weight-loss programs/plans which address energy balance through sustainable diet, exercise, and behavior modifications [30,34-39]. These sessions will be held weekly for 16 weeks (months 1-4) and monthly thereafter (months 5-12) [30]. Vegetable consumption, planning ahead for food intake and portion control will be emphasized [34,35], and general nutrition information, eating away from home, food selection, food preparation, and recipe modification also will be addressed [34-37]. Other topics will include fitness and physical activity, culinary skills, breakfast consumption, healthy snacking and beverage choices, nutrient density, family menu planning, and grocery shopping. Problem solving, motivational concerns, and stress management will be encouraged [30,34-39]. Education sessions will relay constructs of the Social Cognitive Theory (SCT) [40].

Education sessions will follow a three-part format. Each session will begin with a brief review of information covered in the previous session and will address participant progress, including successes, challenges and questions. Next, the leader will deliver the nutrition education component of the session using an interactive group discussion format. Participants will be provided with handouts addressing food choices, dietary patterns, menu plans, and other information pertaining to the lesson. Finally, the content for the lesson will be summarized and participants will have a chance to ask questions, address concerns and set specific behavioral goals for the next session. Education sessions will be randomly selected for evaluation by a process observer who will rate the sessions based on investigator-established criteria.

Four registered dietitians will deliver the intervention to women in the RDG group. All women in the RDG group will equally interact with all four registered dietitians during the study. Four counselors will deliver the intervention to women in the CSG group, with these women having equal interaction with all four counselors across the study. The credentials of the professionals delivering the intervention will not be revealed to participants until after completion of the study. The registered dietitians are all female and have been practicing for <5 years. The

counselors are all female and are graduate teaching assistants at UIUC in programs unrelated to nutrition or dietetics. Compliance will be defined as attendance of >85% of education sessions. If women are unable to attend an education session, virtual make-up sessions will be offered, along with a quiz. Completion and return of the quiz will indicate that the materials were studied and reviewed and that the participant was compliant.

Outcome Measures

Before the intervention (baseline), data on dietary intake, physical activity, anthropometric, and blood pressure measurements, biochemical markers of health, eating behaviors and health perceptions, and SCT mediators of behavioral change were collected. These outcome measures also will be obtained at 3, 6, 9 and 12 months.

Dietary Intake and Physical Activity Assessment

Participants were taught to accurately complete 4-day food records and the Stanford 7-Day Physical Activity Recall Scale [41]. To ensure accuracy in recording foods and beverage consumption, handouts containing examples of standard serving sizes were provided. Participants recorded all food and beverages consumed for three non-consecutive weekdays and one weekend day before the baseline testing session [35]. Four-day food records will be analyzed using the Nutrition Data System for Research (NDSR) dietary analysis software (Nutrition Coordinating Center, Minneapolis, MN, USA) to estimate average daily dietary intake of total energy (kcal/day), carbohydrate (g/day), protein (g/day), fat (g/day), and fiber (g/day) in addition to intake by food groups (svgs/day).

For seven consecutive days before the baseline testing session, participants recorded the number of hours slept, spent in front of a television or computer screen, and engaged in moderate, hard, and very hard physical activity [35]. Participants wore accelerometers at the waist, wrist, or ankle during all waking hours for seven consecutive days while also recording physical activity to provide an objective assessment of energy expenditure. Approximately 70% of participants in each group wore accelerometers as they were not available for all individuals.

Physical activity records will be analyzed by summing total hours of moderate, hard, and very hard activity and dividing by seven to estimate hours of physical activity per day. These records will be further analyzed by converting activities into metabolic equivalents (METs) (hr/d), which will be evaluated as light activity (1-3 METs), moderate activity (>3-6 METs), and vigorous activity (>6 METs) to estimate the number of calories expended per day. Accelerometry data will be analyzed using ActiLife 6.11 (ActiGraph, Pensacola, FL, USA) to estimate the number of calories expended per day, the MET rate per day, and the length of time (minutes) spent in sedentary, light, moderate, vigorous, and very vigorous activities.

Anthropometric and Blood Pressure Measurements

Baseline standing height (cm) was recorded to the nearest 0.1 cm using a calibrated scalemounted stadiometer (Seca 700, Hanover, MD, USA). Body weight (kg) was measured using a calibrated scale (Tanita 410GS, Arlington Heights, IL, USA) to the nearest 0.1 kg. BMI (kg/m²) was calculated using height and body weight measurements. A retractable measuring tape (Gulik II, Country Technology, Inc, Gay Mills, WI) was used to measure waist (cm) and hip (cm) circumferences, in duplicate, to the nearest 0.1 cm according to standard protocol [34]. Waist circumference was measured at the narrowest point of the waist, approximately one inch above the navel, and hip circumference was measured at the widest part of the buttocks [35]. Waist and hip circumference measurements were averaged to obtain a single value for each site; these values were used to calculate the waist:hip ratio. Fat mass (FM; kg) and body fat percentage (BF%) were measured using a Tanita scale (410GS).

Seated systolic and diastolic blood pressures (mm Hg) were measured by a trained study investigator using a standard sphygmomanometer (Baumanometer[®] Desk Model, Copiague, NY, USA) following a 5-minute rest period. Blood pressure measurements were taken in duplicate with a 2-3-minute rest period between readings; mean systolic arterial pressure values and diastolic arterial pressure values will be used in data analyses. Resting heart rate was also measured after the 5-minute rest period.

Biochemical Markers of Health

Venous blood samples (~30 mL) were collected by a trained phlebotomist between 7:00 to 9:30 AM after a 12-hour fast. Whole blood samples were processed and stored at -80°C. Serum will be analyzed for concentrations of insulin, glucose, total cholesterol, high-density lipoprotein-cholesterol (HDL-C), low-density lipoprotein-cholesterol (LDL-C), triacylglycerides (TG), leptin, adiponectin, and resistin.

Serum insulin (µU/mL) (LINCO Research, St Charles, MO, USA) will be measured using enzymelinked immunosorbent assay (ELISA), and serum glucose (mg/dL) (Stanbio Labs, Boerne, TX, USA) will be measured by spectrophotometry. Total cholesterol (mg/dL), HDL-C (mg/dL) and TG (mg/dL) concentrations will be measured by spectrophotometry using the total cholesterol, HDL-C and TG kits, respectively (Stanbio Labs). Total cholesterol, HDL-C and TG concentrations will be used to calculate LDL-C concentration (mg/dL) using the equation: LDL-C=total cholesterol - HDL-C - (TG/5) [42]. Serum leptin (ng/mL), adiponectin (ng/mL), and resistin (ng/mL) will be measured using ELISA (R&D Systems, Minneapolis, MN, USA). All serum samples for each biomarker will be analyzed in duplicate at corresponding study intervals. Intra- and inter-assay coefficients of variations (CV) are <15% for all kits.

Eating Behaviors, Health Perceptions, and SCT Mediators of Behavioral Change

Participants completed questionnaires designed to evaluate eating behaviors, health perceptions, perseverance, and SCT mediators. The Eating Inventory [43] will evaluate ratings of cognitive eating restraint, hunger, and disinhibition. The Short-Form 36 Health Survey (SF-36) [44] will assess self-reported health issues. Perseverance will be examined using the Short Grit Scale (Grit-S) [45], and an investigator-designed questionnaire will evaluate SCT mediators, including self-efficacy, outcome expectations, self-regulation, and social and family support. Standard scoring and interpretation methods will be used to evaluate all questionnaires [43-45].

Statistical Analysis

Baseline characteristics of study participants were characterized using descriptive statistics: mean (SD). Participants in the three intervention groups (Treatment) will participate in five data collection sessions at specified intervals (Time). The Shapiro-Wilk test for normality will be used to test for normality and homogeneity of variance within groups; data will be transformed if necessary. Body weight, BMI, waist : hip ratio, FM, BF%, systolic and diastolic blood pressure, serum insulin, glucose, TC, HDL-C, LDL-C, TG, leptin, adiponectin, and resistin will be analyzed as dependent variables. Baseline variables that differ between groups will be included as covariates in the analysis. Dietary intake of macronutrients as estimated from 4-day food records, estimated energy expenditures, eating behaviors, health perceptions, and ratings of SCT mediators also will be compared among groups. A 3 x 5 (3 treatment groups x 5 time intervals) ANOVA with repeated measures on the time factor will be used to assess differences in outcomes within and between treatment groups over time. The group by time interaction will be examined for differences in time trend among intervention groups. Tukey pairwise comparisons will be used in conjunction with ANOVA to detect differences between treatment groups.

Some attrition is expected, as participants may be unable to comply with the intervention or may choose not to continue participation in the study. Participants who withdraw from the intervention will be asked to complete any remaining data collection sessions, and these data will be included in the statistical analyses (ie, intention-to-treat model). Data also will be analyzed using measurements only from those participants who complete all testing sessions. Statistical tests will be two-tailed with significance set at *P*<.01 to reduce the potential for Type I error. All statistical analyses will be conducted using Statistical Package for the Social Sciences (version 22.0, 2013, IBM Corp, Armonk, NY, USA).

Results

Eighty-one women completed baseline testing. Baseline descriptive characteristics of the sample are displayed in Tables 4.1 and 4.2. Overall, these women were highly educated, with a

majority of participants having at least a 4-year college degree. The racial/ethnic breakdown was reflective of the larger population, with non-Hispanic whites representing the majority. Age range was 18-45 years, and BMI range was 18.5-49.6 kg/m². On average, participants were overweight and normotensive. Participants have been recruited, enrolled, and randomized to one of the three intervention groups. Education sessions will continue through August 2015, and results are expected by early 2016.

Discussion

Principal Findings

The importance of weight gain prevention and maintenance of current weight has recently been recognized by the American College of Sports Medicine [18] and Healthy People 2020 [46] as critical; yet, there are currently no treatment guidelines for weight gain prevention. The gap in the understanding of the determinants, facilitators, and barriers to weight gain prevention likely exists due to the limited number of studies addressing prospective weight changes in adulthood. Awareness and identification of the determinants of weight gain prevention are necessary in order to increase the practicality of weight gain prevention for managing obesity. While it may seem counterintuitive to promote weight gain prevention in overweight and obese individuals rather than weight loss, weight gain prevention is relevant for individuals of all BMI categories [47]. Preventing weight gain over time offers the opportunity to slow the progression of overweight and obesity and to avoid further exacerbations related to excess body weight in individuals who are already overweight or obese [48]. Additionally, weight gain prevention may require less intensive treatment than that required to achieve weight loss [2], and may be more successful in the long term as it avoids the problems associated with weight loss and its maintenance [27]. Weight maintenance, regardless of whether an individual is normal weight, overweight or obese, may be more beneficial and practical than repeated, minimally successful weight-loss attempts. While modest weight losses of 5-10% of body weight have significant effects on risk factors of disease, these benefits may be ameliorated with weight regain. Even with weight loss, metabolically healthy obese individuals may not show improvement in health outcomes, and weight loss in these individuals may promote

weight cycling, or periods of weight loss followed by weight gain, which may have detrimental effects on mental, metabolic, and psychological outcomes [48-52]. Further, the adverse effects associated with weight cycling may be as harmful as maintenance of a high "unhealthy" body weight [51]. However, a recent study by Mason and colleagues [53] found that weight cycling was not associated with negative metabolic outcomes and a history of weight cycling was not related to the ability to lose and successfully maintain weight in the long term.

In a recent qualitative study of women who completed a weight-loss intervention conducted by a registered dietitian, women perceived the registered dietitian to be a credible source of nutrition information and found lack of access to a registered dietitian following completion of the intervention to be a barrier to weight-loss maintenance [29]. As a credible source of nutrition information [37], registered dietitians have a specialized skill set to support and encourage sustainable behavior changes to achieve weight management. Registered dietitians are generally regarded as the experts in weight management, but no studies have compared registered dietitians to other health professionals in the delivery of weight management information. The current study will fill this scientific gap by testing the ability of the registered dietitian to promote weight gain prevention as compared to an untrained professional. If registered dietitians are more effective in promoting weight gain prevention, these findings will support the notion that registered dietitians should be at the forefront in helping individuals attain successful weight management.

Women in the qualitative study [29] identified social support, basic nutrition education, accountability to others, self-motivation, mindfulness and awareness of food choices, planning ahead, portion control, and exercise as facilitators to weight loss and weight-loss maintenance while health status changes, environmental pressures, life transitions, absence of social support, lack of accountability, and internal factors were perceived as barriers. Additionally, women expressed their desire for continual contact with the registered dietitian as well as the group support offered by the education sessions [29].

Although there is no standardized definition of weight gain prevention or weight maintenance, a weight change of ±3% from baseline weight will be considered successful weight gain prevention. A 3% change criterion allows for normal day-to-day fluctuations that may result from measurement error, clothing, food consumption, and/or fluid balance [54].

Limitations

Results from this study will be limited in generalizability to premenopausal women. Future research should examine pre- and post-menopausal women, as these physiological changes appear to be other critical life stage intervals for weight gain. Weight gain prevention should also be examined exclusively in men, as determinants of and strategies for weight gain prevention may differ between men and women. Further, our results may be limited by the length of the study, as the current intervention is only for one year. Surveys, testing sessions or focus groups following completion of the intervention may be useful in order to garner more information about the feasibility of long-term weight gain prevention. There are limitations with using self-reported dietary intake and physical activity; however, participants have been taught to accurately complete food records, and accelerometry data will be used to validate written physical activity records. Finally, our intervention contains multiple components that address weight gain prevention, and the study design does not allow for examination of independent effects of the different elements of this weight gain prevention intervention. Investigator-designed surveys will be used to assist with determining the effects of individual intervention components.

Conclusions

The current study targets women who are at greater risk for weight gain compared to men; with the goal to help further the understanding of the determinants of weight gain prevention [29]. This study will fill a scientific gap in testing the ability of a registered dietitian to promote weight gain prevention as compared to another health professional that lacks formal nutrition and dietetics training. Although several studies have explored weight gain prevention with limited success, this may be the first study that targets young adult and midlife women of all

weight status categories (normal weight, overweight, obese) and focuses on prospective weight gain prevention. Results of this research will be expected to have implications for policy development and recommendations for decreasing the burden of overweight and obesity in the United States through weight gain prevention.

Acknowledgments

This research is supported by a grant from the United States Department of Agriculture, National Institute of Food and Agriculture, Illinois Agricultural Experiment Station, #ILLU-698-337. Graduate research assistantship support is provided by The Hershey Company, Hershey, PA, USA.

Authors' Contributions

CJM and SMN-R contributed equally to the conceptual development, research design, and drafting of the manuscript. CJM was the lead investigator for data collection and analysis, to date. Both authors contributed equally to data interpretation and critical revision of the manuscript.

Conflicts of Interest

CJM discloses Graduate Research Fellowship funding from The Hershey Company. SMN-R discloses research funding from The Hershey Company; the Bell Institute of Health and Nutrition, General Mills, Inc; Dairy Research Institute; and the United States Department of Agriculture. Research funding provided to SMN-R is unrelated to the present study.

Multimedia Appendix 1

CONSORT-EHEALTH checklist V1.6.2 [55].
References

- Williamson DF, Kahn HS, Remington PL, Anda RF. The 10-year incidence of overweight and major weight gain in US adults. Arch Intern Med 1990 Mar;150(3):665-672.
 [Medline]
- Truesdale KP, Stevens J, Lewis CE, Schreiner PJ, Loria CM, Cai J. Changes in risk factors for cardiovascular disease by baseline weight status in young adults who maintain or gain weight over 15 years: the CARDIA study. Int J Obes (Lond) 2006 Sep;30(9):1397-1407
 [FREE Full text] [CrossRef] [Medline]
- Gill TP. Key issues in the prevention of obesity. Br Med Bull 1997;53(2):359-388 [FREE Full text] [Medline]
- Hill JO, Thompson H, Wyatt H. Weight maintenance: what's missing? J Am Diet Assoc 2005 May;105(5 Suppl 1):S63-S66. [CrossRef] [Medline]
- Dansinger ML, Gleason JA, Griffith JL, Selker HP, Schaefer EJ. Comparison of the Atkins, Ornish, Weight Watchers, and Zone diets for weight loss and heart disease risk reduction: a randomized trial. JAMA 2005 Jan 5;293(1):43-53. [CrossRef] [Medline]
- Gardner CD, Kiazand A, Alhassan S, Kim S, Stafford RS, Balise RR, et al. Comparison of the Atkins, Zone, Ornish, and LEARN diets for change in weight and related risk factors among overweight premenopausal women: the A TO Z Weight Loss Study: a randomized trial. JAMA 2007 Mar 7;297(9):969-977. [CrossRef] [Medline]
- Shai I, Schwarzfuchs D, Henkin Y, Shahar DR, Witkow S, Greenberg I, Dietary Intervention Randomized Controlled Trial (DIRECT) Group. Weight loss with a low-carbohydrate, Mediterranean, or low-fat diet. N Engl J Med 2008 Jul 17;359(3):229-241. [CrossRef] [Medline]
- Makris A, Foster GD. Dietary approaches to the treatment of obesity. Psychiatr Clin North Am 2011 Dec;34(4):813-827 [FREE Full text] [CrossRef] [Medline]
- Sacks FM, Bray GA, Carey VJ, Smith SR, Ryan DH, Anton SD, et al. Comparison of weightloss diets with different compositions of fat, protein, and carbohydrates. N Engl J Med 2009 Feb 26;360(9):859-873 [FREE Full text] [CrossRef] [Medline]

- McGuire MT, Wing RR, Klem ML, Seagle HM, Hill JO. Long-term maintenance of weight loss: do people who lose weight through various weight loss methods use different behaviors to maintain their weight? Int J Obes Relat Metab Disord 1998 Jun;22(6):572-577. [Medline]
- 11. Goodrick GK, Poston WS, Foreyt JP. Methods for voluntary weight loss and control: update 1996. Nutrition 1996 Oct;12(10):672-676. [Medline]
- Jeffery RW, Drewnowski A, Epstein LH, Stunkard AJ, Wilson GT, Wing RR, et al. Long-term maintenance of weight loss: current status. Health Psychol 2000 Jan;19(1 Suppl):5-16.
 [Medline]
- 13. Byrne S, Cooper Z, Fairburn C. Weight maintenance and relapse in obesity: a qualitative study. Int J Obes Relat Metab Disord 2003 Aug;27(8):955-962. [CrossRef] [Medline]
- 14. Kumanyika SK, Obarzanek E, Stettler N, Bell R, Field AE, Fortmann SP, American Heart Association Council on Epidemiology and Prevention, Interdisciplinary Committee for Prevention. Population-based prevention of obesity: the need for comprehensive promotion of healthful eating, physical activity, and energy balance: a scientific statement from American Heart Association Council on Epidemiology and Prevention, Interdisciplinary Committee for Prevention (formerly the expert panel on population and prevention science). Circulation 2008 Jul 22;118(4):428-464 [FREE Full text] [CrossRef] [Medline]
- 15. Lee IM, Djoussé L, Sesso HD, Wang L, Buring JE. Physical activity and weight gain prevention. JAMA 2010 Mar 24;303(12):1173-1179 [FREE Full text] [CrossRef] [Medline]
- Reyes NR, Oliver TL, Klotz AA, Lagrotte CA, Vander Veur Stephanie S, Virus A, et al. Similarities and differences between weight loss maintainers and regainers: a qualitative analysis. J Acad Nutr Diet 2012 Apr;112(4):499-505. [CrossRef] [Medline]
- Lowe MR, Miller-Kovach K, Phelan S. Weight-loss maintenance in overweight individuals one to five years following successful completion of a commercial weight loss program. Int J Obes Relat Metab Disord 2001 Mar;25(3):325-331. [CrossRef] [Medline]
- 18. Donnelly JE, Blair SN, Jakicic JM, Manore MM, Rankin JW, Smith BK, American College of Sports Medicine. American College of Sports Medicine Position Stand. Appropriate

physical activity intervention strategies for weight loss and prevention of weight regain for adults. Med Sci Sports Exerc 2009 Feb;41(2):459-471. [CrossRef] [Medline]

- Wing RR, Hill JO. Successful weight loss maintenance. Annu Rev Nutr 2001;21:323-341.
 [CrossRef] [Medline]
- 20. McGuire MT, Wing RR, Hill JO. The prevalence of weight loss maintenance among American adults. Int J Obes Relat Metab Disord 1999 Dec;23(12):1314-1319. [Medline]
- 21. Forster JL, Jeffery RW, Schmid TL, Kramer FM. Preventing weight gain in adults: a pound of prevention. Health Psychol 1988;7(6):515-525. [Medline]
- 22. Jeffery RW, French SA. Preventing weight gain in adults: design, methods and one year results from the Pound of Prevention study. Int J Obes Relat Metab Disord 1997 Jun;21(6):457-464. [Medline]
- 23. Jeffery RW, French SA. Preventing weight gain in adults: the pound of prevention study.
 Am J Public Health 1999 May;89(5):747-751. [Medline]
- 24. Bennett GG, Foley P, Levine E, Whiteley J, Askew S, Steinberg DM, et al. Behavioral treatment for weight gain prevention among black women in primary care practice: a randomized clinical trial. JAMA Intern Med 2013 Oct 28;173(19):1770-1777 [FREE Full text] [CrossRef] [Medline]
- Levine MD, Klem ML, Kalarchian MA, Wing RR, Weissfeld L, Qin L, et al. Weight gain prevention among women. Obesity (Silver Spring) 2007 May;15(5):1267-1277 [FREE Full text] [CrossRef] [Medline]
- 26. ter Bogt NC, Milder IE, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, et al. Changes in lifestyle habits after counselling by nurse practitioners: 1-year results of the Groningen Overweight and Lifestyle study. Public Health Nutr 2011 Jun;14(6):995-1000. [CrossRef] [Medline]
- 27. ter Bogt NC, Bemelmans Wanda JE, Beltman FW, Broer J, Smit AJ, van der Meer Klaas. Preventing weight gain by lifestyle intervention in a general practice setting: three-year results of a randomized controlled trial. Arch Intern Med 2011 Feb 28;171(4):306-313. [CrossRef] [Medline]

- 28. Wing RR, Tate D, Espeland M, Gorin A, LaRose JG, Robichaud EF, et al. Weight gain prevention in young adults: design of the study of novel approaches to weight gain prevention (SNAP) randomized controlled trial. BMC Public Health 2013;13:300 [FREE Full text] [CrossRef] [Medline]
- Metzgar CJ, Preston AG, Miller DL, Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. J Hum Nutr Diet 2014 Sep 18. [CrossRef] [Medline]
- Wing RR. Changing diet and exercise behaviors in individuals at risk for weight gain. Obes Res 1995 Sep;3 Suppl 2:277s-282s. [Medline]
- Ball K, Brown W, Crawford D. Who does not gain weight? Prevalence and predictors of weight maintenance in young women. Int J Obes Relat Metab Disord 2002 Dec;26(12):1570-1578. [CrossRef] [Medline]
- Wane S, van Uffelen Jannique G Z, Brown W. Determinants of weight gain in young women: a review of the literature. J Womens Health (Larchmt) 2010 Jul;19(7):1327-1340.
 [CrossRef] [Medline]
- Zung WWK, Sartorius N, Ban TA. Assessment of depression. In: Zung self-rating depression scale and depression status inventory. Berlin: Springer; 1986:221-231.
- 34. Piehowski KE, Preston AG, Miller DL, Nickols-Richardson SM. A reduced-calorie dietary pattern including a daily sweet snack promotes body weight reduction and body composition improvements in premenopausal women who are overweight and obese: a pilot study. J Am Diet Assoc 2011 Aug;111(8):1198-1203 [FREE Full text] [CrossRef] [Medline]
- 35. Nickols-Richardson SM, Piehowski KE, Metzgar CJ, Miller DL, Preston AG. Changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr Res Pract 2014 Dec;8(6):695-704 [FREE Full text] [CrossRef] [Medline]
- Lutes LD, Winett RA, Barger SD, Wojcik JR, Herbert WG, Nickols-Richardson SM, et al.
 Small changes in nutrition and physical activity promote weight loss and maintenance: 3-

month evidence from the ASPIRE randomized trial. Ann Behav Med 2008 Jun;35(3):351-357. [<u>CrossRef</u>] [<u>Medline</u>]

- 37. Slawson DL, Fitzgerald N, Morgan KT. Position of the Academy of Nutrition and Dietetics: the role of nutrition in health promotion and chronic disease prevention. J Acad Nutr Diet 2013 Jul;113(7):972-979. [CrossRef] [Medline]
- 38. Perri MG, Limacher MC, Durning PE, Janicke DM, Lutes LD, Bobroff LB, et al. Extendedcare programs for weight management in rural communities: the treatment of obesity in underserved rural settings (TOURS) randomized trial. Arch Intern Med 2008 Nov 24;168(21):2347-2354 [FREE Full text] [CrossRef] [Medline]
- 39. Knowler WC, Barrett-Connor E, Fowler SE, Hamman RF, Lachin JM, Walker EA, Diabetes Prevention Program Research Group. Reduction in the incidence of type 2 diabetes with lifestyle intervention or metformin. N Engl J Med 2002 Feb 7;346(6):393-403 [FREE Full text] [CrossRef] [Medline]
- 40. Bandura A. Social foundations of thought and action: a social cognitive theory. Englewood Cliffs, N.J: Prentice-Hall; 1986.
- 41. Richardson MT, Ainsworth BE, Jacobs DR, Leon AS. Validation of the Stanford 7-day recall to assess habitual physical activity. Ann Epidemiol 2001 Feb;11(2):145-153. [Medline]
- 42. Friedewald WT, Levy RI, Fredrickson DS. Estimation of the concentration of low-density lipoprotein cholesterol in plasma, without use of the preparative ultracentrifuge. Clin Chem 1972 Jun;18(6):499-502 [FREE Full text] [Medline]
- 43. Stunkard- A, McLaren-Hume M. The results of treatment for obesity: a review of the literature and report of a series. AMA Arch Intern Med 1959 Jan;103(1):79-85. [Medline]
- 44. Keller SD, Bayliss MS, Ware JE, Hsu MA, Damiano AM, Goss TF. Comparison of responses to SF-36 Health Survey questions with one-week and four-week recall periods. Health Serv Res 1997 Aug;32(3):367-384 [FREE Full text] [Medline]
- 45. Duckworth AL, Quinn PD. Development and validation of the short grit scale (grit-s). J Pers Assess 2009 Mar;91(2):166-174. [CrossRef] [Medline]
- 46. US Department of Health and Human Services: Healthy People 2020. Washington, DC Office of Disease Prevention and Health Promotion

URL:<u>http://www.healthypeople.gov/2020/topicsobjectives2020/objectiveslist.aspx?topic</u> Id=29[accessed [accessed 2014-10-10] [WebCite Cache]

- Durward CM, Hartman TJ, Nickols-Richardson SM. All-cause mortality risk of metabolically healthy obese individuals in NHANES III. J Obes 2012;2012:460321 [FREE Full text] [CrossRef] [Medline]
- St Jeor ST, Brunner RL, Harrington ME, Scott BJ, Daugherty SA, Cutter GR, et al. A classification system to evaluate weight maintainers, gainers, and losers. J Am Diet Assoc 1997 May;97(5):481-488. [CrossRef] [Medline]
- 49. Kayman S, Bruvold W, Stern JS. Maintenance and relapse after weight loss in women: behavioral aspects. Am J Clin Nutr 1990 Nov;52(5):800-807 [FREE Full text] [Medline]
- Jeffery RW. Does weight cycling present a health risk? Am J Clin Nutr 1996 Mar;63(3 Suppl):452S-455S [FREE Full text] [Medline]
- Vergnaud AC, Bertrais S, Oppert JM, Maillard-Teyssier L, Galan P, Hercberg S, et al. Weight fluctuations and risk for metabolic syndrome in an adult cohort. Int J Obes (Lond) 2008 Feb;32(2):315-321. [CrossRef] [Medline]
- 52. French SA, Jeffery RW, Folsom AR, McGovern P, Williamson DF. Weight loss maintenance in young adulthood: prevalence and correlations with health behavior and disease in a population-based sample of women aged 55-69 years. Int J Obes Relat Metab Disord 1996 Apr;20(4):303-310. [Medline]
- 53. Mason C, Foster-Schubert KE, Imayama I, Xiao L, Kong A, Campbell KL, et al. History of weight cycling does not impede future weight loss or metabolic improvements in postmenopausal women. Metabolism 2013 Jan;62(1):127-136 [FREE Full text] [CrossRef] [Medline]
- 54. Stevens J, Truesdale KP, McClain JE, Cai J. The definition of weight maintenance. Int J Obes (Lond) 2006 Mar;30(3):391-399. [CrossRef] [Medline]
- 55. Eysenbach G, Consort- E. CONSORT-EHEALTH: improving and standardizing evaluation reports of Web-based and mobile health interventions. J Med Internet Res 2011;13(4):e126 [FREE Full text] [CrossRef] [Medline]

Abbreviations

BF%: body fat percentage BMI: body mass index **CON:** control group **CSG:** counselor group **CV:** coefficients of variations FM: fat mass **GOAL:** Groningen Overweight and Lifestyle Grit-S: Short Grit Scale HDL-C: high-density lipoprotein cholesterol **IRB:** Institutional Review Board LDL-C: low-density lipoprotein cholesterol **MET:** metabolic equivalent NDSR: Nutrition Data System for Research **POP:** Pound of Prevention **RDG:** registered dietitian group **SCT:** social cognitive theory SF-36: Short-Form 36 Health Survey SNAP: Study of Novel Approaches to Weight Gain Prevention **TG:** triacylglycerides **UIUC:** University of Illinois at Urbana-Champaign



Figure 4.1. Diagram of recruitment, enrollment and randomization of participants in a study examining weight gain prevention in young adult and midlife women.

Characteristic	All participants mean (SD)		
Age (years)	31.4 (8.1)		
Height (cm)	165.2 (5.9)		
Weight (kg)	76.1 (19.0)		
Body mass index (kg/m²)	27.9 (6.8)		
Waist circumference (cm)	83.3 (13.6)		
Hip circumference(cm)	110.7 (14.6)		
Waist:hip ratio	0.8 (0.1)		
Body fat (%)	34.6 (9.1)		
Fat mass (kg)	27.9 (14.2)		
Fat free mass (kg)	48.2 (5.4)		
Systolic blood pressure (mm Hg)	106.2 (11.0)		
Diastolic blood pressure (mm Hg)	70.9 (9.9)		
Resting heart rate (bpm)	65.8 (6.0)		

Table 4.1. Baseline characteristics of women (n=81) participating in a 12-month weight gainprevention intervention and completing baseline testing.

Characteristic	All participants No. (%)
Education	
High school graduate	2 (3)
Some college	15 (18)
2-year associate degree/graduate	2 (3)
4-year college degree/graduate	21 (26)
Some graduate school	8 (10)
Master's degree	27 (33)
Doctorate degree	6 (7)
Race/ethnicity	
White, non-Hispanic	53 (66)
Black, non-Hispanic	10 (12)
Asian	8 (10)
Non-white Hispanic or Latino	4 (5)
Other (including multiracial)	6 (7)
Total annual household income	
<\$15,000	7 (9)
\$15,000 - \$49,999	30 (37)
\$50,000 – \$99,999	25 (31)
>\$100,000 - \$199,999	18 (22)
No response	1 (1)

Table 4.2. Demographic and education characteristics of all women (n=81) randomized to allgroups.

CHAPTER 5: Effect of nutrition education on weight gain prevention: a randomized controlled trial⁴

Abstract

Background: Body weight (BW) reduction through energy restriction is ineffective in impacting the obesity epidemic. Shifting from an obesity treatment to weight gain prevention or BW maintenance focus may be more effective in decreasing the burden of adult obesity. **Methods:** This was a 1-year randomized controlled trial of weight gain prevention in healthy premenopausal women, aged 18-45 y, with a body mass index (BMI) of >18.5 kg/m². Eighty-seven women were randomized to a weight gain prevention intervention delivered by a registered dietitian (RDG) or counselor (CSG), or to a control group (CON). Eighty-one women (mean±SD, age: 31.4±8.1 y; BW: 76.1±19.0 kg; BMI: 27.9±6.8 kg/m²) completed baseline testing and were included in intention-to-treat analyses; anthropometric, blood pressure, dietary intake and physical activity measurements and biochemical markers of health were measured every three months. Data were analyzed using repeated measures ANCOVA; significance was set at P<0.01.

Results: Sixty-two percent of these women met weight gain prevention criteria after one year (BW change ±3%); this did not differ by group assignment. Body fat percentage was significantly lower in the RDG versus CSG and CON groups at all intervals (P<0.001). Systolic blood pressure significantly changed over time in the CON group (P<0.001), with a significant group x time interaction (P<0.01). Fruit intake was significantly different between the RDG and CON groups at month 6 and 12 (both P<0.01). No significant group differences were observed for additional anthropometric measurements, resting heart rate, systolic and diastolic blood pressure, macronutrient intake, food groups, total energy expenditure or biochemical markers of health (all P>0.01). There were no significant time effects for any anthropometric measurements, resting heart rate, diastolic blood pressure, dietary intake, total energy expenditure or biochemical markers of health (all P>0.01).

⁴To be submitted to *Nutrition Journal*.

Conclusions: Although the effect of nutrition education on weight gain prevention was not significant, a majority of participants maintained BW over one year. Further monitoring may determine if nutrition education impacts BW regulation over the long term. Future interventions should continue to examine BW in addition to other clinical outcomes to assess overall health status.

Key words: body weight; obesity; overweight; weight gain prevention; weight loss; weight management; women

Background

More than two-thirds of adults and nearly two-thirds of women in the United States remain overweight and obese, even as the prevalence of overweight and obesity has stabilized in recent years [1]. Energy restriction and other methods to induce body weight (BW) reduction are popular and widely promoted [2,3]. While a variety of approaches result in short-term weight loss success [4-11], none of these methods have significantly impacted the obesity epidemic by permanently reducing BW over the long term for a significant amount of people [2,4, 12-16]. In fact, among individuals who have achieved weight loss, most return to initial weight status within three to five years following weight loss [13-15], and one-third to two-thirds of these individuals will regain more weight than was initially lost [4]. Even individuals who undergo bariatric surgery gradually regain weight over time [17,18]. Therefore, new prevention or treatment efforts and solutions to reduce the burden of adult obesity are necessary.

One such alternative may be shifting from a weight loss treatment approach to a weight gain prevention and health promotion approach. Weight gain prevention may also be referred to as weight maintenance and literally implies no change in BW. Unlike weight loss which is relevant only for individuals with excess BW, weight gain prevention is relevant for individuals with normal weight, overweight or obesity to manage current diseases and related risk factors [19], prevent the development of metabolic abnormalities and prevent the progression to overweight and/or obesity [20]. While other health indicators may be improved with weight loss, these benefits may be mitigated with weight regain. However, positive behavior changes can result in similar improvements in health indicators such as blood pressure [20-24] and blood lipid levels [20,21,24-26], even in the absence of weight change.

Research examining weight gain prevention is limited [27-35], and study populations have differed by gender (female only or males and females) and BW status (normal weight only, normal and overweight, overweight and obese only). As determinants of weight gain prevention may differ between males and females [3,36-38], the current study aimed to identify determinants of weight gain prevention in premenopausal women participating in a 1year randomized controlled trial of weight gain prevention that included nutrition education.

Women randomized to a nutrition education intervention group were hypothesized to maintain current BW, within ±3% [39], over the 1-year intervention period as compared to a control group. Further, it was hypothesized that women randomized to a nutrition education group led by registered dietitians would have lesser weight gain over the 1-year intervention period as compared to women randomized to an identical nutrition education group led by counselors with no formal nutrition training.

Methods

Participants

Premenopausal women with a body mass index (BMI) of >18.5 kg/m² and aged 18-45 years were recruited from the Urbana-Champaign communities and surrounding areas of Illinois. Full details regarding recruitment, screening and enrollment have been previously described along with complete inclusion and exclusion criteria [40]. Briefly, women were eligible to participate if they met age and BMI criteria and desired to prevent weight gain. Women were excluded if they were amenorrheic; presented with depressive symptomology as suggested by a score of >50 on the Zung Self-Rating Depression Scale/Status Inventory [41]; self-reported cardiovascular, metabolic or musculoskeletal abnormalities or used medications to manage such conditions; used supplements and/or medications that may influence BW regulation; had undergone weight loss surgery; or were currently pregnant, lactating or planning to become pregnant.

The Institutional Review Board (IRB) for the Protection of Human Subjects at the University of Illinois at Urbana-Champaign approved the study protocol (IRB#14397). Each participant provided written informed consent prior to participating in the study.

Study design and intervention

This was a 1-year parallel-arm randomized controlled trial of weight gain prevention that was conducted between August 2014-August 2015. After enrollment, participants were randomly assigned to a control group (CON), a weight gain prevention intervention delivered by a registered dietitian (RDG), or a weight gain prevention intervention delivered by a counselor (CSG). The weight gain prevention interventions delivered to the RDG and CSG were identical in

content and materials, but differed in the credentialing of the group leader. Between August 2014-August 2015, women randomized to the CON received no intervention.

Full details of the intervention have been described elsewhere [40]. Women in the RDG and CSG attended a total of 24 nutrition education sessions over the course of the 1-year intervention period. All sessions were 1-hour in length and emphasized portion control, planning ahead and vegetable consumption [42,43]. For the first 16 weeks of the intervention (months 1-4), participants attended weekly sessions; for the remaining 8 months of the study (months 5-12), participants attended monthly sessions [44]. Weekly sessions focused on general nutrition education, while monthly sessions addressed other areas of lifestyle behavior such as stress management, problem solving and motivation [40,42-47]. Six session times were offered each week/month per group. Participants were permitted to attend the session day and time that worked best for them during the respective week/month. Dates, times and building location were matched between RDG and CSG to ensure all participants had the same opportunities to attend sessions.

Education sessions were randomly selected for process evaluation by one process observer, using investigator-established criteria to assess fidelity. The number of participants attending each session was recorded, as was the start and end time. The fidelity checklist included educator-oriented items along with content-related items. The process observer rated the educator using a 'yes/no' rating system on items such as preparedness, familiarity, accuracy and ability to respond appropriately to questions. A comment box was also used to note general feedback on these items as deemed relevant by the process observer. Content-related items addressed whether underlying key concepts (portion control, vegetable consumption, planning ahead) and problem solving were covered. The process observer also recorded comments regarding challenges or difficulties of the education sessions. These evaluations assessed whether content for each session was covered and delivered appropriately and that all session activities were completed.

Four female registered dietitians with <5 years of experience delivered the intervention to RDG participants. Four female graduate teaching assistants from the University of Illinois at Urbana-Champaign in programs not related to nutrition and dietetics delivered the intervention

to CSG participants. Counselors were trained by one registered dietitian, including a code of conduct, before the start of the intervention. At least one week in advance of each session, all leaders were provided with the PowerPoint slides and script that included directions for session activities. Counselors also met with the registered dietitian to receive a brief overview of session handouts and activities and to have any questions answered.

To eliminate the potential for leader bias or participant attachment to a particular dietitian or counselor, education leaders rotated through the class dates and times to ensure equal interactions between the participants in each group and each of the four respective group leaders. Compliance with the intervention was defined as attending >85% of education sessions. For women who could not attend an education session, a virtual make-up session, accompanied by a short quiz, was offered. Women were deemed compliant if the quiz was accurately completed and returned to the primary investigator. Women were not informed of the credentials of session leaders until completion of the 1-year intervention.

Testing sessions

During the 1-year intervention, women participated in testing sessions (between 7:00 and 9:30 AM) before the intervention (baseline) and at month 3, month 6, month 9 and month 12. Anthropometric measurements, blood pressure, physical activity and biochemical markers of health were collected and evaluated at each measurement interval. Dietary intake was evaluated at baseline, month 6 and month 12.

Anthropometrics

A calibrated scale-mounted stadiometer (Seca 700, Hanover, MD, USA) was used to measure standing height (cm). Body weight (BW;kg), fat mass (FM;kg) and body fat percentage (BF%) were measured using a calibrated scale (Tanita 410GS, Arlington Heights, IL, USA). Height and BW measurements were used to calculate BMI (kg/m²) for each participant. Two measurements each of waist circumference (cm) and hip circumference (cm) were taken to the nearest 0.1 cm using a retractable measuring tape (Gulik II, Country Technology, Inc, Gay Mills, WI) and averaged. Waist circumference was measured at the narrowest point of the waist, approximately one inch above the navel, and hip circumference was measured at the widest part of the buttocks [43]. Waist and hip circumference measurements were used to calculate

waist-to-hip ratio. One research team member conducted all height and BW measurements, and a second research team member performed all waist and hip circumference measurements.

Blood pressure

A trained study investigator measured seated systolic and diastolic blood pressure (mm Hg) using a standard sphygmomanometer (Baumanometer®Desk Model, Copiague, NY, USA). Two blood pressure readings were taken with a 2-3 minute rest period between readings; the average systolic arterial pressure and average diastolic arterial pressure were recorded. Resting heart rate was also measured by pulse palpitation following a 5-minute rest period.

Dietary intake

Four-day food records were used to estimate dietary intake. In the week before baseline, month 6 and month 12 testing sessions, participants recorded all foods and beverages, including portion sizes, consumed for three weekdays and one weekend day [40]. The Nutrition Data System for Research (NDSR) dietary analysis software (Nutrition Coordinating Center, Minneapolis, MN, USA) was used to analyze food records to estimate average daily intake of total energy (kcal/day), carbohydrate (g/day), protein (g/day), fat (g/day), fiber (g/day) and food groups (svgs/day).

Physical activity

The Stanford 7-Day Physical Activity Recall Scale [48] was used to estimate physical activity. For seven consecutive days before all testing sessions, participants recorded the number of hours spent engaged in moderate, hard, and very hard activities, screen time (television, computer) and hours slept [48]. Minutes of physical activity per day were estimated by summing total minutes of moderate, hard, and very hard activity and dividing by seven. Calories expended per day were estimated by converting activities into metabolic equivalents (METs; hr/d) [40,43].

Biochemical markers of health

Venous blood samples (~30 mL) were collected from each participant following a 12-hour fast. Samples were processed, and serum was stored at -80°C until completion of bioassays for

glucose, insulin, low-density lipoprotein cholesterol (LDL-C), high-density lipoprotein cholesterol (HDL-C), total cholesterol and triglycerides (TG).

Spectrophotometry was used to measure serum glucose (mg/dL), total cholesterol (mg/dL), HDL-C (mg/dL) and TG (mg/dL) concentrations (all Stanbio Labs Boerne, TX, USA). Total cholesterol, HDL-C and TG concentrations were used to calculate LDL-C concentration (mg/dL) using the equation: LDL-C = total cholesterol - HDL-C - (TG/5) [49]. Enzyme-linked immunosorbent assay (ELISA) was used to measure serum insulin (μ U/mL; LINCO Research, St Charles, MO, USA). All serum samples were analyzed in duplicate for each study interval. Intra-assay coefficients of variations (CV) for serum glucose, total cholesterol, HDL-C and TG were 5.2, 5.4, 4.9 and 7.2%, respectively.

Statistical analyses

Participants completing baseline testing (n=81) were included in the intention-to-treat analyses, and the last observation carried forward approach was employed. A secondary efficacy analysis of only study completers also was conducted (n=48). The three groups (RDG, CSG and CON) differed at baseline by age, BW and BMI; hence, these variables were entered as covariates in all analyses examining effects of intervention. A 3x5 (3 treatment groups x 5 time intervals) analysis of covariance (ANCOVA) with repeated measures on the time factor was used to assess differences in outcome measures between and within treatment groups. If sphericity was violated, the Greenhouse-Geisser correction was used. The interaction of group (treatment) x interval (time) was assessed if a main effect of group or time was detected. Bonferroni adjustments for multiple comparisons were completed when significant group, time or group x time interactions were found.

Before conducting data analyses, outliers were identified for each outcome variable using the outlier labeling rule [50,51] and excluded from all analyses specific to that variable. Data are presented as unadjusted means ± standard deviation (SD) unless otherwise indicated. All data analyses were conducted using the Statistical Package for the Social Sciences (version 22.0, 2013, IBM Corp, Armonk, NY, USA). Statistical tests were two-sided, and significance was set at P<0.01.

Results

Figure 5.1 displays the flow diagram of enrollment and study completion of women in the current randomized controlled trial of weight gain prevention. Of the 330 women that responded to recruitment efforts between June-August 2014, 266 women met prescreening criteria and were sent additional screening materials and informed consent, which were returned by 146 women [40]. After review of materials by investigators, 102 women met eligibility criteria, and 97 of these women attended an informational session. Eighty-seven women were enrolled and randomized to the RDG (n=29), CSG (n=29) or CON (n=29) group. Baseline testing was completed by 81 healthy premenopausal women (White, non-Hispanic, n=53; Black, non-Hispanic, n=10; Asian, n=8; Non-white Hispanic or Latino, n=4; Other, including multiracial, n=6). Seventy-one women completed month 3 testing, 62 women completed month 6 testing, 60 women completed month 9 testing, and 48 women completed month 12 testing. The 81 women completing baseline testing (age: 31.4±8.1 y; BW: 76.1±19.0 kg; BMI: 27.9 ± 6.8 kg/m²) were included in intention-to-treat analyses. Women who completed the intervention (n=48; age: 33.4±7.2 y; BW: 79.7±19.7 kg; BMI: 29.5±7.2 kg/m²) were significantly older (P<0.01) as compared to women who did not complete the intervention (n=33; age: 28.4 \pm 8.6 y; BW: 70.7 \pm 17.0 kg; BMI: 25.6 \pm 5.6 kg/m²). When controlling for baseline age, BW and BMI, group assignment did not have an effect on participant dropout (P>0.01).

Weight gain prevention

Approximately 62.0% of the original sample (n=50) and 56.3% of completers (n=27) maintained BW (\pm 3%) over the intervention period. Using intention-to-treat analysis, 65.0% of the RDG (n=17), 65.5% of the CSG (n=19) and 53.8% of the CON (n=14) group met weight gain prevention criteria and were classified as successful. This distribution was similar among completers with 63.6% of the RDG (n=7), 56.3% of the CSG (n=9) and 52.4% of the CON (n=11) group meeting weight gain prevention criteria.

Education session compliance and process observation

Weekly, monthly and overall compliance did not differ by group assignment. Compliance was defined as attendance at >85% of education sessions (21/24 overall, 14/16 for weekly sessions, 7/8 for monthly sessions). Using this criteria, 88.6% (n=39) of women randomized to the RDG and CSG who were still enrolled after 16 weeks (n=44) were deemed

compliant with weekly attendance (RDG=20; CSG=19), and 81.5% (n=22) of the RDG and CSG who completed the intervention (n=27) were compliant with monthly attendance (RDG=10; CSG=12). Overall, 88.5% (n=23) of women who were randomized to the RDG and CSG and completed month 12 testing (n=27) were compliant (RDG=10; CSG=13).

Twenty-two RDG and 17 CSG education sessions were observed. On average, sessions lasted 39.1±11.1 min; however, RDG sessions were significantly longer than CSG sessions (43.3±9.2 vs. 33.5±11.2 min, P<0.01). On average, 2.9±1.6 participants attended each session, and this did not differ between groups. Based on the fidelity checklist, participants in both groups were equally engaged in education sessions, and vegetable consumption, portion control and planning ahead for food intake were addressed equally across all sessions. The process observer perceived that the registered dietitians were more likely to give specific scenarios and reinforce points made in various education sessions than counselors. It was also perceived that counselors tended to read from the script, while registered dietitians were more familiar with the topics and able to discuss without relying on supporting materials.

Anthropometric measurements

There were no significant group differences for BW, BMI, FM, waist circumference, hip circumference or waist-to-hip ratio (Table 5.1). BF% was significantly lower in the RDG compared to the CSG and CON groups at all intervals. Women in all groups were able to prevent gains in BW, BMI, FM, BF%, waist circumference, hip circumference and waist-to-hip ratio over time, and there was no time effect for these anthropometric measurements. Results were similar with efficacy analysis; however, the P-value for group differences in BF% increased to P<0.01 (P<0.001 for intention-to-treat analysis; data not shown).

Blood pressure

Group assignment had no significant effect on resting heart rate, systolic blood pressure or diastolic blood pressure (Table 5.2). Significant changes in systolic blood pressure were observed over time within the CON group between month 6 and 9 and month 6 and 12. A significant group x time interaction was detected for systolic blood pressure [F(7.1, 264.5) = 2.7, P<0.01] such that change within the CON group was significantly different from the RDG and CSG groups from month 6 to 9 and from month 6 to 12. The efficacy analysis revealed a

significant time difference (P<0.01) for the CON group between month 6 and month 9 (data not shown) which differed from the intention-to-treat analysis. Additionally, the group x time interaction for systolic blood pressure was no longer significant in the efficacy analysis.

Dietary intake

Dietary intake data are displayed in Table 5.3. Total energy intake, total and percent carbohydrate, total and percent protein, total and percent fat and total fiber did not differ significantly by group and did not significantly change over time. Servings of fruits, vegetables, whole and refined grains, lean meats and non-meat protein sources and low-fat and fat-free dairy did not change over time. Only fruit intake was significantly different between the RDG and CON at month 6 and 12. There was no group x time interaction for any food group servings, however. The efficacy analysis revealed no significant differences in macronutrient or food group serving intakes between groups or over time (data not shown).

Physical activity

Total energy expenditure did not significantly change over time and was not significantly different by group (Table 5.3). Results from the efficacy analysis were congruent with intention-to-treat findings (data not shown).

Biochemical markers of health

No significant group differences were observed for total cholesterol, HDL-C, LDL-C, TG, glucose or insulin concentrations and there were no significant changes for any of these biomarkers over time (Table 5.4). A group x time interaction was not found for any blood lipids, glucose or insulin concentrations. Similar results were found with the efficacy analysis (data not shown).

Discussion

This randomized controlled trial aimed to prevent weight gain in healthy premenopausal women over one year and to compare the effects of an intervention delivered by a registered dietitian to a counselor on weight gain prevention in the same sample of women. Although women randomized to a nutrition education intervention were able to maintain BW within $\pm 3\%$, women who did not receive a nutrition education intervention also were able to maintain BW within $\pm 3\%$. Despite that neither hypothesis was supported, approximately 62% of women

enrolled in the study were able to maintain BW over the 1-year intervention. Further, more women in the RDG group (absolute BW change: 1.4±2.9 kg; relative BW change: 0.79±1.6%) were successful in weight gain prevention as compared to CSG (absolute BW change: 0.50±3.2 kg; relative BW change: 0.41±2.3%) and CON groups (absolute BW change: 0.71±3.8 kg; relative BW change: 0.61±2.6%), though these differences were not significant. The average absolute and relative change in BW for the entire sample was 0.85±3.3 kg and 0.61±2.6%, respectively. The lack of differences between the three groups demonstrates that women can, in fact, successfully prevent weight gain over one year. However, a large proportion of the population still struggles with weight gain, and further evaluation is needed to have a significant impact on the current obesity epidemic in the United States.

Overall findings from this study related to changes in BW and weight gain prevention are consistent with those of Pound of Prevention [28,29], Levine and colleagues [30] and the Groningen Overweight and Lifestyle (GOAL) Study [31-33] which found interventions or treatments to have no significant effect on BW over time. In Pound of Prevention, a no-contact control group was compared to a group that received monthly nutrition education via newsletters and a group that received the same nutrition education plus lottery incentives for participation [28,29]. Over a 3-year period, weight gain did not differ significantly between groups [29], and mean BW change after one year in Pound of Prevention [28] was similar to that of the current study. Similarly, no effect on BW in normal weight and overweight women was observed by Levine and colleagues [30], who compared 15 group education sessions, 15 correspondence education lessons and an information-only control over three years. The GOAL Study randomized overweight and obese men and women with hypertension and/or dyslipidemia to receive usual care from a general practitioner or computer-guided lifestyle counseling from a nurse practitioner [31-33]. After one year, men randomized to the nurse practitioner group had a significantly greater weight change as compared to the usual care group [31], but these differences were not apparent after three years [33]. In women, there was no difference between the two groups at either one year [31,32] or three years [33].

When examining the prevalence of weight gain prevention or weight maintenance in the context of an entire population, it becomes somewhat problematic, as the definition of

weight maintenance has not been consistently used. Jeffery and French found that 37% of participants in Pound of Prevention maintained or lost weight, but investigators did not specify how "maintained" was defined [29]. A similar prevalence of weight gain prevention (40%) was found by Levine and colleagues [30], who defined maintenance as at or below ±2 pounds of baseline weight. The GOAL Study found a prevalence of 71.4% and 62.7% at one [31] and three years [33], respectively, using a definition of <1% BW gain [31-33]. The 62% of women that were classified as successful at weight gain prevention in the current study may be slightly underestimated as women were only classified as weight maintainers if percent BW change was $\pm 3\%$; therefore, women who lost >3% of BW were classified as non-weight maintainers.

Only two of the six previous studies that have examined weight gain prevention have been successful in preventing weight gain over time [27,34]; final results of the Study of Novel Approaches to Weight Gain Prevention (SNAP) are not yet available [35]. Age has been identified as a predictor of weight gain prevention [27,30] which may explain why a lowintensity nutrition education intervention via monthly newsletters was effective in producing a significantly greater BW change over one year when compared to a no-contact control in which the mean age of participants was 45.9 years [27]. The Shape Program [34] compared usual care to a primary care-based medium-intensity behavioral weight gain prevention intervention in premenopausal overweight and obese class 1 black women. Weight change was significantly larger in the intervention group compared to usual care at one year and 18 months; however, there were no differences between groups in any other outcome measures [34]. While findings of the Shape Program were significant, results are limited in generalizability due to race and socioeconomic status of the sample population.

The current randomized controlled trial of weight gain prevention builds upon the strengths and recommendations of previous weight gain prevention trials. The current trial was unique in that it included only women and did not exclude on the basis of BMI (except for underweight). Treatment had a greater effect on weight gain prevention after one year in men in A Pound of Prevention [27] and the GOAL Study [31-33] which suggests women may need different types or intensities of interventions. Both of these interventions [27,31-33] were relatively low in intensity, while the current trial was moderate in intensity. The nutrition

education component of the current intervention included group education classes, as previous research has demonstrated that group therapy results in greater weight loss when compared to individual therapy, even among individuals who prefer individual counseling [52]. Jeffery and French [29] previously recommended that more attention should be given to frequency of messages, interactive components and motivational concerns, all of which were addressed in the current randomized controlled trial of weight gain prevention in women.

With further regard to the nutrition education component of the intervention, this is the first trial to test a weight gain prevention nutrition education intervention delivered by a registered dietitian compared to an individual without formal nutrition training. The absence of differences in main outcomes between the RDG and CSG suggests that the two interventions were equally effective in promoting weight gain prevention in this sample of healthy premenopausal women. Similar findings related to the use of peer or lay educators have been observed in programs and interventions aimed at reducing chronic disease risk factors [53-57]. Although it was hypothesized that the RDG would have less weight gain over time due to the specialized training of registered dietitians, the counselors were trained by a registered dietitian and received all lesson slides and materials, including scripts. These findings are similar to those of Katula and colleagues in which community health workers trained and supported by registered dietitians delivered a 24-month lifestyle intervention that resulted in significant reductions in BW, BMI, waist circumference, glucose, insulin and insulin resistance in individuals with pre-diabetes [55]. When provided with adequate training by registered dietitians, lay educators may be able to promote outcomes equal to those of registered dietitians. These findings present an opportunity for registered dietitians to expand their sphere of influence by training and supporting lay educators to have an impact on health promotion by providing accurate and credible information. By training these individuals, registered dietitians can help reduce the amount of nutrition misinformation that these individuals may otherwise communicate with the public.

As a focus on weight and weight loss has not produced positive long-term results in reducing the obesity epidemic, a continued focus on weight as the primary indicator of health may cause more harm than good by increasing preoccupation with food, increasing the

likelihood of weight cycling and decreasing body image and self-esteem [2]. Therefore, collecting and examining other indicators of health, beyond BW and anthropometrics, is warranted. Weight is one component of metabolic health [58,59] and other indicators of metabolic health should be considered when assessing overall health status. Even in this sample of women who were overweight, on average, blood pressure, lipid levels and glucose were within normal ranges. Further, lifestyle and behavior changes can positively improve clinical health indicators, even in the absence of weight change [20-26]. Using only BW or BMI as proxies for health may misidentify healthy overweight and obese individuals as unhealthy and in need of treatment and unhealthy normal weight individuals as healthy and not in need of treatment [2,58,59]. Both of these situations may translate into increased healthcare costs through unnecessary treatment or worsening of conditions that were not identified due to the use of BW or BMI as the primary indicators of health. The current trial is one of the few studies to collect clinical health indicators beyond BW and anthropometrics in addition to questionnaires assessing health behaviors and psychosocial outcomes. Even though there were no significant findings with regards to other indicators of health, the current study provides an example for future interventions.

While the findings of the current study do not support the *a priori* hypotheses, there are several explanations for these findings. Weight changes over the course of one year were expected to be relatively small, so the absence of a stronger effect of the nutrition education weight gain prevention intervention is not entirely surprising. As a majority of women in the RDG and CSG successfully prevented weight gain over one year, this is promising that nutrition education may play a role in weight gain prevention over the long term. Further, similar to Wong and colleagues [60] baseline indicators of health were not clinically abnormal in this population of women. Therefore, lifestyle and behavior changes made during the intervention period, while positive, may not have been large enough to produce a substantial impact [60]. Over time, these sustained lifestyle and behavior changes may result in more clinically meaningful improvements.

The lack of statistically significant findings also may be explained by the Hawthorne effect or observation bias [61]. Even though women in the CON group did not receive any

intervention or information, their BW was assessed, along with a variety of other outcome measures at the same intervals as women receiving the nutrition education intervention. The Hawthorne effect may have resulted in CON participants modifying or altering behavior more so than under different circumstances, which may explain why there were no differences observed between the RDG and CSG compared to the CON. Additionally, women who enrolled in the study may have had more motivation to make lifestyle and behavior changes than individuals who were eligible and chose not to participate. Individuals who may benefit most from this type of intervention may not be adequately represented due to volunteer or selfselection bias [62].

Another explanation for non-significant findings may be the purposeful rotation of education leaders across education sessions. Providing exposure to multiple educators (i.e., four each) within the specified RDG or CSG group was done to specifically minimize potential effects of a particular educator on participant outcomes resulting from educator-participant bonding or external responsibility or support [63] rather than the intervention itself.

The inclusion of free-living women in this weight gain prevention intervention is a major strength of the current study, and findings from this study may be generalizable to similar populations of women with a desire to prevent weight gain. The nutrition education component of this intervention not only provided credible nutrition information, but also accountability and group support, all of which have been identified as facilitators to weight loss and weight loss maintenance [3]. Further, a majority of participants were compliant, indicating this was a feasible intervention with high participation. Although overall participant attrition was high at the conclusion of the intervention, nearly 75% of the original sample remained through month 9 of the study, indicating that the intervention was realistic for free-living individuals.

This study is not without limitations. As previously mentioned, the final retention rate was moderate which limits statistical power of the study; results should be interpreted with caution. However, a majority of individuals moved away from the area in the final month of the study and were unable to attend the final testing session. These "dropouts" were much different than the women who chose to withdraw from the study for other reasons. While the

use of free-living women was a strength, it was also a weakness as it is difficult to fully assess dietary intake and physical activity due to the limitations of self-report. Further, results from this study are only generalizable to women similar to those included in this study. More research is needed to target women of a wider range of ethnicities. Future research should also examine weight gain prevention in men of this same age range, as well as postmenopausal women.

Conclusions

In conclusion, while this randomized controlled trial of weight gain prevention did promote weight maintenance in a sample of healthy premenopausal women, there were no differences between women who received the weight gain prevention intervention compared to those randomized to a control group. Non-significant findings demonstrate that weight gain prevention over one year is possible; however, longer follow up periods are necessary. Clinically meaningful or significant benefits of participation in this study may become apparent over time, but long-term follow up data are not available. As weight loss and weight loss maintenance remain a challenge for much of the population, future interventions should focus on indicators of health in addition to BW and utilize a health promotion and weight gain prevention approach.

References

- 1. Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. JAMA. 2012;307:491-7.
- Bacon L, Aphramor L. Weight science: evaluating the evidence for a paradigm shift. Nutr J. 2011;10:9. doi: 10.1186/1475-2891-10-9.
- Metzgar CJ, Preston AG, Miller DL, Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. J Hum Nutr Diet. 2015;28:593-603.
- 4. Mann T, Tomiyama J, Westling E, Lew AM, Samuels B, Chatman J. Medicare's search for effective obesity treatments: diets are not the answer. Am Psychol. 2007;62:220-33.
- Hill JO, Thompson H, Wyatt H. Weight maintenance: what's missing? J Am Diet Assoc. 2005;105:S63-6.
- Dansinger ML, Gleason JA, Griffith JL, Selker HP, Schaefer EJ. Comparison of the Atkins, Ornish, Weight Watchers, and Zone diets for weight loss and heart disease risk reduction: a randomized trial. JAMA. 2005;293:43-53.
- Gardner CD, Kiazand A, Alhassan S, et al. Comparison of the Atkins, Zone, Ornish, and LEARN diets for change in weight and related risk factors among overweight premenopausal women: the A TO Z Weight Loss Study: a randomized trial. JAMA. 2007;297:969-77.
- Shai I, Schwarzfuchs D, Henkin Y, et al. Weight loss with a low-carbohydrate, Mediterranean, or low-fat diet. N Engl J Med. 2008;359:229-41.
- Makris A, Foster GD. Dietary approaches to the treatment of obesity. Psychiatr Clin North Am. 2011;34:813-27.
- 10. Sacks FM, Bray GA, Carey VJ, et al. Comparison of weight-loss diets with different compositions of fat, protein, and carbohydrates. N Engl J Med. 2009;360:859-73.
- 11. McGuire MT, Wing RR, Klem ML, Seagle HM, Hill JO. Long-term maintenance of weight loss: do people who lose weight through various weight loss methods use different behaviors to maintain their weight? Int J Obes Relat Metab Disord. 1998;22:572-7.

- Wing RR, Phelan S. Long-term weight loss maintenance. Am J Clin Nutr. 2005;82:2225-5S.
- 13. Goodrick GK, Poston WS, Foreyt JP. Methods for voluntary weight loss and control: update 1996. Nutrition. 1996;12:672-6.
- 14. Jeffery RW, Drewnowski A, Epstein LH, et al. Long-term maintenance of weight loss: current status. Health Psychol. 2000;19:5-16.
- 15. Byrne S, Cooper Z, Fairburn C. Weight maintenance and relapse in obesity: a qualitative study. Int J Obes Relat Metab Disord. 2003;27:955-62.
- 16. McGuire MT, Wing RR, Klem ML, Lang W, Hill JO. What predicts weight regain in a group of successful weight losers? J Consult Clin Psychol. 1999:67;177-85.
- 17. Sjöström L, Lindroos AK, Peltonen M, et al. Lifestyle, diabetes, and cardiovascular risk factors 10 years after bariatric surgery. N Engl J Med. 2004;351:268-93.
- 18. Christou NY, Look D, MacLean LD. Weight gain after short- and long-limb gastric bypass in patients followed for longer than 10 years. Ann Surg. 2006;244:734-40.
- 19. Peirson L, Fitzpatrick-Lewis D, Ali MU, et al. Prevention of overweight/obesity in adult populations: a systematic review with meta-analyses. Ottawa: Canadian Task Force on Preventive Health Care; 2014. <u>http://canadiantaskforce.ca/files/guidelines/2015-</u> <u>obesity-adults-treatment-systematic-review-en.pdf</u>
- 20. van Genugten L, van Empelen P, Oenema A. From weight management goals to action planning: identification of a logical sequence from goals to actions and underlying determinants. J Hum Nutr Diet. 2012;25:354-64.
- 21. Bacon L, Stern J, Van Loan M, Keim N. Size acceptance and intuitive eating improve health for obese, female chronic dieters. J Am Diet Assoc. 2005;105:929-36.
- 22. Fagard RH. Physical activity in the prevention and treatment of hypertension in the obese. Med Sci Sports Exerc. 1999;31:S624-30.
- 23. Appel LJ, Moore TJ, Obarzanek E, et al. A clinical trial of the effects of dietary patterns on blood pressure. N Engl J Med. 1997;33:1117-24.
- 24. Gaesser GA. Exercise for prevention and treatment of cardiovascular disease, type 2 diabetes, and metabolic syndrome. Curr Diab Rep. 2007;7:14-19.

- 25. Kraus WE, Houmard JA, Duscha BD, et al. Effects of the amount and intensity of exercise on plasma lipoproteins. N Engl J Med. 2002;347:1483-92.
- 26. Lamarche B, Despres JP, Pouliot MC, et al. Is body fat loss a determinant factor in the improvement of carbohydrate and lipid metabolism following aerobic exercise training in obese women? Metabolism. 1992;41:124-56.
- 27. Forster JL, Jeffery RW, Schmid TL, Kramer FM. Preventing weight gain in adults: a pound of prevention. Health Psychol. 1988;7:515-25.
- Jeffery RW, French SA. Preventing weight gain in adults: design, methods and one year results from the Pound of Prevention study. Int J Obes Relat Metab Disord. 1997;21:457-64.
- 29. Jeffery RW, French SA. Preventing weight gain in adults: the pound of prevention study. Am J Public Health. 1999;89:747-51.
- Levine MD, Klem ML, Kalarchian MA, et al. Weight gain prevention among women.
 Obesity (Silver Spring). 2007;15:1267-77.
- ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing weight gain: one-year results of a randomized lifestyle intervention. Am J Prev Med. 2009;37:270-7.
- 32. ter Bogt NC, Milder IE, Bemelmans WJ, et al. Changes in lifestyle habits after counselling by nurse practitioners: 1-year results of the Groningen Overweight and Lifestyle study. Public Health Nutr. 2011;14:995-1000.
- 33. ter Bogt NC, Bemelmans WJ, Beltman FW, Broer J, Smit AJ, van der Meer K. Preventing weight gain by lifestyle intervention in a general practice setting: three-year results of a randomized controlled trial. Arch Intern Med. 2011; 171:306-13.
- 34. Bennett GG, Foley P, Levine E, et al. Behavioral treatment for weight gain prevention among black women in primary care practice: a randomized clinical trial. JAMA Intern Med. 2013;173:1770-7.
- 35. Wing RR, Tate D, Espeland M, et al. Weight gain prevention in young adults: design of the study of novel approaches to weight gain prevention (SNAP) randomized controlled trial. BMC Public Health. 2013:13;300. doi: 10.1186/1471-2458-13-300.

- 36. Colvin RH, Olson SB. A descriptive analysis of men and women who have lost significant weight and are highly successful at maintaining the loss. Addict Behav. 1983;8:287-95.
- 37. Jeffery RW, Bjornson-Benson WM, Rosenthal BS, Lindquist RA, Kurth CL, Johnson SL. Correlates of weight loss and maintenance over two years of follow-up among middleaged men. Prev Med. 1984;13:155-68.
- 38. Tiggerman M, Rothblum ED. Gender differences in internal beliefs about weight and negative attitudes towards self and others. Pyschol Women Q. 1997;21:581-93.
- 39. Stevens J, Truesdale KP, McClain JE, Cai J. The definition of weight maintenance. Int J Obes (Lond). 2006;30:391-9.
- 40. Metzgar CJ, Nickols-Richardson SM. Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial. JMIR Res Protoc. 2015;4:e36. doi:10.2196/resprot.4008.
- 41. Zung, W. W. A self-rating depression scale. Archives of General Psychiatry. 1965;12:63-70.
- 42. Piehowski KE, Preston AG, Miller DL, Nickols-Richardson SM. A reduced-calorie dietary pattern including a daily sweet snack promotes body weight reduction and body composition improvements in premenopausal women who are overweight and obese: a pilot study. J Am Diet Assoc. 2011;111:1198-1203.
- 43. Nickols-Richardson SM, Piehowski KE, Metzgar CJ, Miller DL, Preston AG. Changes in body weight, blood pressure and selected metabolic biomarkers with an energyrestricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr Res Pract. 2014;8:695-704.
- 44. Knowler WC, Barrett-Connor E, Fowler SE, et al. Reduction in the incidence of type 2 diabetes with lifestyle intervention or metformin. N Engl J Med. 2002;346:393-403.
- 45. Lutes LD, Winett RA, Barger SD, et al. Small changes in nutrition and physical activity promote weight loss and maintenance: 3-month evidence from the ASPIRE randomized trial. Ann Behav Med. 2008;35:351-7.

- 46. Slawson DL, Fitzgerald N, Morgan KT. Position of the Academy of Nutrition and Dietetics: the role of nutrition in health promotion and chronic disease prevention. J Acad Nutr Diet. 2013;113:972-9.
- 47. Perri MG, Limacher MC, Durning PE, et al. Extended-care programs for weight management in rural communities: the treatment of obesity in underserved rural settings (TOURS) randomized trial. Arch Intern Med. 2008;168:2347-54.
- 48. Richardson MT, Ainsworth BE, Jacobs DR, Leon AS. Validation of the Stanford 7-day recall to assess habitual physical activity. Ann Epidemiol. 2001;11:145-53.
- 49. Friedewald WT, Levy RI, Fredrickson DS. Estimation of the concentration of low-density lipoprotein cholesterol in plasma, without use of the preparative ultracentrifuge. Clin Chem. 1972;18:499-502.
- 50. Hoaglin DC, Iglewicz B, Tukey JW. Performance of some resistant rules for outlier labeling. J Am Stat Assoc. 1986;81:991-9.
- 51. Hoaglin DC, Iglewicz B. Fine tuning some resistant rules for outlier labeling. J Am Stat Assoc. 1987;82:1147-9.
- 52. Renjilian DA, Perri MG, Nezu AM, McKelvey WF, Shermer RL, Anton SD. Individual versus group therapy for obesity: effects of matching participants to their treatment preferences. J Consult Clin Psychol. 2001;69:717-21.
- 53. Dutton GR, Phillips JM, Kukkamalla M, Cherrington AL, Safford MM. Pilot study evaluating the feasibility and initial outcomes of a primary care weight loss intervention with peer coaches. Diabetes Educ. 2015;41:361-8.
- Leahey TM, Wing RR. A randomized controlled pilot study testing three types of health coaches for obesity treatment: professional, peer, and mentor. Obesity (Silver Spring). 2013;21:928-34.
- 55. Katula JA, Vitolins MZ, Morgan TM, et al. The Healthy Living Partnerships to Prevent Diabetes study: 2-year outcomes of a randomized controlled trial. Am J Prev Med. 2013;44:S324-32.

- 56. West DS, Bursac Z, Cornell CE, et al. Lay health educators translate a weight-loss intervention in senior centers: a randomized controlled trial. Am J Prev Med. 2011;41:385-91.
- 57. Parikh P, Simon EP, Fei K, et al. Results of a pilot diabetes prevention intervention in East Harlem, New York City: Project HEED. Am J Public Health. 2010;100:S232-9.
- Durward CM, Hartman TJ, Nickols-Richardson SM. All-cause mortality risk of metabolically healthy obese individuals in NHANES III. J Obes. 2012;2012:460321. doi: 10.1155/2012/460321.
- 59. Wildman RP, Muntner P, Reynolds K, et al. The obese without cardiometabolic risk factor clustering and the normal weight with cardiometabolic risk factor clustering: prevalence and correlates of 2 phenotypes among the US population (NHANES 1999-2004). Arch Intern Med. 2008;168:1617-24.
- 60. Wong MC, Wang HH, Kwan MW, et al. Dietary counselling has no effect on cardiovascular risk factors among Chinese grade 1 hypertensive patients: a randomized controlled trial. Eur Heart J. 2015;36:2598-607.
- McCambridge J, Witton J, Elbourne DR. Systematic review of the Hawthorne effect: new concepts are needed to study research participant effects. J Clin Epidemiol. 2014;67:267-77.
- 62. Hedge MN. A methodological review of randomized controlled trials. CDR. 2007;1:15-36.
- 63. Melnyk B, Morrison-Beedy D. Intervention research: design, conducting, analyzing and funding. New York: Springer Publishing Company, LLC; 2012.



Figure 5.1. Flow diagram of participant enrollment and completion of a 1-year randomized controlled trial of weight gain prevention in which women were randomized to a registered dietitian-led nutrition education group (RDG), a counselor-led nutrition education group (CSG) or a control group (CON).

RDG=registered dietitian group; CSG=counselor group; CON=control group.

^aReasons for withdrawal in RDG: time (n=2 before month 3 testing), family emergency (n=1 before baseline testing, n=1 before month 3 testing, n=2 before month 12 testing), moved out of the area (n=6 before month 12 testing), pregnancy (n=1 before month 9 testing, n=1 before month 12 testing), lost to follow up (n=2 before baseline testing, n=1 before month 6 testing, n=1 before month 12 testing); 1 participant in the RDG did not attend the month 6 testing session due to a family emergency, but completed all remaining testing sessions; 1 participant in the RDG did not attend the month 9 testing, lost to follow up (n=1 before month 12 testing), moved out of the area (n=1 before month 9 testing, n=1 before month 3 testing), bot to follow up (n=1 before month 12 testing), moved out of the area (n=1 before month 9 testing, n=2 before month 12 testing), time (n=4 before month 3 testing), undisclosed (n=1 before month 3 testing), personal health issues (n=1 before month 9 testing), Zung >50 (n=1 before month 12 testing).

^cReasons for withdrawal in CON: time (n=3 before baseline testing, n=1 before month 3 testing), Zung >50 (n=1 before month 6 testing), moved out of the area (n=1 before month 12 testing), family issues (n=1 before month 12 testing), pregnancy (n=1 before month 6 testing); 3 participants in the CON did not attend month 9 testing (illness n=2, time n=1), but returned for month 12 testing.

Table 5.1. Anthropometric measurements of premenopausal women in a 1-year randomized controlled trial of weight gain prevention in which women were randomized to a registered dietitian-led nutrition education group (RDG), a counselor-led nutrition education group (CSG) or a control group (CON).

		Baseline	Month 3	Month 6	Month 9	Month 12	
Variable	Group	Mean ± standard deviation					P-value
Body weight (kg)	RDG (n=26)	59.8±6.5	60.2±6.5	60.5±6.8	60.8±6.6	60.6±6.6	Group = NS
	CSG (n=29)	71.1±7.6	71.3±7.6	71.8±7.9	71.6±8.1	71.8±8.7	Time = NS
	CON (n=25	96.1±14.0	95.3±14.3	95.3±14.9	95.7±14.8	96.1±15.1	G x T = NS
Body mass index (kg/m ²)	RDG (n=26)	21.7±1.5	21.8±1.5	21.9±1.6	22.1±1.5	22.0±1.5	Group = NS
	CSG (n=29)	26.3±1.9	26.4±2.1	26.6±2.3	26.5±2.3	26.6±2.6	Time = NS
	CON (n=25)	35.2±5.4	35.0±5.6	34.9±5.7	35.0±5.6	35.2±5.8	G x T = NS
Fat mass (kg)	RDG (n=26)	15.4±4.5	16.0±4.5	16.5±4.7	16.5±4.7	16.3±4.5	Group = NS
	CSG (n=29)	24.2±5.3	25.0±5.6	25.3±6.0	25.0±6.1	25.3±6.6	Time = NS
	CON (n=25)	43.3±9.8	43.6±10.6	43.6±10.6	43.8±10.8	43.9±10.7	G x T = NS
Body fat percentage (%)	RDG (n=26)	25.3±5.4	26.1±5.2	26.1±6.2	26.7±5.3	26.6±4.9	Group <0.001
	CSG (n=29)	33.7±4.3†	34.6±4.2†	34.8±4.8†	34.5±5.0†	34.7±5.1†	Time = NS
	CON (n=25)	44.6±3.9†	45.2±4.2†	45.2±4.8†	45.1±5.0†	45.3±4.2†	G x T = NS
Waist circumference (cm)	RDG (n=26)	70.8±4.5	70.7±4.7	70.8±4.4	70.6±4.5	70.8±4.4	Group = NS
	CSG (n=29)	80.3±5.0	80.1±5.0	80.5±5.0	80.2±5.4	80.5±5.6	Time = NS
	CON (n=25)	98.1±10.2	96.9±10.2	96.0±10.2	96.1±10.4	97.5±10.7	G x T = NS
Hip circumference (cm)	RDG (n=26)	97.4±4.7	99.4±3.8	100.0±4.3	99.0±3.6	98.7±4.2	Group = NS
	CSG (n=29)	108.0±6.5	108.2±5.5	108.0±5.7	107.3±5.8	107.4±6.0	Time = NS
	CON (n=24)	124.3±10.2	124.3±10.2	123.9±9.4	123.0±9.9	123.8±10.6	G x T = NS
Waist-to-hip ratio	RDG (n=26)	0.73±0.04	0.71±0.04	0.71±0.04	0.71±0.04	0.72±0.04	Group = NS
	CSG (n=29)	0.75±0.05	0.74±0.05	0.75±0.05	0.75±0.05	0.75±0.05	Time = NS
	CON (n=25)	0.78±0.07	0.78±0.06	0.78±0.06	0.78±0.07	0.78±0.06	G x T = NS

P-values using analysis of covariance, adjusted for baseline age, body weight and body mass index, with repeated measures on the time factor, using P<0.01 for statistical significance; data presented as unadjusted means ± standard deviation.

†different from RDG.

G x T = group x time interaction.

Table 5.2. Resting heart rate and blood pressure measurements of premenopausal women in a 1-year randomized controlled trial of weight gain prevention in which women were randomized to a registered dietitian-led nutrition education group (RDG), a counselor-led nutrition education group (CSG) or a control group (CON).

		Baseline	Month 3	Month 6	Month 9	Month 12	
Variable	Group	Mean ± standard deviation P-value				P-value	
Resting heart rate (bpm)	RDG (n=23)	64.0±2.8	68.3±9.5	65.7±7.6	63.1±8.3	61.8±6.1	Group = NS
	CSG (n=22)	65.1±2.1	64.9±8.4	62.5±6.3	62.9±6.2	61.4±4.1	Time = NS
	CON (n=20)	65.4±2.0	64.2±9.9	66.9±4.9	67.5±6.7	62.1±5.9	G x T = NS
Resting systolic blood pressure (mm Hg)	RDG (n=26)	100.4±7.0	99.6±7.7	100.5±6.8	99.3±5.9	100.0±5.7	Group = NS
	CSG (n=29)	107.8±9.8	106.3±10.3	107.5±9.9	105.8±11.0	105.2±10.5	Time <0.001
	CON (n=25)	110.5±13.4	110.8±10.6	111.8±8.7	108.6±9.1 ^a	108.4±9.1 ^a	G x T <0.01
Resting diastolic blood pressure (mm Hg)	RDG (n=26)	65.2±10.0	64.2±6.4	60.5±9.1	62.2±6.2	61.9±6.6	Group = NS
	CSG (n=29)	72.6±7.5	69.5±9.3	67.7±8.4	67.3±7.9	67.1±8.5	Time = NS
	CON (n=25)	74.9±10.2	70.8±8.7	67.2±8.4	67.4±8.3	68.7±8.1	G x T = NS

P-values using analysis of covariance, adjusted for baseline age, body weight and body mass index, with repeated measures on the time factor, using P<0.01 for statistical significance; data presented as unadjusted means ± standard deviation.

^adifferent from Month 6.

G x T = group x time interaction.
Table 5.3. Macronutrient and food group intakes and energy expenditure of premenopausal women in a 1-year randomized
controlled trial of weight gain prevention in which women were randomized to a registered dietitian-led nutrition education group
(RDG), a counselor-led nutrition education group (CSG) or a control group (CON).

		Baseline	Month 3	Month 6	Month 9	Month 12	
Variable	Group	4	→ Mean ± standard deviation → P				P-value
	RDG (n=26)	1798.2±729.7		1682.7±595.5		1607.8±533.9	Group = NS
Energy intake (kcal/d)	CSG (n=29)	1971.7±604.7		1802.2±646.2		1837.6±656.5	Time = NS
	CON (n=26)	1795.9±707.8		1619.2±640.1		1709.7±719.0	G x T = NS
Total carbohydrato	RDG (n=26)	219.6±106.0		202.9±81.4		186.7±68.6	Group = NS
	CSG (n=29)	214.2±82.2		216.4±89.8		218.4±93.9	Time = NS
(g/u)	CON (n=26)	204.0±84.0		179.4±75.2		193.4±87.0	G x T = NS
	RDG (n=26)	72.1±32.2		72.2±30.7		69.1±28.4	Group = NS
Total protein (g/d)	CSG (n=29)	69.7±27.3		72.6±27.6		74.7±28.8	Time = NS
	CON (n=26)	74.0±38.6		70.9±33.4		72.1±34.2	G x T = NS
	RDG (n=26)	71.5±36.9		66.4±33.1		66.1±32.0	Group = NS
Total fat (g/d)	CSG (n=29)	73.1±34.7		71.8±32.7		71.8±30.5	Time = NS
	CON (n=26)	76.1±38.9		69.9±34.9		72.6±36.8	G x T = NS
	RDG (n=26)	19.0±11.8		19.6±9.2		16.9±7.1	Group = NS
Total fiber (g/d)	CSG (n=29)	19.3±11.3		19.9±11.3		18.4±9.8	Time = NS
	CON (n=26)	17.4±8.4		18.3±9.3		17.8±9.5	G x T = NS
Percentage	RDG (n=26)	48.1±11.3		47.5±11.1		46.0±11.0	Group = NS
carbohydrate	CSG (n=29)	47.4±11.2		47.0±10.0		46.3±9.3	Time = NS
(%kcal/d)	CON (n=26)	45.5±11.5		44.1±10.7		44.5±10.5	G x T = NS
Percentage protein (%kcal/d)	RDG (n=26)	16.5±6.4		17.6±6.6		17.7±7.0	Group = NS
	CSG (n=29)	15.8±5.2		16.6±5.1		16.7±4.8	Time = NS
	CON (n=26)	16.6±5.6		17.7±6.0		17.3±5.7	G x T = NS
Percentage fat (%kcal/d)	RDG (n=26)	33.9±9.0		33.8±8.7		35.2±8.5	Group = NS
	CSG (n=29)	35.1±8.9		34.9±8.4		34.6±8.6	Time = NS
	CON (n=26)	36.5±9.2		36.8±9.0		36.9±8.7	G x T = NS

Table 5.3 (cont.)							
Fruit (svgs/d)	RDG (n=26)	1.9±2.3		2.1±2.3		2.0±2.7	Group <0.01
	CSG (n=29)	1.4±1.8		1.4±1.8		1.4±2.2	Time = NS
	CON (n=26)	1.6±2.2		1.1±1.4†		1.3±1.5†	G x T = NS
	RDG (n=26)	3.4±2.7		3.3±2.5		3.4±2.8	Group = NS
Vegetables (svgs/d)	CSG (n=29)	3.3±2.7		3.3±2.4		3.1±2.3	Time = NS
	CON (n=26)	3.3±2.3		3.3±2.4		3.4±2.4	G x T = NS
	RDG (n=26)	1.7±2.8		1.6±1.4		1.4 ± 1.4	Group = NS
Whole grains (svgs/d)	CSG (n=29)	1.8±2.2		1.7±1.9		1.6±1.9	Time = NS
	CON (n=26)	1.8±1.8		1.7±1.9		1.8±2.0	G x T = NS
	RDG (n=26)	3.9±2.9		3.6±3.2		3.4±3.1	Group = NS
Refined grains (svgs/d)	CSG (n=29)	4.4±3.1		4.3±2.9		4.6±3.2	Time = NS
	CON (n=26)	4.1±3.6		3.3±3.0		3.8±3.3	G x T = NS
	RDG (n=26)	1.7±2.7		2.1±2.6		2.1±2.8	Group = NS
Lean meats (svgs/d)	CSG (n=29)	1.4±2.3		1.7±2.5		2.0±2.7	Time = NS
	CON (n=26)	1.8±2.7		1.7±2.3		1.9±2.6	G x T = NS
Non most protoin	RDG (n=26)	1.7±2.7		2.3±3.3		1.8±2.3	Group = NS
sources (sygs/d)	CSG (n=29)	1.5±2.7		1.4±2.3		1.1±1.6	Time = NS
sources (svgs/u)	CON (n=26)	1.3±1.9		1.5±2.0		1.3±1.9	G x T = NS
Low-fat and fat-free dairy (svg/d)	RDG (n=26)	0.54±0.68		0.80±0.97		0.71±0.85	Group = NS
	CSG (n=29)	0.79±0.95		0.83±1.1		0.86±0.94	Time = NS
	CON (n=26)	0.68±0.87		0.65±0.92		0.64±0.94	G x T = NS
Energy expenditure (kcal/d)	RDG (n=26)	1062.5±462.2	1002.2±411.4	1001.8±412.8	966.6±405.6	1009.2±389.2	Group = NS
	CSG (n=27)	1132.0±430.6	1003.1±341.3	978.2±330.4	1049.4±352.3	1069.1±354.8	Time = NS
	CON (n=21)	1322.2±489.6	1386.7±515.2	1330.1±561.7	1267.2±336.5	1302.9±371.4	G x T = NS

P-values using analysis of covariance, adjusted for baseline age, body weight and body mass index, with repeated measures on the time factor, using P<0.01 for statistical significance; data presented as unadjusted means ± standard deviation.

†different from RDG.

G x T = group x time interaction.

Table 5.4. Blood lipid, glucose and insulin concentrations in premenopausal women in a 1-year randomized controlled trial of weight gain prevention in which women were randomized to a registered dietitian-led nutrition education group (RDG), a counselor-led nutrition education group (CSG) or a control group (CON).

		Baseline	Month 3	Month 6	Month 9	Month 12	
Variable	Group	Mean ± standard deviation					P-value
Total cholostoral	RDG (n=26)	165.6±28.8	172.1±28.9	173.5±28.7	177.1±26.3	169.9±23.7	Group = NS
(mg/dl)	CSG (n=28)	163.7±32.9	176.8±36.1	174.5±32.7	178.2±35.3	176.0±32.3	Time = NS
(IIIg/uL)	CON (n=24)	159.2±30.4	171.9±32.9	175.7±30.1	178.5±30.2	167.1±32.7	G x T = NS
High-density	RDG (n=26)	55.6±15.8	55.2±13.6	52.8±13.5	54.4±14.6	53.1±15.3	Group = NS
lipoprotein cholesterol	CSG (n=29)	52.7±12.1	52.3±12.8	52.1±11.9	52.3±12.5	51.3±12.2	Time = NS
(mg/dL)	CON (n=24)	44.3±10.0	44.0±10.7	45.7±9.5	47.0±10.2	45.9±11.1	G x T = NS
Low-density	RDG (n=26)	97.3±24.8	105.4±22.1	109.2±23.7	111.2±26.0	104.3±20.9	Group = NS
lipoprotein cholesterol	CSG (n=28)	98.1±26.8	111.6±30.1	108.7±27.4	112.6±30.5	111.9±27.6	Time = NS
(mg/dL)	CON (n=23)	93.6±21.5	109.5±21.8	112.9±21.8	111.3±22.8	100.7±22.0	G x T = NS
	RDG (n=25)	58.6±28.1	53.6±26.9	56.0±25.1	55.7±24.5	57.1±22.3	Group = NS
Triglycerides (mg/dL)	CSG (n=28)	66.9±29.0	63.8±25.3	65.3±27.4	68.4±32.2	67.2±34.2	Time = NS
	CON (n=22)	78.3±33.0	65.9±25.2	62.1±21.6	78.8±29.0	78.9±24.9	G x T = NS
	RDG (n=23)	84.2±11.8	79.2±22.9	91.6±6.8	91.8±8.9	89.6±9.4	Group = NS
Glucose (mg/dL)	CSG (n=25)	87.6±19.2	86.0±18.3	92.6±8.1	92.1±7.7	91.4±8.5	Time = NS
	CON (n=23)	88.2±11.9	85.5±21.9	92.0±7.5	93.6±12.9	91.7±8.4	G x T = NS
Insulin (μU/mL)	RDG (n=25)	3.4±1.6	4.6±3.4	4.3±3.0	3.8±2.0	3.6±1.7	Group = NS
	CSG (n=28)	4.8±2.2	5.6±2.1	5.0±1.9	5.1±2.1	5.0±2.4	Time = NS
	CON (n=21)	6.9±2.9	7.4±3.4	6.6±2.3	7.1±2.8	7.3±2.9	G x T = NS

Fasting serum used to measure all biomarker concentrations.

P-values using analysis of covariance, adjusted for baseline age, body weight and body mass index, with repeated measures on the time factor, using P<0.01 for statistical significance; data presented as unadjusted means ± standard deviation.

G x T = group x time interaction.

CHAPTER 6: Grit: association with eating behaviors, body weight and body mass index⁵

ABSTRACT

OBJECTIVE: The aim of this cross-sectional study was to explore the association between grit and cognitive eating restraint (CER), disinhibition and hunger and body mass.

METHODS: Seventy-nine women [mean \pm SD, age: 31.5 \pm 8.2 y, body weight (BW): 76.3 \pm 19.2 kg, body mass index (BMI): 28.0 \pm 6.9 kg/m²] completed the Short Grit Scale (Grit-S) and Eating Inventory (EI) upon enrolling in a 1-year weight gain prevention trial.

RESULTS: The average Grit-S score was 3.4 ± 0.7 . Average EI scores for CER (21 items), disinhibition (16 items) and hunger (14 items) were 9.1 ± 3.5 , 7.6 ± 3.2 and 5.6 ± 2.8 , respectively. Grit-S score was negatively associated with disinhibition score (r=-0.47, P<0.01), hunger score (r=-0.19, P=0.05), BW (r=-0.24, P<0.05) and BMI (r=-0.23, P<0.05) and positively related to CER score (r=0.23, P<0.05). Disinhibition was the only predictor of BW (F=8.02, P<0.01) and BMI (F=7.05, P=0.01).

CONCLUSIONS: Women with more grit may have better weight management control as reflected by lower disinhibition and hunger scores and higher CER scores, as well as lower BW and BMI. Disinhibition, a cognitive disposition, may inversely reflect grit, a non-cognitive personality trait, to benefit BW and BMI; however, this requires further evaluation. Additional research is warranted to more fully understand grit in relation to other eating behavior traits in BW regulation.

KEYWORDS: Eating behaviors; Grit; Obesity; Weight management; Women

⁵To be submitted to *Obesity Facts*.

INTRODUCTION

Influenced by a multitude of factors at the individual, interpersonal, community, institutional and environmental levels, weight management is the result of complex interactions [1] and remains a challenge for many individuals. A more complete understanding of individual factors, including certain personality traits, may help nutrition professionals more fully support individuals in achieving weight management goals.

Grit – the perseverance and passion for sustaining efforts to pursue goals over time – may be one non-cognitive personality trait worth exploring in the field of weight management. While grit has not yet been examined in the context of weight management, previous studies show that grit is an indicator of success in life commitments, including scholarly pursuits [2-4], work efforts [4], military achievement [2,4], and marriage [4]. Among medical surgery residents, grit is a predictor of sound psychological health [5], and low grit may be a risk factor for withdrawal from surgical residency [6]. Further, grit is positively associated with high- and moderate-intensity exercise participation, as evaluated by stages of change in university students, faculty and staff [7].

Grit may be related to body weight (BW) regulation via cognitive control of food intake as cognitive eating restraint (CER), disinhibition and hunger are three eating behaviors associated with BW [8-12]. Cognitive eating restraint is defined as the tendency of some individuals to consciously limit food intake to manage body weight [8], disinhibition as the tendency of some individuals to overconsume food(s) in times of emotional distress [9], and hunger as the intensity and susceptibility to signs and symptoms related to food intake [9]. Though independent effects of CER, disinhibition and hunger on anthropometric measures are unclear, the interaction among these eating behaviors may be a useful predictor of BW status. Cognitive control of eating is essential to management and regulation of BW [8-14].

The current study explored the relationship between grit and CER, disinhibition and hunger (eating behaviors) and BW and body mass index (BMI) in a sample of adult women upon enrollment in a 1-year weight gain prevention trial. Based on grit as an indicator of perseverance, it was hypothesized that grit would be negatively associated with disinhibition

and hunger, BW and BMI and positively related to CER. It was also hypothesized that grit would predict body mass in this cross-sectional analysis.

METHODS

Participants

Premenopausal women, aged 18-45 y with BMI of >18.5 kg/m², were recruited from the Urbana-Champaign (IL, USA) communities. Full recruitment, screening and enrollment details have been described elsewhere [15]. Briefly, screening criteria included appropriate age, BMI >18.5 kg/m² and desire to prevent weight gain. Inclusion criteria included: eumenorrhea (≥8 menstrual cycles/y); absence of depressive symptomology as indicated by a score of <50 on Zung Self-Rating Depression Scale/Status Inventory [16]; absence of metabolic, cardiovascular or musculoskeletal abnormalities and no use of medications to manage these conditions or medications or supplements influencing BW regulation; non-smoker; non-pregnant, non-lactating and no plans to become pregnant; and no history of weight-loss surgery.

The Institutional Review Board (IRB) for the Protection of Human Subjects at the University of Illinois at Urbana-Champaign approved the study protocol (IRB#14397). Each participant provided written informed consent.

Measurements

Before the intervention (baseline), women completed self-administered paper-based questionnaires. Trained investigators collected anthropometric measurements from participants.

Grit

The Short Grit Scale (Grit-S), an 8-item questionnaire, was used to measure grit [3]. Using a 5-point (1=strongly disagree to 5=strongly agree) scale per item, women rated statements such as "I often set a goal but later choose to pursue a different one" [3] and "I finish whatever I begin" [3]. Items were averaged to produce an overall Grit-S score for each woman; possible scores ranged from 1-5 with higher scores indicating greater grit. Previous studies have found good internal reliability ranging from 0.73 to 0.84 [3,4,17,18].

Eating behaviors

Ratings of CER, disinhibition and hunger were measured using the Eating Inventory (EI) [8], a 51-item questionnaire. Subscale scores for each construct were calculated using standard scoring methods [8], with possible scores ranging from 0-21, 0-16 and 0-14 for CER, disinhibition and hunger, respectively. Higher scores within each subscale reflect higher levels of the respective construct.

Anthropometric measurements

Body weight (to the nearest 0.1 kg) and standing height (to the nearest 0.1 cm) were measured using a calibrated scale (Tanita 410GS, Arlington Heights, IL, USA) and stadiometer (Seca 700, Hanover, MD, USA), respectively. Investigators used these measurements to calculate BMI (kg/m²).

Statistical Analyses

Descriptive statistics (means±SD) were used to describe participant characteristics. Pearson's correlation coefficients were used to examine the relationship between Grit-S score and BW, BMI and CER, disinhibition and hunger scores. Backward elimination regression was used to further evaluate predictor variables of BW and BMI. The Statistical Package for the Social Sciences (version 22.0, 2013, IBM Corp, Armonk, NY, USA) was used to conduct all statistical analyses. Significance was set at P<0.05.

RESULTS

Seventy-nine of 81 participants (97.5%) provided complete questionnaires and were included in analyses. Women (age: $31.5\pm8.2 \text{ y}$) self-identified as primarily white, non-Hispanic (n=53, 67%), with at least a college degree (n=60, 76%) and middle to upper-middle income (n=43, 54%). Participants were overweight, on average (height: $165.1\pm5.9 \text{ cm}$, weight: $76.3\pm19.2 \text{ kg}$, BMI: $28.0\pm6.9 \text{ kg/m}^2$).

The average Grit-S score was 3.4 ± 0.7 (range 2.0-4.6). Average scores (ranges) for CER, disinhibition and hunger were 9.1 ± 3.5 (2-17), 7.6 ± 3.2 (1-14) and 5.6 ± 2.8 (0-13), respectively. Grit-S score was negatively related to disinhibition score (r=-0.47, P<0.001), hunger score (r=-

0.19, P=0.05), BW (r=-0.24, P<0.05) and BMI (r=-0.23, P<0.05) and positively associated with CER score (r=0.23, P<0.05).

Together, Grit-S and disinhibition scores explained approximately 8% of the variability in BW and 7% of the variability in BMI. However, of the variables examined, disinhibition was the only significant predictor of BW and BMI in this sample of women (Table 6.1).

DISCUSSION

The purpose of this study was to cross-sectionally explore the association between grit and eating behaviors and body mass in premenopausal women who enrolled in a 1-year weight gain prevention intervention [15]. The first hypothesis was supported in that women with higher grit had lower disinhibition and hunger scores and BW and BMI, along with higher CER score. Grit was not found to be a significant predictor of BW or BMI. To our knowledge, this is the first study to examine the relationship between grit and eating behaviors known to influence body mass status.

The mean Grit-S score of women in the current study (3.4 ± 0.7) is comparable to previous studies which have reported mean scores ranging from 2.8 ± 0.7 to 3.7 ± 0.7 in adult populations [2,5,18]. Thus, grit among the current sample of women is likely representative of other adult women.

Grit is a non-cognitive personality trait that has been shown to predict success or commitment in various life settings [2-7]. Similarly, individuals with higher grit may have the propensity for greater BW control, as they may better maintain effort and interest in weight management activities, even during setbacks (e.g., minor weight gain) [2-7] or challenges (e.g., in the presence of food/diet saboteurs). Though high CER has inconsistently been linked to lower BW, more highly restrained individuals tend to have lower energy and overall food intakes compared to individuals with lower CER [19]. Disinhibition is consistently and positively related to BW and BMI [20] and may increase an individual's susceptibility to food overconsumption in certain situations and/or emotional states that may result in weight gain over time. The significant relationship between grit and disinhibition suggests that noncognitive traits may facilitate dispositions that foster control over BW regulation and that

disinhibition may serve as the behavioral proxy of grit. Grit can be easily measured through eight non-food related questions, thereby preventing questionnaire fatigue and avoiding foodand weight-related triggers that may arise from eating-related survey instruments. Because the usefulness of grit as a predictor of BW or BMI, compared to other eating behavior traits, is unclear, further evaluation of grit in BW management success over time is necessary. For example, individuals with higher grit and lower disinhibition may require less intervention and/or support to achieve BW goals, compared to individuals with lower grit and higher disinhibition who may require additional support, as suggested by Salles and colleagues [5], as well as for those that may be susceptible to barriers preventing goal achievement (in this case, successful weight gain prevention or management). If such a relationship exists, grit could be easily assessed during screening visits and interventions and/or treatment options could be tailored to improve adherence or outcomes.

The current cross-sectional study provides a snapshot of grit and eating behaviors in a small sample of women and cannot infer causation. No studies have examined the ability of an intervention to modify grit toward the goal of weight management or whether grit is correlated with weight management over time. Further, it is unknown whether grit is an antecedent to disinhibition in body mass regulation. Relationships explored here should be expanded to larger samples of men and women and tested in longitudinal studies. Instruments used in the current study can be easily administered in clinical and community settings.

Women with higher grit may have better control over BW and BMI as supported by lower disinhibition and hunger and higher CER, yet, disinhibition score but not Grit-S score predicted BW and BMI. Further research is warranted to determine the salience of grit in the field of weight management and to gain a better understanding of how grit relates to other eating behaviors in the context of long-term BW regulation.

REFERENCES

- Mitchell NS, Catenacci VA, Wyatt HR, Hill JO: Obesity: overview of an epidemic. Psychiatr Clin North Am 2011;34:717-732.
- 2. Duckworth AL, Peterson C, Matthews MD, Kelly DR: Grit: perseverance and passion for long-term goals. J Pers Soc Psychol 2007;92:1087-1101.
- Duckworth AL, Quinn PD: Development and validation of the short grit scale (grit-s). J Pers Assess 2009;91:166-174.
- Eskreis-Winkler L, Shulman EP, Beal SA, Duckworth AL: The grit effect: predicting retention in the military the workplace school and marriage. Front Psychol 2014;5:36 doi:103389/fpsyg201400036.
- Salles A, Cohen GL, Mueller CM: The relationship between grit and resident well-being. Am J Surg 2014;207:251-254.
- 6. Burkhart RA, Tholey RM, Guinto D, Yeo CJ, Chojnacki KA: Grit: a marker of residents at risk for attrition? Surgery 2014;155:1014-1022.
- 7. Reed J, Pritschet BL, Cutton DM: Grit conscientiousness and the transtheoretical model of change for exercise behavior. J Health Psychol 2013;18:612-619.
- 8. Stunkard AJ, Messick S: The three-factor eating questionnaire to measure dietary restraint, disinhibition and hunger. J Psychosom Res 1985;29:71-83.
- Lowe MR, Maycock B: Restraint, disinhibition, hunger and negative affect eating. Addict Behav 1988;13:369-377.
- Beiseigel JM Nickols-Richardson SM: Cognitive eating restraint scores are associated with body fatness but not with other measures of dieting in women. Appetite 2004;43:47-53.
- 11. Nickols-Richardson SM, Coleman MD, Volpe JJ, Hosig KW: Perceived hunger is lower and weight loss is greater in overweight premenopausal women consuming a lowcarbohydrate/high-protein vs high-carbohydrate/low-fat diet. J Am Diet Assoc 2005;105:1433-1437.

- 12. Urbanek JK, Metzgar CJ, Hsiao PY, Piehowski KE, Nickols-Richardson SM: Increase in cognitive eating restraint predicts weight loss and change in other anthropometric measurements in overweight/obese premenopausal women. Appetite 2015;87:244-250.
- 13. Hill JO, Wyatt HR, Reed GW, Peters JC: Obesity and the environment: where do we go from here? Science 2003; 299:853-855.
- 14. Savage JS, Hoffman L, Birch LL: Dieting, restraint, and disinhibition predict women's weight change over 6 y. Am J Clin Nutr 2009;90:33-40.
- Metzgar CJ, Nickols-Richardson SM: Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial. JMIR Res Protoc 2015;4:e36 doi:102196/resprot4008.
- 16. Zung WW: A self-rating depression scale. Arch Gen Psychiatry 1965;12:63-70.
- Pozzebon J, Damian RI, Hill PL, Lin Y, Lapham S, Roberts BW: Establishing the validity and reliability of the Project Talent Personality Inventory. Front Psychol 2013;4:968 doi:103389/fpsyg201300968.
- 18. Von Culin KR, Tsukayama E, Duckworth AL: Unpacking grit: motivation correlates of perseverance and passion for long-term goals. J Posit Pyschol 2014;9:306-312.
- 19. McLean JA, Barr SI: Cognitive dietary restraint is associated with eating behaviors lifestyle practices personality characteristics and menstrual irregularity in college women. Appetite 2003;40:185-192.
- 20. Bryant EJ, King NA, Blundell JE: Disinhibition: its effects on appetite and weight regulation. Obes Rev 2007;9:409-419.

Variable	Beta	b-value	F-value	Adjusted R ²
Body weight				
Model 1 (n=79)			4.458*	0.081
Grit-S score	-0.117	-3.428		
Disinhibition score	0.252*	1.516		
Model 2 (n=79)			8.022**	0.083
Disinhibition score	0.307**	1.850		
Body mass index				
Model 1 (n=79)			4.002*	0.071
Grit-S score	-0.121	-1.272		
Disinhibition score	0.232	0.503		
Model 2 (n=79)			7.054**	0.072
Disinhibition score	0.290**	0.627		

Table 6.1. Standardized (beta) and unstandardized (b-value) regression coefficients for predictors of body weight and body mass index in a sample of women before undergoing a weight gain prevention intervention

*P≤0.05; **P≤0.01. Backward elimination regression, including significantly related variables in Model 1 for BW (Grit-S score: r=−0.24, P<0.05, disinhibition score: r=0.31, P<0.01) and BMI (Grit-S score: r=−0.23, P<0.05, disinhibition score: r=0.29, P<0.01).

ABSTRACT

OBJECTIVE: This prospective study explored the association of grit, cognitive eating restraint (CER), disinhibition and hunger in predicting body weight (BW) and BW change over 12 months. **METHODS:** The Short Grit Scale (Grit-S) and Eating Inventory (EI) were completed by 79 women [mean±SD, age: 31.5±8.2 y, BW: 76.3±19.2 kg, body mass index (BMI): 28.0±6.9 kg/m²] at baseline and after completion of a 1-year weight gain prevention intervention.

RESULTS: Average Grit-S, CER, disinhibition and hunger scores did not significantly change over time. Body weight and BMI, respectively, at baseline and month 12 were negatively associated with baseline grit (baseline: r=-0.24, P<0.05; r=-0.23, P<0.05; month 12: r=-0.25, P<0.05; r=-0.23, P<0.05) and positively associated with disinhibition (baseline: r=0.31, P<0.01; r=0.29, P<0.01; month 12: r=0.34, P<0.01; r=0.31, P<0.01). Change in BW was not associated with any measure. Lower baseline disinhibition and greater increase in CER over time were related to successful weight gain prevention (both P<0.05). Disinhibition predicted month 12 BW (F=9.80, P<0.01) and BMI (F=8.08. P<0.01). No measures significantly predicted BW change. **CONCLUSIONS:** Successful body weight regulation over 12 months was characterized by low disinhibition at baseline and an increase in CER over time. Additionally, high grit and low disinhibition at baseline were related to lower BW and BMI at end of study. Future interventions should consider strategies that manage disinhibition and increase CER. **KEYWORDS:** Disinhibition; Grit; Obesity; Weight management; Weight regulation; Women

⁶To be submitted to *Eating Behaviors*.

1. INTRODUCTION

Regulation and management of body weight (BW) over time remains a challenge for most individuals, namely women during young adulthood (Ball, Brown, & Crawford, 2002; Sheehan, DuBrava, DeChello, & Fang, 2003). Weight management is made difficult by the complex interplay among lifestyle, environmental and genetic factors (Mitchell, Catenacci, Wyatt, & Hill, 2011). Additionally, personality traits may play a role in weight management (Sutin, Ferrucci, Zonderman, & Terracciano, 2011) and have been shown to be significant predictors of BW (Chapman, Fiscella, Duberstein, Kawachi, & Coletta, 2009; Terracciano, Sutin, McCrae, Deiana, Ferrucci, Schlessinger, Uda, & Costa, 2009). More attention to certain personality traits may help to tailor weight management approaches.

Grit is one such non-cognitive personality trait that may play a role in BW regulation. Defined as the "perseverance and passion for long-term goals" (Duckworth, Peterson, Matthews, & Kelly, 2007, p. 1087), grit encompasses the sustained interest and efforts of individuals to achieve long-term objectives even in presence of failures, plateaus or diversity. Previous research has shown grit to be a predictor of success in academic pursuits (Duckworth et al., 2007; Duckworth & Quinn, 2009; Eskreis-Winkler, Shulman, Beal, & Duckworth, 2014), workplace settings (Duckworth, Quinn, & Seligman, 2009; Eskreis-Winkler et al., 2014), military retention (Duckworth et al., 2007; Eskreis-Winkler et al., 2014) and relationships (Eskreis-Winkler et al., 2014). In the medical field, low grit may be a risk factor for not completing residency (Burkhart, Tholey, Guinto, Yeo, & Chojnacki, 2014) and has been predictive of lower levels of general psychological health among surgical residents (Salles, Cohen, & Mueller, 2014).

The cognitive personality traits of cognitive eating restraint (CER), disinhibition and hunger are known to influence BW management (Stunkard & Messick, 1985; Lowe & Maycock, 1988; Dykes, Brunner, Martikainen, & Wardle, 2003; Beiseigel & Nickols-Richardson, 2004; Nickols-Richardson, Coleman, Volpe, & Hosig, 2005; Hays & Roberts, 2008; Urbanek, Metzgar, Hsiao, Piehowski, & Nickols-Richardson, 2015). Cognitive eating restraint is the conscious restriction of dietary intake with the motive to control BW or induce weight loss (Stunkard et al., 1985; Hays et al., 2008), while disinhibition is characterized by the tendency to consume

excess amounts of foods in response to a variety of stimuli (Lowe et al., 1988; Hays et al., 2008). Hunger is defined as the susceptibility to consume foods in response to physiological signs and symptoms that signal the need for nourishment (Lowe et al., 1988). Disinhibition is consistently positively associated with anthropometric measurements (Bryant, King, & Blundell, 2007; French, Epstein, Jeffery, Blundell, & Wardle, 2012), while the relationships between CER and hunger and anthropometric measurements are not as clear (Drapeau, Provencher, Lemieux, Després, Bouchard, & Temblay, 2003). The interaction of these cognitive personality traits may predict current BW and BW regulation over time, and when considered in combination with non-cognitive personality traits such as grit, may provide further insight about the predictors of BW and BMI over time.

The current prospective study aimed to explore the association and role of grit, CER, disinhibition and hunger in predicting BW status and BW change over 12 months in a sample of healthy premenopausal women enrolled in a randomized controlled trial of weight gain prevention. It was hypothesized that baseline grit, disinhibition and hunger scores would be negatively associated and CER would be positively associated with BW and BMI at month 12. Grit also was hypothesized to be a significant predictor of body weight change over time.

2. METHODS

2.1 Participants

Healthy premenopausal women between the ages of 18 and 45 years with a BMI of >18.5kg/m² were recruited from the East Central Illinois (USA) communities of Champaign-Urbana. Women who met age (18-45 years) and BMI (>18.5kg/m²) criteria and had a desire to prevent weight gain were further screened for eligibility. Women who self-reported metabolic, cardiovascular or musculoskeletal abnormalities, used medications to manage chronic conditions or self-reported use of supplements or medications known to influence BW regulation were excluded. Additional exclusion criteria included a score of >50 on the Zung Self-Rating Depression Scale/Status Inventory (Zung, 1965), indicating the presence of depressive symptomology; amenorrhea (<8 menstrual cycles/y); or history of weight loss surgery. Women who were pregnant or planned to become pregnant, or who were currently lactating also were

excluded. Full details of this 1-year randomized controlled trial of weight gain prevention have been previously described (Metzgar & Nickols-Richardson, 2015).

The study protocol (#14397) was approved by the Institutional Review Board (IRB) for the Protection of Human Subjects at the University of Illinois at Urbana-Champaign. Prior to study participation, all participants provided written informed consent.

2.2 Measurements

Participants completed self-administered paper-based questionnaires upon enrollment (baseline) and at the end of the 1-year intervention (month 12). Body weight and BMI also were collected and evaluated at baseline and month 12.

2.2.1 Grit

Grit was assessed at baseline and month 12 using the Short Grit Scale or Grit-S (Duckworth & Quinn, 2009). This 8-item questionnaire utilizes a 5-point Likert scale where 1 = strongly disagree and 5 =strongly agree. Women rated how well statements such as "I have difficulty maintaining my focus on projects that take more than a few months to complete" and "Setbacks don't discourage me" described their tendencies (Duckworth & Quinn, 2009). Grit-S score was calculated by averaging the eight items with possible scores ranging from 1-5. A high Grit-S score is indicative of greater grit. Internal consistency reliability ranges from 0.73 to 0.84 in previous studies (Duckworth & Quinn, 2009; Von Culin, Tsukayama, & Duckworth, 2014; Eskreis-Winkler et al., 2014).

2.2.2 Eating constructs

Cognitive eating restraint, disinhibition and hunger were measured at baseline and month 12 using the Eating Inventory (EI) (Stunkard et al., 1985). This 51-item questionnaire is comprised of 36 true/false statements (Part 1) and 15 4- or 6-point Likert scale questions (Part 2), with 21 items that assess CER, 16 items that assess disinhibition and 14 items that address hunger. Standard scoring methods were used to calculate subscale scores for each construct (Stunkard et al., 1985), with higher scores indicating greater tendencies related to the corresponding construct.

2.2.3 Body weight and body mass index

Body weight and standing height were measured at baseline and month 12; these measurements were used to calculate BMI (kg/m²). A calibrated scale (Tanita 410GS, Arlington Heights, IL, USA) and stadiometer (Seca 700, Hanover, MD, USA) were used to measure BW and standing height, respectively; these measures were recorded to the nearest 0.1 kg and 0.1 cm, respectively.

2.2.4 Body weight change and weight gain prevention

Percent BW change was calculated by subtracting baseline BW from month 12 BW and dividing by baseline BW. Currently, there is no standardized definition of weight gain prevention; thus, weight gain prevention, or weight maintenance, was defined in the current study as a BW change of ±3% from baseline to month 12 (Stevens, Truesdale, McClain, & Cai, 2006). A weight loss of >3% of baseline BW or a weight gain of >3% of baseline BW was considered unsuccessful weight gain prevention.

2.3 Statistical Analyses

All women who provided complete Grit-S and El (n=79) questionnaires at baseline were included in intention to-treat analysis. Participant characteristics were analyzed by descriptive statistics (means±SD). Paired *t*-tests were used to evaluate changes in grit, El constructs, BW and BMI. Independent *t*-tests were used to compare differences between women who met weight gain prevention criteria with those who did not. The relationship between baseline Grit-S score and month 12 BW, BMI and percent BW change were examined using Pearson's correlation coefficients. To assess usefulness of grit as a predictor of percent BW change, backward elimination regression was used. All statistical analyses were conducting using the Statistical Package for the Social Sciences (version 22.0, 2013, IBM Corp, Armonk, NY, USA) with significance set at P<0.05.

3. RESULTS

Eighty-one participants enrolled in the study with 79 of these women providing complete Grit-S and EI data. A majority of women (age: 31.5±8.2 y) were non-Hispanic white (n=53, 67%), had earned at least a college degree (n=60, 76%) and had a total annual household

income of \geq \$50,000 (n=43, 54%). On average, women were overweight (BMI: 28.0±6.9 kg/m²), and BW ranged from 47.4 to 139.2 kg (mean: 76.3±19.2 kg). Body weight and BMI did not significantly change over the 1-year study interval. Percent BW change was 0.8±4.3%, with 48 women (60.8%) meeting weight gain prevention criteria.

Average Grit-S scores (ranges) at baseline and month 12 were 3.4 ± 0.7 (2.0-4.6) and 3.5 ± 0.7 (1.5-4.8). Baseline and month 12 Grit-S scores were not significantly different. At baseline, the average score (range) was 9.1 ± 3.5 (2-17) for CER, 7.6 ± 3.2 (1-14) for disinhibition and 5.6 ± 2.8 (0-13) for hunger. These scores did not change significantly over time with average scores of 9.4 ± 3.4 (2-18), 7.3 ± 3.4 (1-16) and 5.4 ± 3.1 (0-13) for CER, disinhibition and hunger, respectively, at month 12. Women who met weight gain prevention criteria had significantly lower baseline disinhibition scores (P<0.05) and significantly increased CER over the intervention (P<0.05) as compared to women who did not prevent weight gain.

Baseline Grit-S was negatively associated with BW and BMI at baseline (r=-0.24, P<0.05; r=-0.23, P<0.05) and month 12 (r=-0.25, P<0.05; r=-0.23, P<0.05). Baseline disinhibition was positively associated with BW and BMI at baseline (r=0.31, P<0.01; r=0.29, P<0.01) and month 12 (r=0.34, P<0.01; r=0.31, P<0.01). Neither CER nor hunger were significantly associated with BW and BMI at either baseline or month 12. Baseline Grit-S, CER, disinhibition and hunger were not associated with percent BW change.

Together, baseline CER, hunger, disinhibition and Grit-S scores explained approximately 8% of the variability in month 12 BW. Approximately 9% of the variability in month 12 BW and 8% of the variability in month 12 BMI was explained by baseline hunger, disinhibition and Grit-S scores. Together, disinhibition and Grit-S scores at baseline explained 10% of the variability in month 12 BW and 8% of the variability in month 12 BMI. Disinhibition alone explained approximately 10% of the variability in month 12 BW and 8% of the variability in month 12 BMI. Of the variability included in the current analyses, baseline disinhibition score was the only significant predictor of BW and BMI at the end of this 1-year intervention (Table 7.1). No variables were significant predictors of percent BW change.

4. DISCUSSION

The purpose of this study was to examine the role of grit and eating behaviors in predicting BW, BMI and BW change in a sample of premenopausal women enrolled in a 12month randomized controlled trial of weight gain prevention (Metzgar et al., 2015). The hypotheses established *a priori* were partially supported. Baseline grit and disinhibition were negatively associated with BW and BMI at month 12; no associations for CER, hunger or BW change were observed. Further, grit was not found to be a significant predictor of BW change over a 12-month intervention period. To our knowledge, this is the first study to examine grit as a predictor of BW change over time and only one other study has cross-sectionally explored the association of grit with BW and BMI (Metzgar & Nickols-Richardson, under review).

In this sample of healthy premenopausal women, grit at baseline and month 12 were highly correlated, and grit did not change over time. These results are consistent with others that have shown grit to be relatively stable over time (Duckworth et al., 2007; Duckworth & Quinn, 2009; Von Culin et al., 2014; Salles et al., 2014). Further, Grit-S scores in the current study are comparable with those reported in previous studies (Duckworth et al., 2007; Von Culin et al., 2014; Salles et al., 2014), indicating that the women enrolled in the present study are representative of women with similar characteristics.

Eating behavior scores at baseline also were highly correlated with scores at month 12. Women who successfully prevented weight gain over the 12-month intervention period were characterized by lower disinhibition levels at baseline and increases in CER over time. These findings are consistent with a previous weight gain prevention study of adult women (Levine, Klem, Kalarchian, Wing, Weissfeld, Qin, & Marcus, 2007) in which women who met weight gain prevention criteria (>2 pound BW loss or ±2 pounds of baseline weight) significantly increased CER and significantly decreased disinhibition over a 3-year study period. An additional study of weight gain prevention in young adults (Wing, Tate, Espeland, Gorin, LaRose, Robichaud, Erickson, Perdue, Bahson, & Lewis, 2013) also collected eating behaviors data, but results are not yet available.

Although the non-cognitive personality trait of grit did not predict success in BW management in this sample of women enrolled in weight gain prevention study, these findings

may be different in a study designed to induce BW loss. Grit has been identified as a marker of success or commitment in a variety of settings, even during challenges and failures (Duckworth et al., 2007; Duckworth & Quinn, 2009; Duckworth, Quinn, & Seligman, 2009; Eskreis-Winkler et al., 2014; Salles et al., 20014; Burkhart et al., 2014). The current weight gain prevention trial did not employ a dietary intervention or energy restriction so there was no marker of compliance; individuals with higher grit may be more likely to comply with an energy restricted diet and achieve weight loss.

While baseline levels of grit, CER, disinhibition and hunger did not predict percent BW change, they did predict absolute BW and BMI at 12 months. The finding of disinhibition as the strongest predictor of 12 month BW and BMI was to be expected as there is a consistent positive relationship between disinhibition, BW and BMI (Bryant et al., 2007; French et al., 2012). The identification of the stimuli or emotional states that lead to the susceptibility to overconsume foods may be useful in identify periods that may accompany weight gain and tailor interventions to individuals in order to better facilitate weight gain prevention. The relationships between CER and BW and BMI are not as consistent, but increasing levels of CER may be helpful in preventing weight gain over time or inducing BW loss (Levine et al., 2007; Savage, Hoffman, & Birch, 2009; Nickols-Richardson et al., 2005). Interventions that include self-regulation and other educational strategies that address CER may be useful in preventing weight gain over time.

Even though the non-cognitive personality trait of grit is relatively stable over time (Duckworth et al., 2007; Duckworth & Quinn, 2009; Von Culin et al., 2014; Salles et al., 2014), grit has been suggested to increase over the lifespan (Duckworth et al., 2007) and differ by individual domain (i.e., careers vs. hobbies vs. relationships) (Duckworth & Quinn, 2009). No studies have examined the role of an intervention to modify grit, but it has been suggested that targeted interventions may improve grit (Duckworth et al., 2007; Duckworth 2009).

Grit and disinhibition levels were significantly and negatively associated at both baseline and month 12 which supports that grit may serve as the non-cognitive proxy for disinhibition (Metzgar et al., under review). The Grit-S is shorter and more easily administered when compared to the EI which may be lengthy for some individuals and trigger emotions related to

statements that address feelings and perceptions related to food intake. In more intensive interventions or programs, grit may also be useful in identifying individuals who may require additional support (Burkhart et al., 2014; Salles et al., 2014). Although grit was not related to participant drop out in the current study, grit has been identified as a possible marker for attrition in surgical residents (Burkhart et al., 2014) and this relationship should be examined in more intensive or "strict" weight loss interventions or programs.

The relationships explored in the current study should be examined in similar samples of women and repeated in men. Additionally, more research is needed regarding the role of grit in weight management. The current study only assessed weight gain prevention; the role of grit in weight loss and weight loss maintenance offer opportunities for future research.

Weight regulation over time is a challenging task and a better understanding of the interaction between cognitive and non-cognitive personality trait is necessary. High levels of grit and low levels of disinhibition translated to lower BW and BMI at the end of a 12-month intervention in a sample of premenopausal women. Additionally, women with low disinhibition and those who were able to increase CER over 12 months were more successful in preventing weight gain over time. In order to facilitate weight gain prevention, attention should be given to managing disinhibition and increasing CER over time.

REFERENCES

- Ball, K., Brown, W., & Crawford, D. (2002). Who does not gain weight? Prevalence and predictors of weight maintenance in young women. *International Journal of Obesity and Related Metabolic Disorders, 26:* 1570-1578.
- Beiseigel, J. M. & Nickols-Richardson, S. M. (2004). Cognitive eating restraint scores are associated with body fatness but not with other measures of dieting in women. *Appetite*, 43: 47-53.
- Bryant, E. J., King, N. A., & Blundell, J. E. (2007). Disinhibition: its effects on appetite and weight regulation. *Obesity Reviews*, *9*: 409-419.
- Burkhart, R. A., Tholey, R. M., Guinto, D., Yeo, C. J., & Chojnacki, K. A. (2014). Grit: a marker of residents at risk for attrition? *Surgery*, *155*: 1014-1022.
- Chapman, B. P., Fiscella, K., Duberstein, P., Kawachi, I., & Colette, M. (2009). Can the influence of childhood socioeconomic status on men's and women's adult body mass be explained by adult socioeconomic status or personality? Findings from a national sample. *Health Psychology, 28:* 419-427.
- Drapeau, V., Provencher, V., Lemieux, S., Després, J. P., Bouchard, T. C., & Tremblay A. (2003).
 Do 6-y changes in eating behaviors predict changes in body weight? Results from the Québec
 Family Study. *International Journal of Obesity and Related Metabolic Disorders, 27:* 808-814.
- Duckworth, A. L., Peterson, C., Matthews, M. D., & Kelly, D. R. (2007). Grit: perseverance and passion for long-term goals. *Journal of Personality and Social Psychology*, *92*: 1087-1101.
- Duckworth, A. L. & Quinn, P. D. (2009). Development and validation of the short grit scale (grit-s). *Journal of Personality Assessment, 91*: 166-174.
- Duckworth, A. L., Quinn, P. D., Seligman, M. E. P. (2009). Positive predictors of teacher effectiveness. *The Journal of Positive Psychology*, *4*: 540-547.
- Dykes, J., Brunner E. J., Martikainen P. T., & Wardle, J. (2003). Socioeconomic gradient in body size and obesity among women: the role of dietary restraint, disinhibition and hunger in the Whitehall II study. *International Journal of Obesity, 28:* 262-268.

- Eskreis-Winkler, L., Shulman, E. P., Beal, S. A., & Duckworth, A. L. (2014). The grit effect: predicting retention in the military, the workplace, school and marriage. *Frontiers in Psychology*, *5*: 36. doi:10.3389/fpsyg.2014.00036.
- French, S. A., Epstein, L. H., Jeffery, R. W., Blundell, J. E., & Wardle, J. (2012). Eating behavior dimensions: associations with energy intake and body weight: a review. *Appetite*, *59*: 542-549.
- Hays, N. P. & Roberts, S. B. (2008). Aspects of eating behaviors "disinhibition" and "restraint" are related to weight gain and BMI in women. *Obesity (Silver Spring)*, *16*: 52-58.
- Levine, M. D., Klem, M. L., Kalarchian, M. A., Wing, R. R., Weissfeld, L., Qin, L., & Marcus, M.D. (2007). Weight gain prevention among women. *Obesity*, *15*: 1267-1277.
- Lowe, M. R. & Maycock, B. (1988). Restraint, disinhibition, hunger and negative affect eating. Addictive Behaviors, 13: 369-377.
- Metzgar, C. J. & Nickols-Richardson, S. M. (2015). Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial. *JMIR Research Protocols*, *4*: e36. doi:10.2196/resprot.4008.
- Metzgar, C. J. & Nickols-Richardson. (under review). Grit: association with eating behaviors, body weight and body mass index.
- Mitchell, N. S., Catenacci, V. A., Wyatt, H. R., & Hill, J. O. (2011). Obesity: overview of an epidemic. *Psychiatric Clinics of North America*, *34*: 717-732.
- Nickols-Richardson, S. M., Coleman, M. D., Volpe, J. J., & Hosig, K. W. (2005). Perceived hunger is lower and weight loss is greater in overweight premenopausal women consuming a lowcarbohydrate/high-protein vs high-carbohydrate/low-fat diet. *Journal of the American Dietetic Association, 105*: 1433-1437.
- Salles, A., Cohen, G. L., & Mueller, C.M. (2014). The relationship between grit and resident well-being. *American Journal of Surgery*, 207: 251-254.
- Savage, J. S., Hoffman, L., & Birch, L.L. (2009). Dieting, restraint, and disinhibition predict women's weight change over 6 y. *American Journal of Clinical Nutrition*, *90*: 33-40.

- Sheehan, T. J., DuBrava, S., DeChello, L. M., & Fang, Z. (2003). Rates of weight change for black and white Americans over a twenty year period. *International Journal of Obesity and Related Metabolic Disorders, 27*: 498-504.
- Stevens, J., Truesdale K. P., McClain, J. E., & Cai, J. (2006). The definition of weight maintenance. International Journal of Obesity (London), 30: 391-399.
- Sutin, A. R., Ferrucci, L., Zonderman, A. B., & Terracciano, A. (2011). Personality and obesity across the adult life span. *Journal of Personality and Social Psychology, 101*: 579-592.
- Stunkard, A. J. & Messick, S. (1985). The three-factor eating questionnaire to measure dietary restraint, disinhibition and hunger. *Journal of Psychometric Research, 29*: 71-83.
- Terracciano, A., Sutin, A. R., McCrae, R. R., Deiana, B., Ferrucci, L., Schlessinger, D., Uda, M., & Costa, P. T. (2009). Facets of personality linked to underweight and obesity. *Psychosomatic Medicine*, 71: 682-689.
- Urbanek, J. K., Metzgar, C. J., Hsiao, P. Y., Piehowski, K. E., & Nickols-Richardson, S. M. (2015).
 Increase in cognitive eating restraint predicts weight loss and change in other
 anthropometric measurements in overweight/obese premenopausal women. *Appetite*, 87: 244-250.
- Von Culin, K. R., Tsukayama, E., & Duckworth, A. L. (2014). Unpacking grit: motivation correlates of perseverance and passion for long-term goals. *The Journal of Positive Psychology*, 9: 306-312.
- Wing, R. R., Tate, D., Espeland, M., Gorin, A., LaRose J. G., Robichaud, E. F., Erickson, K., Perdue,
 L., Bahson, J., & Lewis C. E., (2013). *BMC Public Health*, *13*: 300. doi:10.1186/1471-2458-13-300.
- Zung, W. W. A self-rating depression scale (1965). Archives of General Psychiatry, 12: 63-70.

Variable	Beta	b-value	F-value	Adjusted R ²
Body weight (month 12)				
Model 1 (n=79)			2.696*	0.080
Baseline restraint score	0.014	0.076		
Baseline hunger score	-0.057	-0.392		
Baseline Grit-S score	-0.126	-3.731		
Baseline disinhibition score	0.293*	1.782		
Model 2 (n=79)			3.638*	0.092
Baseline hunger score	-0.058	-0.397		
Baseline Grit-S score	-0.123	-3.642		
Baseline disinhibition score	0.293*	1.783		
Model 3 (n=79)			5.374**	0.101
Baseline Grit-S score	-0.119	-3.510		
Baseline disinhibition score	0.280*	1.701		
Model 4 (n=79)			9.803**	0.101
Baseline disinhibition score	0.336**	2.043		
Body mass index (month 12)				
Model 1 (n=79)			3.137*	0.076
Baseline hunger score	-0.090	-0.223		
Baseline Grit-S score	-0.115	-1.237		
Baseline disinhibition score	0.278*	0.614		
Model 2 (n=79)			4.414*	0.080
Baseline Grit-S score	-0.108	-1.163		
Baseline disinhibition score	0.257*	0.568		
Model 3 (n=79)			8.081**	0.083
Baseline disinhibition score	0.308**	0.681		

Table 7.1. Standardized (beta) and unstandardized (b-value) regression coefficients for predictors of body weight and body mass index at the conclusion of a 1-year randomized controlled trial of weight gain prevention in in a sample of premenopausal women

*P≤0.05; **P≤0.01. Backward elimination regression including significantly related variables.

The current 1-year randomized controlled trial was successful in preventing weight gain in a sample of healthy normal weight, overweight and obese premenopausal women between the ages of 18 and 45 years, but there were no differences between women who received nutrition education and those who were randomized to a control group (CON). Further, there were no significant differences in weight gain prevention between women randomized to receive the weight gain prevention intervention from a dietitian (RDG) compared to those who received the same weight gain prevention intervention from a counselor (CSG). Taken together, these findings demonstrate that weight gain prevention is possible; however, nearly 40% of the sample population still struggled with weight gain prevention. Further evaluation of determinants of prospective weight gain and strategies for preventing weight gain is necessary to effectively reduce the burden of adult obesity.

Although no clinically significant differences were observed between the RDG, CSG and CON over time for any primary or secondary outcome measures, the collection and evaluation of clinical markers beyond body weight is a major strength of this study. Health is multidimensional, and using body weight as the "gold standard" for health may actually be more harmful, as body weight is only one component of the multidimensional health puzzle. Other indicators of health, including hip and waist circumference, body composition, blood pressure and blood lipid levels, in combination with body weight may provide greater insight into metabolic health of an individual.

In addition to clinical outcomes, health perceptions, behaviors and psychosocial outcomes should also be considered as behavior and lifestyle choices directly impact body weight and other health outcomes. Grit, along with other personality traits, should be evaluated as these may play a role in body weight regulation. Behavior and lifestyle changes, in the form of a weight gain prevention/health promotion approach, may play a larger role in improving long-term health and weight status as compared to a weight loss treatment approach. The current weight gain prevention trial was designed to remove the emphasis from energy restriction in order to encourage "big picture" healthy dietary, physical activity and

lifestyle behaviors (i.e., sleep, stress management) in order to promote health and weight maintenance.

Energy restriction and other methods to induce weight loss are successful in the short term (1,2), but these methods have failed to produce sustained weight loss and significantly impact rates of overweight and obesity (3,4). Beyond lack of success related to weight loss and weight loss maintenance, unintended consequences such as metabolic abnormalities due to weight cycling, increased preoccupation with foods and decreased self-esteem may result from an emphasis on weight loss (3). Weight gain prevention and health promotion approaches that serve as an alternative to treatment via weight loss are warranted, but little is known regarding the determinants of weight gain prevention due to a limited body of evidence with discouraging findings.

The Health at Every Size (HAES) movement is gaining momentum (5), as is the notion that individuals may be overweight or obese and metabolically healthy or normal weight and metabolically unhealthy (6,7). The HAES approach to weight management focuses on health promotion by improving metabolic fitness and decreasing chronic disease risk by increasing physical activity and consumption of nutrient dense foods (3,5). Treatment approaches, including dose and intensity, may differ between metabolically healthy and unhealthy individuals (7).

Future research should focus on identifying strategies to improve overall health and well-being while minimizing weight gain that accompanies young adulthood and continues through midlife. While the current trial included only premenopausal women, weight gain prevention should be addressed across the lifespan for both genders. Specifically, strategies for and determinants of weight gain prevention may differ between different age segments of the population. The perceived risk of weight gain for the college population may be much different than that of post-partum woman. Further, men and post-menopausal women are underrepresented in much of the weight management literature, and more attention should be given to these populations. A one-size-fits-all approach is not appropriate to significantly impact health by preventing weight gain.

Another consideration for future weight gain prevention interventions is the inclusion of couples. Research has shown that spousal body weights are correlated and that when one spouse's body weight increases, so does that of the other spouse; the likelihood of developing obesity increases as well (8). Even though outcome expectations, self-efficacy or perceived susceptibility may differ between partners, similar behaviors are still likely shared. Additionally, spouses may serve as an external source of support and accountability, both of which have been identified as key facilitators to weight loss and weight loss maintenance (9).

As most weight gain prevention studies have fallen short of the objective to prevent weight gain, further consideration should be given to intervention design. The FITT (frequency, intensity, time and type) principle, is used in the development of exercise prescriptions (10); these components can also be applied to developing successful weight gain prevention interventions. Frequency would relate to how often the intervention component occurs (i.e., daily, weekly, monthly, yearly); intensity (i.e., the level of personal commitment); time (i.e., the length of the intervention, including follow up); and type (i.e., newsletters, web or technology based, individual counseling, group sessions). Using the FITT principle also allows for tailored programs or interventions.

Finally, an agreement as to what constitutes successful weight gain prevention is needed. Multiple definitions have been used inconsistently throughout the literature which makes comparisons difficult in the already limited body of evidence. A standardized definition of weight gain prevention would allow comparisons across studies to determine the true prevalence of successful weight gain prevention. Further, identifying individuals who are successful at weight gain prevention would allow researchers to gain insight into habits, characteristics or strategies used by these individuals, similar to the concept of the National Weight Control Registry (11).

In summary, more interventions, that include long-term follow ups with continued contact and that target strategies for and determinants of weight gain prevention are warranted. Further, larger sample sizes are necessary, as are studies that target different segments of the lifespan. Additionally, future interventions should include more robust indicators of health such as blood pressure, blood lipid levels, dietary and physical activity

behaviors and related psychosocial outcomes such as self-efficacy, social support and selfregulation and move away from body weight as the best (or only) indicator of health.

REFERENCES

- 1. Jeffery RW, Drewnowski A, Epstein LH, et al. Long-term maintenance of weight loss: current status. Health Psychol. 2000;19:5-16.
- Hill JO, Thompson H, Wyatt H. Weight maintenance: what's missing? J Am Diet Assoc 2005 May;105(5 Suppl 1):S63-S66.
- Bacon L, Aphramor L. Weight science: evaluating the evidence for a paradigm shift. Nutr J. 2011;10:9. doi: 10.1186/1475-2891-10-9.
- 4. Mann T, Tomiyama J, Westling E, Lew AM, Samuels B, Chatman J. Medicare's search for effective obesity treatments: diets are not the answer. Am Psychol. 2007;62:220-33.
- Ikeda J, Amy NK, Ernsberger P, et al. The National Weight Control Registry: a critique. J Nutr Educ Behav. 2005;34:203-5.
- Durward CM, Hartman TJ, Nickols-Richardson SM. All-cause mortality risk of metabolically healthy obese individuals in NHANES III. J Obes. 2012;2012:460321. doi: 10.1155/2012/460321.
- Fabbrini E, Yoshino J, Yoshino M, et al. Metabolically normal obese people are protected from adverse effects following weight gain. J Clin Invest. 2015;125:787-98.
- Cobb LK, McAdams-DeMarco MA, Gudzune KA, et al. Changes in body mass index and obesity risk in married couples over 25 years: the ARIC cohort study. Am J Epidemiol. 2015; doi: 10.1093/aje/kwv112 [Epub ahead of print].
- Metzgar CJ, Preston AG, Miller DL, Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. J Hum Nutr Diet. 2015;28:593-603.
- 10. American College of Sports Medicine. ACSM's guidelines for exercise testing and prescription (7th ed). Baltimore: Lippincott Williams & Wilkins; 2006.
- Klem ML, Wing RR, McGuire MT, Seagle HM, Hill JO. A descriptive study of individuals successful at long-term maintenance of substantial weight loss. Am J Clin Nutr. 1997;66:239-46.

UNIVERSITY OF ILLINOIS AT URBANA-CHAMPAIGN

Office of the Vice Chancellor for Research Institutional Review Board 528 East Green Street Suite 203 Champaign, IL 61820



March 7, 2014

Sharon Nickols-Richardson Food Science & Human Nutrition 260A Bevier Hall 905 S. Goodwin Ave M/C 182

RE: Determinants of weight gain prevention in adult women IRB Protocol Number: 14397

Dear Dr. Nickols-Richardson:

Your response to stipulations for the project entitled *Determinants of weight gain prevention in adult women* has satisfactorily addressed the concerns of the UIUC Institutional Review Board (IRB) and you are now free to proceed with the human subjects protocol. The UIUC IRB approved, by expedited review, the protocol as described in your IRB-1 application with stipulated changes. The expiration date for this protocol, UIUC number 14397, is 03/05/2015. The risk designation applied to your project is *no more than minimal risk.* Certification of approval is available upon request.

Copies of the attached date-stamped consent form(s) must be used in obtaining informed consent. If there is a need to revise or alter the consent form(s), please submit the revised form(s) for IRB review, approval, and date-stamping prior to use.

Under applicable regulations, no changes to procedures involving human subjects may be made without prior IRB review and approval. The regulations also require that you promptly notify the IRB of any problems involving human subjects, including unanticipated side effects, adverse reactions, and any injuries or complications that arise during the project.

If you have any questions about the IRB process, or if you need assistance at any time, please feel free to contact me or the IRB Office, or visit our Web site at http://www.irb.illinois.edu.

Sincerely,

Anita Balgopal, PhD Director, Institutional Review Board

Attachment(s)

c: Catherine Metzgar

UNIVERSITY OF ILLINOIS AT URBANA-CHAMPAIGN



Dr. Shelly Nickols-Richardson, Professor and Head nickrich@illinois.edu

Department of Food Science and Human Nutrition College of Agricultural, Consumer and Environmental Sciences 260 Bevier Hall

Informed Consent Form

905 S. Goodwin Avenue Urbana, IL 61801

The University of Illinois, Urbana-Champaign

Title of Project: Determinants of Weight Gain Prevention in Adult Women

 Principal Investigator:
 Sharon M. (Shelly) Nickols-Richardson, PhD, RD

 Department of Food Science and Human Nutrition
 260A Bevier Hall

 Urbana, IL 61801
 E-mail: nickrich@illinois.edu

1. **Purpose of the study:** The purpose of this research is to identify factors that will help to prevent weight gain in women who are between the ages of 18 to 45 years. This study will look at the ability of nutrition education to prevent weight gain and to see which things related to this education are the most helpful in weight gain prevention. A total of 180 premenopausal women who are normal weight, overweight or obese (have a body mass index of greater than 18.5 kg/m²) will be in this study. Individuals who have a body mass index of less than 18.5 kg/m² are considered underweight and are not eligible for the study. Body mass index is a measure of your weight in relation to your height and is used as an indicator of weight status. The Agricultural Experiment Station at the University of Illinois is funding this study.

2. **Procedures to be followed:** As a participant, you will be asked to complete screening forms. If enrolled in the study, you will attend 2 informational meetings, 5 blood collection sessions (testing sessions), and weekly nutrition education sessions for 4 months and monthly nutrition education sessions for 8 months during this this 12-month study. During the study, you will complete several procedures:

Informational Meeting - 1 hour of your time.

You will attend an initial informational meeting where the details of the study will be explained. You will have plenty of time to ask questions of the investigators and will have up to 2 weeks after the meeting to decide not to be in the study. No further measurements will be taken until we receive your final agreement or consent to be in the study.

At the informational meeting, you will:

- review your signed Informed Consent Form and reaffirm or agree again your desire and willingness to be in the study (within 2 weeks of the meeting);
- ask questions about the study; and

• be instructed on how to use an accelerometer (a tool that measures your activity) and how to properly fill out a 4-day food record and 7-day physical activity record.

After the informational meeting and your agreement to be in the study, you will undergo baseline testing. After this testing, you will be randomized into a nutrition education group. This means that you will be put into 1 of 3 groups by chance. Once you are assigned to a group, you will attend another meeting specifically for your assignment. At this meeting that will take 1 hour of your time, we will discuss the study, answer any of your questions, and give you a schedule for your testing sessions, weekly nutrition education meetings, and the date on which the study will start. You will stop using or taking any vitamin and mineral supplements until the end of the study.

Nutrition Education Groups

The 3 groups include: 1) I nutrition education group; 2) 1 nutrition education group; and 3) 1 waitlist control group. If you are in 1 of the nutrition education groups, you will meet weekly for the first 16 weeks, and then once each month for the remaining 8 months. Each of these meeting will last for 1 hour. If you are in the wait-list control group, you will attend only the testing sessions for the first 12 months of the study, and then you will be assigned to 1 of the nutrition education groups at a later date.

Testing Sessions

At baseline (before the nutrition education begins), and after 3, 6, 9, and 12 months of the study, you will complete a testing session. This will take about 2 hours of your time at each testing session. During each testing session, you will need to:

- arrive in Bevier Hall on the campus of the University of Illinois, Urbana-Champaign at your scheduled appointment day and time after having not had anything to eat from 11:00 pm until after the testing session is over;
- turn in a 4-day food record, accelerometer counts, and a 7-day physical activity record;
- have your height, weight, body fat percentage, waist circumference, and hip circumference measured;
- complete a General Demographic Form; Current Illness questionnaire; a General Symptoms questionnaire; and the SF-36 Health Status questionnaire;
- · complete the Zung Scale; Eating Inventory; Grit-S Questionnaire; and SCT Questionnaire;
- · have your blood pressure and resting heart rate measured;
- have about 30 mL (about 2 Tablespoons) of whole blood drawn from your arm by a trained professional who is trained in drawing blood from people; and
- · eat breakfast foods and beverages if desired.

<u>Weekly Nutrition Education Sessions</u> -1 hour of your time each week for the first 16 weeks of the study and then 1 hour of your time each month for the last 8 months of the study.

At these sessions, we will discuss general nutrition information, food purchasing and preparation, eating in restaurants and in social settings, recipe modifications, physical activity, fruits and vegetables, portion control, breakfast consumption, healthy snacking, family menu planning, and grocery shopping, among other nutrition topics. A professional will lead your nutrition education session. Other women in the study will attend your sessions.

Follow-up Meeting

Once everyone has finished the whole study, we will have 1 last meeting. At this meeting, you will be given your study results, and we will provide information about the study.

Clothing during Testing Sessions

You should wear a short-sleeved or loose-fitting shirt for the blood draw and elastic-waisted pants for the waist and hip circumference measurements. Do not wear expensive jewelry or many clothing accessories to the test sessions, as you will have to remove these items during measurement of weight.

Length of Time for Study Procedures

Your appointments may take more or less time than estimated to complete each procedure. You will be given plenty of time to complete measurements and blood draws and to understand your nutrition education information.

3. **Discomforts and risks:** There are 3 potential risks: 1) blood draws, 2) still photos and 3) exposure of personal information. There is minimal risk involved in blood draws. A bruise or a little bleeding may result from blood collection procedures with no known harmful effects to your health or well-being. In order to minimize bruising and bleeding, a trained individual will draw all blood samples. You may become slightly lightheaded or nauseous during blood draws, but you may sit or recline for as long as you need to minimize discomfort. After each blood draw, you will be provided with breakfast foods and beverages. Two attempts to draw blood (or 2 needle sticks) will be allowed. If a second attempt is unsuccessful, no further tries for blood collection will be done. Research personnel will take universal blood precautions during handling of all blood samples.

During the study, still photographs will be taken of participants. Photographs will be taken during nutrition education sessions and during testing sessions. Your image as captured in a photograph may be used during presentations and in publications put together by the investigators of the study. Your name will not be attached to your picture. Any information that you give to the investigators will not be individually attached to your picture. It is possible that someone will recognize your image or your picture. Therefore, your participation in the study will not be anonymous and the confidentiality of your being is this study may not be guaranteed. There is minimal risk involved with a still photo; however, your image as part of this study may pose a discomfort to you. If you do not want to have your picture or still photo taken during the study, please inform the investigators.

Please check one of the boxes below.

I consent to have my photograph taken throughout the course of the study: 🗌 Yes 🗌 No

Exposure of personal information may occur during data collection and education sessions, and you may experience slight embarrassment and/or discomfort. You may skip any questions you do not wish to answer and your data collection sessions will be completed in private. During group education sessions, investigators will keep all information private, but it cannot be guaranteed that other participants will do the same.

4. Abnormal test results: Your blood pressure result will be shared with you at each testing session. If your systolic (top number) blood pressure measurement is 140 mmHg or more and/or if your diastolic (bottom number) blood pressure measurement is 90 mmHg or more, you will be told to seek immediate medical attention from your health care provider, and the investigator will remove you from the study.

The blood samples will not be analyzed until the study has been completed. However, after all participants have completed their blood draws, we will have a follow-up meeting. At this meeting, you will be given your personal results. If there are any study results that suggest that medical attention is needed (for example, high blood cholesterol), you will be made aware of this within 30 days of analysis and recommended to contact your health care provider for an examination. Dietary and physical activity records will also be analyzed at the end of the study, and information will be shared with you at this time.

This study is being done for research purposes and not for medical reasons. If you have a test result that is outside of the normal range, you will be informed and told to seek medical care from your health care provider.

The blood samples that you provide during this study will become property of the University of Illinois, Urbana-Champaign. Samples will be stored as needed to complete the study.

5. **Benefits to individual:** You may benefit from participation in this study in several ways including: 1) nutrition education; 2) analysis of blood cholesterol and blood sugar levels; 3) blood pressure measurements; and 4) dietary intake analysis. You may or may not prevent weight gain, but you will receive nutrition education. You will be provided with your results from tests after the study has been completed. Referral to appropriate health care professionals will be provided, if necessary based on your results after the completion of the study.

6. **Benefits to society:** This research will benefit society by finding what nutrition-related information helps women prevent weight gain.

7. **Duration/time of the procedures and study:** The study will have 2 informational meetings, 5 testing sessions, weekly nutrition education sessions for 4 months and then monthly nutrition education sessions for 8 months, and a follow-up meeting. The informational meetings will last approximately 1 hour, 5 testing sessions will last approximately 2 hours each, the weekly and then monthly nutrition education sessions will last 1 hour each, and the follow-up meeting will last about 1 hour. The total time for this study is estimated at 37 to 40 hours over a 12-month period. You may also spend time during each day thinking about, preparing and eating food and recording food intake and physical activity.

Absences and vacations will be expected and accommodated with attention given to scheduling of your data collection sessions and educational sessions. Make-up sessions will be offered if you are unable to attend an educational session, and educational materials will be mailed if necessary.

8. Statement of confidentiality: Your participation in this study is confidential. Data that you provide to the investigators will be stored and secured in the investigators' offices in locked filing cabinets. A 3-digit code or identification number will be assigned to you and used in place of your name. A master list of participants' code numbers will be kept in a separate locked filing cabinet. In the event of any publication or presentation resulting from the research, no personally identifiable information will be shared. Only the investigators of this study or students of the Primary Investigator will be allowed access to any data. It is up to you to share your individual results with your primary health care provider, if you so choose.

Your picture or still photo may be used in presentations and publications related to this study. Your name will not be attached to your photograph, and your data will not be able to be identified to your photograph.

The following may review records related to this research: the University of Illinois' Office for Human Research Protections, the University of Illinois' Institutional Review Board, and the Office of Human Research Protections in the U.S. Department of Health and Human Services.

9. **Right to ask questions:** Please contact Dr. Shelly Nickols-Richardson (Primary Investigator) at <u>nickrich@illinois.edu</u> with questions, complaints or concerns about the research. You can also contact the investigator if you feel that this study has harmed you. If you have any questions, concerns, problems about your rights as a research participant or would like to offer input, please contact the University of Illinois' Office for Human Research Protections at 217-333-2670 or e-mail <u>irb@illinois.edu</u>. The Office for Human Research Protections cannot answer questions about research procedures. All questions about research procedures can only be answered by the research personnel.

10. **Payment for participation:** You will receive a \$10 gift card for completing each of the testing sessions of the study. If you participate in all of the testing sessions, you will receive a total of \$50 in gift cards for the whole study. Total payments within one calendar year that exceed \$600 will require the University to report these payments to the IRS annually. This may require you to claim the compensation that you receive for being in this study as taxable income.

11. **Voluntary participation:** Your decision to be in this research is voluntary. You can stop at any time. You do not have to answer any questions you do not want to answer. Refusal to take part in or withdrawing from this study will involve no penalty or loss of benefits you would otherwise receive. There may be reasons for which the investigators find that you should discontinue the study, and they can remove you from the study at any time.

12. Event of injury: The University of Illinois does not provide medical or hospitalization insurance coverage for participants in this research study nor will the University of Illinois provide compensation for any injury sustained as a result of participation in this research study, except as required by law.

You must be between the ages of 18 to 45 years to be in this research study. If you agree to take part in this research study and with the information outlined above, please sign your name and indicate the date below.

You will be given a copy of this signed and dated consent form for your records.

UNIVERSITY OF ILLINDIS APPROVED CONSENT

MAR 0 5 2015

Participant Signature

Date

Date

Person Obtaining Consent
From:	Carmen <jmir.editorial.office@gmail.com></jmir.editorial.office@gmail.com>
Sent:	Tuesday, May 12, 2015 5:01 AM
То:	Metzgar, Catherine J
Cc:	jmir.editorial.assistant@gmail.com
Subject:	Re: Permission to reprint

Please see the copyright at the bottom of your article and include that statement. You do not need our permission as it is your work and under a CC-BY license. Regards, Carmen

Sent from my iPhone

On 12 May, 2015, at 01:39, Metzgar, Catherine J <<u>metzgar2@illinois.edu</u>> wrote:

Hi Allison,

As I am putting my dissertation together, I would like to request permission to reprint the following paper of research that I conducted as a graduate student. Can you please advise on how to obtain permission to reprint?

Metzgar CJ, Nickols-Richardson SM. Determinants of weight gain prevention in young adult and midlife women: study design and protocol of a randomized controlled trial. *JMIR Res Protoc.* 2015;4(1):e36. doi: 10.2196/resprot.4008.

http://www.ncbi.nlm.nih.gov/pubmed/25831450

Thanks for your help.

Catherine Metzgar, RD Doctoral Candidate University of Illinois at Urbana-Champaign

From:	김정원 <kimjwe@snue.ac.kr></kimjwe@snue.ac.kr>
Sent:	Tuesday, June 09, 2015 2:08 AM
То:	Metzgar, Catherine J
Subject:	Re:FW: RE: RE: Permission to reprint

Dear Ms. Catherine Metzgar,

For your request on the citation of your article published in NRP in your dissertation, the editorial board of NRP agreed to let you inform that you can make a statement as below:

This article previously appeared in its entirety as Sharon M. Nickols-Richardson, Kathryn E. Piehowski, Catherine J. Metzgar, Debra L. Miller and Amy G. Preston. Changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr Res Pract. 2014;8(6):695-704; doi:10.4162/nrp.2014.8.6.695.[Epub ahead of print]. This is an Open Access article distributed under the terms of the Creative Commons Attribution Non-Commercial License (<u>http://creativecommons.org/licenses/by-nc/3.0/</u>)which permits unrestricted non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

We feel sorry for the delay and any inconvience while communicating with us, and hope you have continuous interests in publishing in NRP.

Thank you.

Sincerely,

Jeong-Weon Kim, Ph.D Editor of NRP

----- 원본 메일 내용 ------

보낸 사람 : "NutritionRP"<<u>nutritionrp@daum.net</u>>

받는 사람: "김정원" <kimjwe@snue.ac.kr>

메일 제목 : FW: RE: RE: Permission to reprint

보낸 날짜 : Tue, 09 Jun 2015 14:58:24 +0900 (KST)

----- 원본 메일 ------

보낸사람: "NutritionRP" <<u>nutritionrp@daum.net</u>>

받는사람: "Metzgar, Catherine J" <<u>metzgar2@illinois.edu</u>>

날짜: 2015년 6월 09일 화요일, 11시 51분 01초 +0900 **제목**: RE: RE: Permission to reprint

Dear. Dr. Metzgar, Catherine J

We are under discussion for that part. We will give a reply as soon as possible.

Sincerely,

Editorial office of NRP ------ 원본 메일 ------보낸사람: "Metzgar, Catherine J" <<u>metzgar2@illinois.edu</u>> 받는사람 : "Inutritionrp@daum.net'" <<u>nutritionrp@daum.net</u>> 날짜: 2015년 6월 09일 화요일, 04시 29분 21초 +0900 제목: RE: Permission to reprint Hello,

Please see my previous request below. How should I proceed? Any advisement is much appreciated.

Ms. Catherine Metzgar

From: Metzgar, Catherine J Sent: Monday, June 01, 2015 11:39 AM To: 'nutritionrp@daum.net' Subject: RE: Permission to reprint

Hello,

Do you have any further response to this request?

Thanks.

From: Metzgar, Catherine J Sent: Friday, May 29, 2015 8:41 AM To: <u>nutritionrp@daum.net</u> Subject: RE: Permission to reprint

The final file is attached. What I would like to do for my dissertation is to include the paper as a text file only. A statement would be included to state that the paper had been previously published in it's an entirety.

Here is an examples: This article previously appeared in its entirety as Metzgar CJ, Preston AG, Miller DL. Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. *J Hum Nutr Diet.* 2014 Sept 18. doi: 10.1111/jhn.12273. [Epub ahead of print]. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is noncommercial and no modifications or adaptations are made.

Does your journal have any statement like this that should be made.

Thank you,

Ms. Catherine Metzgar, RD

From: nutritionrp@daum.net [mailto:nutritionrp@daum.net] Sent: Friday, May 29, 2015 6:57 AM To: Metzgar, Catherine J Subject: RE: Permission to reprint

Dear Sir,

Reprints will not print again.

Can you send the final file?

If you have any other questions, please contact us.

Sincerely, Editorial office of NRP

----- 원본 메일 ------

보낸사람: "Metzgar, Catherine J" <<u>metzgar2@illinois.edu</u>>

世는사람: "<u>nutritionrp@hanmail.net</u>" <<u>nutritionrp@hanmail.net</u>>

날짜: 2015년 5월 28일 목요일, 09시 49분 20초 +0900

제목: Permission to reprint

As I am putting my dissertation together, I would like to request permission to reprint the following paper of research that I conducted as a graduate student. Can you please advise on how to obtain permission to reprint?

Nickols-Richardson SM, Piehowski KE, Metzgar CJ, Miller DL, Preston AG. Changes in body weight, blood pressure and selected metabolic biomarkers with an energy-restricted diet including twice daily sweet snacks and once daily sugar-free beverage. Nutr Res Pract. 2014 Dec;8(6):695-704.

Thanks for your help.

Catherine Metzgar, RD Doctoral Candidate University of Illinois at Urbana-Champaign

Nutrition Research and Practice 편집국 e-mail: <u>nutritionrp@hanmail.net</u> 한국영양학회 윤미옥 연구원: tel: 02-3452-0433, fax:02-3452-3018 대한지역사회영양학회 손수현 간사 : <u>tel:02-749-0745</u>, fax:02-749-0746 Homepage : <u>http://www.e-nrp.org</u>

nutritionrp @ daum.net

Nutrition Research and Practice 편집국 e-mail: <u>nutritionrp@hanmail.net</u> 한국영양학회 윤미옥 연구원: tel: 02-3452-0433, fax:02-3452-3018 대한지역사회영양학회 손수현 간사 : <u>tel:02-749-0745</u>, fax:02-749-0746 Homepage : http://www.e-nrp.org

📬 nutritionrp @ daum.net

Nutrition Research and Practice 편집국 e-mail: <u>nutritionrp@hanmail.net</u> 한국영양학회 윤미옥 연구원: tel: 02-3452-0433, fax:02-3452-3018 대한지역사회영양학회 손수현 간사 : <u>tel:02-749-0745</u>, fax:02-749-0746 Homepage : http://www.e-nrp.org

nutritionrp @ daum.net

From:	Wiley Global Permissions <permissions@wiley.com></permissions@wiley.com>
Sent:	Thursday, May 14, 2015 8:58 AM
То:	Metzgar, Catherine J
Subject:	RE: Permission to reprint

Dear Catherine:

Thank you for your request to use Wiley content in your thesis.

Permission is hereby granted for the use requested subject to the usual acknowledgements (author, title of material, title of book/journal, ourselves as publisher). You shall also duplicate the copyright notice that appears in the Wiley publication in your use of the Material.

Any third party material is expressly excluded from this permission. If any of the material you wish to use appears within our work with credit to another source, authorization from that source must be obtained.

This permission does not include the right to grant others permission to photocopy or otherwise reproduce this material except for accessible versions made by non-profit organizations serving the blind, visually impaired and other persons with print disabilities (VIPs).

Sincerely,

Paulette Goldweber Associate Manager, Permissions Wiley

pgoldweb@wiley.com

T +1 201-748-8765 F +1 201-748-6008

111 River Street, MS 4-02 Hoboken, NJ 07030-5774 U.S. permissions@wiley.com



From: Metzgar, Catherine J [mailto:metzgar2@illinois.edu]
Sent: Monday, May 11, 2015 1:25 PM
To: Geary, Daniel - Boston
Subject: Permission to reprint

Hi Daniel,

As I am putting my dissertation together, I would like to request permission to reprint the following paper of research that I conducted as a graduate student. Can you please advise on how to obtain permission to reprint?

Metzgar CJ, Preston AG, Miller DL, Nickols-Richardson SM. Facilitators and barriers to weight loss and weight loss maintenance: a qualitative exploration. *J Hum Nutr Diet*. 2014, Sep 18. doi: 10.1111/jhn.12273. [Epub ahead of print]

http://www.ncbi.nlm.nih.gov/pubmed/25231461

Thanks for your help.

Catherine Metzgar, RD Doctoral Candidate University of Illinois at Urbana-Champaign